DOCUMENT RESUME

ED 417 350 CE 076 113

AUTHOR Van den Berghe, Wouter

TITLE Application of ISO 9000 Standards to Education and Training.

Interpretation and Guidelines in a European Perspective.

CEDEFOP Document.

INSTITUTION European Centre for the Development of Vocational Training,

Thessaloniki (Greece).

ISBN -92-828-2822-0

PUB DATE 1997-10-00

NOTE 103p.

AVAILABLE FROM Bernan Associates, 4611-F Assembly Drive, Lanham, MD

20706-4391; phone: 800-274-4888 (catalogue no. HX-12-98-053-EN-C: 8.50 European Currency Units).

PUB TYPE Reports - Research (143) EDRS PRICE MF01/PC05 Plus Postage.

DESCRIPTORS *Certification; Developed Nations; Educational Assessment;

Educational Quality; Foreign Countries; Job Training; Postsecondary Education; *Quality Control; *Schools;

Secondary Education; *Standards; *Total Quality Management;

Vocational Education

IDENTIFIERS *European Union; *ISO 9000; Quality Assurance

ABSTRACT

This report brings together European experience on the interpretation and implementation of ISO 9000 in education and training (ET) environments. Chapter 1 discusses the importance of quality concepts in ET and summarizes key concepts of total quality management (TQM) and its relevance for ET. Chapter 2 introduces the ISO 9000 standards. It explains the relationship with quality assurance and TQM and gives an overview of the main principles of the standards: structure of the ISO 9000 family, underlying principles, main components, and certification process. Chapter 3 looks at the context for applying ISO 9000 in an ET environment. It examines the rationale behind adopting the standards: reasons for seeking certification, expected benefits, and different considerations by type of ET organization. Then, it highlights the main interpretation problems, explains the terminology, and proposes a more logical rearrangement of the paragraphs of the standards. Chapter 4 includes a detailed discussion and interpretation of every paragraph of ISO 9001. Practical suggestions for implementing the requirements are provided throughout the text. Chapter 5 discusses implementation issues. It indicates the minimum requirements before launching a certification exercise, summarizes positive and negative messages emerging from implementing the standards in ET institutions, and gives a qualitative overview of the benefits resulting from successful certification. Chapter 6 presents overall conclusions, one of which is that it will be the market which will decide whether the cost of ISO 9000 certification is worthwhile, whether its benefits will outweigh the drawbacks, and whether any other national or international quality scheme is more appropriate. (Contains 60 references.) (YLB)



Application of ISO 9000 standards to education and training

SEDEFOP

Interpretation and guidelines in a European perspective

European Centre for the Development of Vocational Training

U.S. DEPARTMENT OF EDUCATION Office of Educational Research and Improvement EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)

- This document has been reproduced as received from the person or organization originating it.
- Minor changes have been made to improve reproduction quality.
- Points of view or opinions stated in this document do not necessarily represent official OERI position or policy.

PERMISSION TO REPRODUCE AND DISSEMINATE THIS MATERIAL HAS BEEN GRANTED BY

TO THE EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)



BEST COPY AVAILABLE

Document

Application of ISO 9000 standards to education and training

Interpretation and guidelines in a European perspective

Author:

Wouter Van den Berghe

October 1997

First edition, Thessaloniki 1997

EDEFOP

Published by:

CEDEFOP — European Centre for the Development

of Vocational Training

Marinou Antipa 12, GR-57001 Thessaloniki

Tel. (30-31) 49 01 11 Fax (30-31) 49 01 02

E-mail: info@cedefop.gr

Internet: http://www.cedefop.gr

The Centre was established by Regulation (EEC) No 337/75 of the Council of the European Communities, last amended by Council Regulation (EC) No 251/95 of 6 February 1995 and Council Regulation (EC) No 354/95 of 20 February 1995.



A great deal of additional information on the European Union is available on the Internet. It can be accessed through the Europa server (http://europa.eu.int).

Cataloguing data can be found at the end of this publication.

Luxembourg: Office for Official Publications of the European Communities, 1998

ISBN 92-828-2822-0

© European Communities, 1998 Reproduction is authorised provided the source is acknowledged.

Printed in Italy



PREFACE

This report is one of the series of studies elaborated within the context of CEDEFOP's project on quality in vocational training. In conformity with the European Council Resolution of 5 December 1994 on the quality and attractiveness of the VET, CEDEFOP has commissioned and is currently commissioning studies on some key aspects of the debate on quality at European level. It should be clear that the Centre is working both on quality assurance and quality assessment issues. Quality being a multi-dimensional and relative concept, CEDEFOP has started its works on it with a synthesis of ideas and experiences in seven EU Member States complemented by additional information. This publication of "Quality issues and trends in vocational education and training in Europe" is actually available in English, German, Spanish and French.

A second report has been drawn up on one of the main tools for quality assurance and assessment, namely indicators. An overview of the various types of indicators, their scope and implementation with a proposal concerning design methodology for quality indicators is elaborated and it will soon be published in English, French and Spanish.

In relation to quality assessment CEDEFOP has organised jointly with the Greek OAED and the German FHVR-Berlin a conference on "Approaches to the evaluation of European Training, Employment and Human Resource Programmes", aiming to contribute to the definition of priorities so that European Programmes can be more efficiently implemented and evaluation methods improved.

An international and interdisciplinary exchange of views and information on employment policy, vocational training and social policy aspects of the evaluation of European training, employment and human resource programmes took place during the two days of this conference in Athens. The interested reader may find the related papers in CEDEFOP's panorama series in English, Greek, Spanish and German.

Following this, CEDEFOP has proceeded to the analysis of the evaluation practices of quality aspects in vocational training programmes. Five countries have been studied and a synthesis report is in preparation. It will be available in English and French.

Concerning exclusively initial vocational education and quality, CEDEFOP has chosen to limit itself to a comparison of school-based quality concepts and practices in two countries, which are well-advanced in this field: The Netherlands and Denmark.

Two national reports, merged into one under the title "Quality Debate in Initial Vocational Education", have been drawn up. The report, published in English, gives an overview of the policy context on quality and a detailed presentation of the cases of ten schools in both countries, which have adopted different and often complementary approaches to quality assurance.

Based on discussions with experts in the field, CEDEFOP realised that over the last decades the "quality wave" originally started within the manufacturing industry, has also hit the training institutions in Europe. Certification, and especially certification on the basis of the ISO 9000 standards being one of the most sought after certification mechanisms, CEDEFOP has entrusted Mr Wouter van den Berghe, an international expert with a con-



siderable experience in this area, with drawing up the present report on the ISO 9000 standards family in Education and training.

The study does not limit itself to presenting the ISO 9000 standards, and especially the ISO 9001 and ISO 9002, which are the two most relevant ones to highly and less highly regulated training and education providers and institutions, but it also provides for an interpretation of each paragraph of them in a European perspective. It also contains practical suggestions for their implementation and underlines the requirements set up before launching such an operation.

Based mainly on the experience gained by their implementation in the United Kingdom, France, Germany, Belgium, The Netherlands and Denmark, the report underlines the difficulties in adopting ISO 9000 standards in VET, some inherent to the very nature of the educational process and others to the standards themselves.

Now that the ISO 9000 standards are under revision for 2000, CEDEFOP published this report on their implementation in education and training environments in the hope of contributing to the exchange of experience and information on them.

We would like to thank the author, Mr Wouter van den Berghe and all the experts he has consulted for the present work, which – we are convinced – will be useful to the reader.

Tina Bertzeletou Project manager Stavros Stavrou Deputy director



Table of Contents

Ackno	Introduction 3 Context of the report 3 Background 3 Purpose of this study 4 Approach adopted 4 The importance of quality concepts for education & training 6 Some observations on quality in education and training 6 Emphasis on quality as a result of the dynamics of education and training structures 7	cknowledgements	
1.	Introduction	3	
1.1	Context of the report	3	
1.1.1	Background	3	
1.1.2			
1.1.3	Approach adopted	4	
1.2	The importance of quality concepts for education & training	6	
1.2.1	Some observations on quality in education and training		
1.2.2	Emphasis on quality as a result of the dynamics of education and		
	training structures	7	
1.3	Total Quality Management (TQM)		
1.3.1	Introduction	8	
1.3.2	Key concepts, principles and characteristics of TQM	10	
1.3.3	What can TQM achieve in education and training?	12	
2.	The ISO 9000 standards for quality assurance	15	
2.1	Introduction	15	
2.2	The relation between quality assurance, ISO 9000 and TQM	16	
2.3	Main principles of ISO 9000	17	
2.3.1	Structure of the ISO 9000 family of norms		
2.3.2	Main concepts and principles		
2.3.3	The content of ISO 9001		
2.3.4	The certification process	20	
3.	ISO 9000 in education and training	23	
3.1	The rationale for adopting ISO 9000 in an education or training organisation	23	
3.1.1	Main reasons for seeking certification	23	
3.1.2	Problems and disadvantages	26	
3.1.3	Differences by education and training organisations	30	
3.1.4	Conclusion		
3.2	Terminology and interpretation issues	32	
3.2.1	Main interpretation differences of the ISO 9001 requirements	32	
3.2.2	Using ISO 9000 terminology in an education and training context	34	
3.2.3	Rearrangement of the requirements of ISO 9001	36	



4.	Interpretation of ISO 9001 requirements for education and training organisations	39
4.1 4.1.1	Management responsibilityQuality Policy	40
4.1.2	Organisation	
4.1.3	Management review	
4.2	Quality system	43
4.2.1	General	
4.2.2	Quality systems procedures	
4.2.3	Quality planning	
4.3	Contract review	
4.3.1	General	
4.3.2 4.3.3	ReviewAmendments to contracts	
4.3.4	Records	47
4.4	Design control	
4.4.1	General	48
4.4.2	Design and development planning	
4.4.3	Organisation and technical interfaces	50
4.4.4	Design input	50
4.4.5	Design output	51
4.4.6	Design review	52 52
4.4.7 4.4.8	Design verification Design validation	
4.4.6 4.4.9	Design changes	53
4.5	Document and data control	
4.5.1	General	53
4.5.2	Document and data approval and issue	55
4.5.3	Document and data changes	55
4.6	Purchasing	56
4.6.1	General	
4.6.2	Evaluation of subcontractors	
4.6.3 4.6.4	Purchasing dataVerification of purchased products	58 58
	·	
4.7	Control of customer supplied product	
4.8	Product identification and traceability	
4.9	Process control	60
4.10	Inspection and testing	63
4.10.1	General	
4.10.2	Receiving inspection and testingIn-process inspection	
4.10.3 4.10.4	Final inspection and testing	66
4.10.4	Inspection and test records	66



4.11 4.11.1	Control of inspection, measuring and test equipment	
4.11.2	Control procedures	
4.12	Inspection and test status	
4.13	Control of nonconforming product	69
4.13.1	General	69
4.13.2	Review and disposition of nonconforming product	70
4.14	Corrective and preventive action	
4.14.1	General	
4.14.2 4.14.3	Corrective actionPreventive action	
		•
4.15	Handling, storage, packaging, preservation and delivery	
4.16	Control of quality records	
4.17	Internal quality audits	76
4.18	Training	78
4.19	Servicing	79
4.20	Statistical techniques	79
4.20.1	Identification of need	
4.20.2	Procedures	80
5.	Implementation of a quality system based on ISO 9000 standards	81
5.1	Introduction	81
5.1.1	Purpose of this chapter	
5.1.2	Conditions to be met before launching a certification exercise	
5.1.3	What is the minimum quality documentation?	
5.2 5.2.1	Implementation of the quality systemGeneric issues	
5.2.1 5.2.2	Generic issues Positive messages from implementation experiences	
5.2.3	Negative messages from implementation experiences	
5.3	First results of ISO 9000 certification in education and	
5.3.1	training	85
5.3.1 5.3.2	A marginal phenomenonBenefits of certification for education and training	
5.3.3	Drawbacks of certification	
5.3.4	10 tips and conditions for success	
6.	Conclusion	91
Bibliog	raphy	93
About 1	he author	96



Acknowledgements

This report could never have been written without the kind support of many individuals who have shared their experience and insights with me. I am also much indebted to all the people and organisations who, over the last few years, have provided me with useful information on quality issues in education and training, and on ISO 9000 in particular. There are a number of people who awoke the 'quality' flame in me and strongly encouraged me to undertake further work in this area: Ed Prosser, then Director of the Comett Technical Assistance Office, who paved the way for developing my quality management skills; David Pentacost, who taught me the concepts and methods of TQM; Ludo Gelders and his quality team at the Department 'Industrieel Beleid' of the KULeuven, where I first saw ISO 9000 in operation in an education environment; William Govaert, formerly in the Flemish Comett Information Centre, who supported my first research into the issues of quality in education and training; Luk Vermeulen, of the Technisch Instituut Glorieux, who was always willing to share his experience with quality management in a technical school.

When I became a consultant for education and training organisations regarding ISO 9000, I learned a lot from the people and organisations I worked with, of which are to be mentioned in particular: Paul Garré and Dirk De Ceulaer (EHSAL), Hans Romaen (Technologisch Instituut), and Valerie McConaghy (May International).

Over the last two years, or specifically for this study, numerous individuals and organisations have forwarded me relevant information concerning ISO 9000 in education and training, often of an internal nature. These include, in addition to the people already listed: Wilfried Bartz and Rüder Keuper, Technische Akademie Esslingen (DE); Karl-Heinz Brehm, Siemens (DE) and ISO; Vincent Calmettes, SISIFE (FR); Søren Casparii. DIEU (DK); Dominique Delferrière, Management Information (BE); Sabine De Ruelle, Vlerick School for Management (BE); L. Jungerius, NCATB-Elsevier Opleidingen (NL); Mr Frances, CESI Languedoc-Roussillon (FR); Peter Hector (†), Aquaforce project (UK); Monika Kegelmann and colleagues, Certqua (DE); Annalis Larsen, Hillerød Technical College (DK); Mr Mol, Aval Lasinstituut (NL); staff from NAACB (UK); Søren Nielsen, DEL (DK); Edgar Sauter, BIBB (DE); Ria Sturm, NS Opleidingen (NL); Eduoard Touboul, CEIFCO (FR). Special thanks also to staff of certifying bodies, in particular Patrick Bellon, ISO 9000 lead auditor with AIB-Vincotte (BE), who kindly reviewed the interpretation of the Standard in Chapter 4. Staff from DNV (NL) and BSI Quality Assurance (UK) also were helpful in locating additional information and providing lists of certified education and training organisations.

Finally, I wish to thank CEDEFOP and its staff, particularly Tina Bertzeletou, for the support received, but also for the trust and encouragement.

Wouter Van den Berghe

October 1997



Introduction

1.1 Context of the report

1.1.1 Background

Over the last decade the topics Quality, Quality Assurance and (Total) Quality Management have become a central preoccupation of hundreds of thousands of organisations in Europe. It was the industrial world which first recognized that much of the growth and economic success of Japan could be attributed to the consistent focus of companies on quality and quality improvement. Since the mid-80s in particular, European firms have started to adopt the concepts and methods of Total Quality Management – in short 'TQM' - to meet the quality level which customers expect and for improving continuously the quality of products and services they deliver.

One of the more visible features of this 'quality wave' has been, particularly in Europe. the certification of the quality assurance mechanisms on the basis of the ISO 9000 Standards. Despite the fact that the ISO 9000 norms cover only a sub-set of the TQM principles, this form of certification is becoming the de facto basic quality standard in many industrial sectors of the more developed European countries. As an example, in the chemical industry in the Rotterdam port area it has become almost impossible for any type of supplier to deliver products or services without being an ISO 9000 certified company. In the United Kingdom, an ISO 9000 certificate has already been awarded to tens of thousands of organisations. Most other European industrial sectors and regions are catching up in this regard. By the end of 1996), over 100 000 organisations in the world had been certified, the large majority of these being European.

The ISO 9000 standards had originally been conceived for companies in the manufacturing industry, in particular subcontractors to large industrial concerns. However, since the early 90s in particular, application of the norms has quickly spread to other sectors of the economy. Indeed, notwithstanding terminology and interpretation problems, most quality experts agree that the requirements set forward in the standards can provide a suitable framework for the quality assurance system of any type of organisation, whether large or small, or whether product or service-oriented. Although this framework is not the optimal one for all types of organisations, and the value of an ISO 9000 certificate differs by sector and country, the developments over the last years have resulted, at least in Europe, in a broad recognition of the value of an ISO 9000 certificate and its function as a quality label.

In this context, it will not come as a surprise that interest for the ISO 9000 quality standards also grew in the education and training world. The first groups to pay attention were providers of continuing education and training for the business world, as well as, to a lesser extent, vocationally oriented schools. Both types of organisations are indeed closer to market needs and business developments than regular education institutions. Some general education schools and higher education institutions have also taken the ISO 9000 road; a few pioneers have already passed the certification stage. This development is still in its very early stages. It is only since the early 90s that the first education and training institutions in Europe have obtained an ISO 9000 certificate, some of them even for only part of their activities. The numbers are increasing, slowly but steadily, particularly for continuing education and training providers; about a hundred of these had been certified by the end of 1996, mainly from the UK, Germany, the Netherlands and France.

The pace of this development is not dissimilar to trends in other service sectors (and public sectors in general) which have not been as eager as manufacturing to jump on the



ISO 9000 bandwagon. Additional hurdles for education and training providers are their limited resources and/or their small size. But even when such – important – practical issues are left aside, for many practitioners the real added value of such a certification process remains doubtful. Information is lacking on the conditions under which ISO 9000 certification can be considered as a viable and, eventually, successful quality strategy for particular types of education or training institutions.

1.1.2 Purpose of this study

Individuals across Europe have recently gained valuable experience regardingthe interpretation, application and relevance of ISO 9000 for education and training organisations. These people include staff members of the establishments concerned, researchers, consultants and certified ISO 9000 auditors. So far little exchange of experience has taken place, in particular across borders and language areas; what is known has not yet received widespread attention in the education and training world. Relevant information is often hard to find, specifically in countries where few or no education or training providers have been certified. Moreover, little effort has been undertaken to compare the approaches and results obtained across different European countries.

This report aims at bringing together European experience on the interpretation and implementation of ISO 9000 in education and training environments. At the same time, the document has the ambition to serve as an introductory guide for education or training establishments who might consider certification. Not all aspects could be covered to the same extent. The focus was on:

- the correct interpretation of the ISO 9000 standards specification for education and training providers;
- the experience gained with implementing such a quality assurance system;
- the first results regarding added value, potential benefits and drawbacks.

Originally this study intended considering vocational education and training providers only. In the course of the research, it appeared more appropriate to establish a document report which might be of benefit for all types of education and training organisations: schools, higher education institutions, adult education providers, private training centres, training departments within firms, etc.

1.1.3 Approach adopted

Ideally, this report should have been based on the in-depth analysis of some twenty or more case studies. Given the considerable geographical spread of certified institutions in Europe and the modest level of support available, a more pragmatic approach had to be taken, however. No *new* cases were investigated, but extensive use was made of the material which was already available or which was kindly provided to the author (cf. Acknowledgements). It was not feasible either to launch and process a European-wide survey on this matter. Despite these limitations, an attempt was made to gather and study as much information as possible. This process included the analysis of:

- existing "interpretations" or clarifications of the use of ISO 9000 in various education and training settings;
- descriptions of case studies or accounts of ISO 9000 implementation stories;
- reports or articles which discuss issues related to the relevance, benefits and problems concerning the application of ISO 9000 to education and training;
- quality manuals of ISO 9000 certified education and training organisations;
- specific checklists used by ISO 9000 auditors.



Almost all the documents referred to (some of which are confidential) were obtained from people and organisations in the United Kingdom, France, Germany, Belgium, the Netherlands and Denmark. Extensive searches were also conducted on the Internet World Wide Web, but on this particular topic, this proved not very successful. The analysis of documents was complemented by oral discussions and interviews (mostly via telephone) with over twenty people in six countries who – as practitioner, consultant or auditor – had been involved in the implementation of ISO 9000 standards in education or training organisations. The research could be undertaken efficiently since the author had a profound knowledge of ISO 9000, provides training on the standards and their application in education and training, and acts as a consultant for several education and training institutes who are working towards (or have already obtained) an ISO 9001 certificate.

After this introductory session, this chapter also recalls the importance of quality concepts in education and training (Section 1.2) and summarizes the key concepts of Total Quality Management (TQM) and its relevance for education and training (Section 1.3).

Chapter 2 provides a short introduction to the ISO 9000 standards. The relationship with quality assurance and TQM is explained, and an overview is given of the main principles of the standards: structure of the ISO 9000 family, underlying principles, main components, and the certification process. This Chapter can be skipped by those who are already familiar with the standards.

The next chapter looks at the context for applying ISO 9000 in an education and training environment. Two main issues are discussed. First, Section 3.1 examines the rationale behind adopting the standards in an education or training organisation: the reasons for seeking certification, the expected benefits, and different considerations by type of education or training organisation. Section 3.2 discusses terminology and interpretation issues. The main interpretation problems are highlighted, the terminology is explained, and a more logical rearrangement of the paragraphs of the standards is proposed.

Chapter 4 is the core of the report. It includes a detailed discussion and interpretation of every paragraph of ISO 9001. Where appropriate, differences by type of education or training organisations are highlighted. Practical suggestions for implementing the requirements are provided throughout the text. Incidentally, ISO 9001 lists in its fourth chapter all its requirements; consequently, the numbering of the sections in this report corresponds exactly with the related sections in ISO 9001 (for instance, Section 4.5 of this report discusses the requirements of section 4.5 of ISO 9001).

Chapter 5 discusses implementation issues. It indicates the minimum requirements before launching a certification exercise (Section 5.1), and summarizes both positive and negative messages emerging from implementing the standards in education and training institutions (Section 5.2). The final section of this chapter gives a qualitative overview of the benefits (or otherwise) resulting from successful certification.

Chapter 6 presents the overall conclusions of the study.

The report does not include a detailed implementation strategy for putting ISO 9000 into practice. The main reason is that implementation varies considerably from one organisation to another. The approach to be adopted depends on factors like: size of the organisation, type of education and training programmes offered, characteristics of customers, degree of standardisation, ancillary services provided, other activities within the scope of the quality system, etc. Implementation strategies will also depend on whether certification is sought for the whole or only part of the organisation. Implementation issues have, however, not been discarded in this report. Section 2.3.4 provides general information about the key steps involved in certification, Chapter 4 includes many suggestions on



how particular requirements of the Standards can be put into operation, and Chapter 5 contains a series of observations resulting from successful implementation.

This report was primarily intended as a guide for education and training organisations, and for policy makers. It does not include, therefore, a critical discussion or in-depth comparison of the differences and interpretations and approaches adopted across Europe. All public and non-confidential sources consulted have been grouped and listed in the Bibliography.

1.2 The importance of quality concepts for education and training

1.2.1 Some observations on quality in education and training

It is not the purpose of this report to discuss the reasons for the increasing emphasis on quality issues in the education and training world – and how governments, institutions and learners are coping with these challenges. These topics are dealt with more extensively in the recent CEDEFOP report "Quality Issues and Trends in Vocational Education and Training in Europe". Therefore, only some general points are recalled here.

The first observation is that quality is not a new subject in education and training. Institutions, teachers, administrators, policy makers and learners have always been concerned with quality. Indeed, the quality of an education or training provider eventually depends on the performance of the learners. Even without a formalised 'quality' approach such as *Total Quality Management (TQM* – see next chapter) or ISO 9000, schools and training providers have needed methods, norms, procedures and standards to ensure the quality of their provision. But it is equally true that, traditionally, quality has often been interpreted fairly narrowly, focussing on particular features of the education and training services delivered.

This brings us to the second general observation: in education and training, just like in most other sectors, the dominant quality ethos tends to change over time. The current trend is one whereby quality concerns are increasingly focusing on the total effectiveness of an education provider, whether that is a vocational school, a university or a private training institution. This tendency mirrors the industrial developments where quality considerations are moving towards the *organisational* capacity to deliver high quality goods and services (the systems thinking behind the Total Quality Management concepts).

This leads us to a third important observation: the current concerns for quality in education are no isolated or temporary phenomenon, but are part of broader macro-economic trends. It does not imply that 'older' types of quality considerations (e.g. didactics) are no longer valuable. Rather, an effective 'Total' Quality strategy in education should be able to incorporate the more traditional quality perspectives.

A fourth general observation is that the 'new' approaches to quality assurance and management in education and training mean something different for each type of education and training. Simply stated:

- a shift in emphasis in schools, from a focus on the quality of the teacher towards the performance of the institution as a whole;
- the introduction of new or additional quality control mechanisms in higher education;
- the creation, for the first time, of quality assurance systems and performance related mechanisms in continuing education and training.



A common characteristic for all types of education and training is the increased concern for the performance of the learner (the effectiveness of learning) and the effectiveness and efficiency of the provider as a whole. This explains also the interest in TQM and quality assurance approaches.

A final observation: notwithstanding the increased emphasis on organisational factors as a prerequisite for quality assurance, it is recalled that no effective learning can take place without high commitment of the teacher or trainer. The corollary is that quality approaches – including TQM and ISO 9000 – are doomed to fail if they do not support the inner motivations of teachers and are able to sustain or increase their commitment.

1.2.2 Emphasis on quality as a result of the dynamics of education and training structures

The current education and training systems in Europe have been shaped by many and different social, cultural, pedagogical, economical and employment factors. All these elements have always been present, but their relative importance tends to change over time and also by type of education and training. The impact of external factors on the education system has also been variable over time. Two important trends may be observed in this regard:

- external demands (from governments, students, employers,...) on the education system are increasing; this puts pressure on the deployment of resources and the efficiency of the organisation
- the continuing education and training sector is becoming a more mature and 'established' economic sector, alongside many other service sectors.

Such trends suggest that the education and training world is losing much of its special status, and is more and more considered like an 'ordinary' economic sector. It also implies that schools, universities and training providers are increasingly expected to perform at high level, behave professionally and provide quality services throughout. The education and training paradigms are changing from supply-driven teaching to demandled learning. Although many educators do not feel comfortable with such developments, they would seem to be inevitable. Indeed, similar custom-driven trends can also be witnessed in other areas, such as public services.

Amongst the specific factors which contribute to the growing emphasis on quality in education and training, the following may be listed (some of these concern all types of education and training, while other arguments apply only to particular segments of the education and training system):

- When it comes to quality, people increasingly reject the historically grown distinctions between products and services, between profit and non-profit, public and private, and small and large organisations. The broad choice and high levels of product and service quality which are now available in developed countries raise citizens' expectations and make them critical of low-quality performance anywhere including in education and training.
- The wide choice available in society gives citizens more power and increases the appetite for change, flexibility and customising. Moreover, people's ways of life and expectations are less uniform than in the past. In education and training this trend is visible through the more complex and variable qualification requirements.
- In this information age, once certain information enters the public domain, citizens tend to expect to get such information in the future as well. Thus, the more information be



comes available on quality and performance results in education and training, the more demand there will be for it.

- Public education institutions are, like most other public services, increasingly called upon to become publicly accountable for what they do and to demonstrate that they deliver a quality service.
- A long term argument for quality in education is that of its own survival, since a well-funded education system can only exist under a flourishing economy. In the current global competitive environment any inefficiency and lack of flexibility, also in education, will be penalized by lower economic growth and hence, eventually, to fewerless resources available for education.
- In the relatively stable environment of the past, quality in education and training could almost entirely be attributed to the inherent abilities of the teachers and trainers. Because of the rapidly changing environment, new demands, more complex activities, differentiation, and increased customer involvement, these abilities alone no longer suffice for guaranteeing quality.
- Competition between education institutions is becoming a major issue. For higher and continuing education it is already even crossing national borders. This competition is more and more based on facts and on the real quality of a particular institution, and no longer on its historical standing and reputation. In such a more competitive environment, there is the need to be able to sustain a high quality image with customers, and be able to demonstrate quality on a ongoing basis.
- Many types of education and training organisations, particularly in higher education and most of vocational and continuing education, face an escalation of costs, in order to provide high quality provision. At the same time, public education also faces constraints on public expenditure, which forces governments to find ways to get more for less. Similarly, companies increasingly expect to get more out of their training budget.
- The increased variety and complexity of education and training provision calls for mechanisms to ensure better transparency of the quality on offer.
- The quality trends in continuing education and training are also related to changing attitudes within firms themselves. Companies increasingly recognize the importance of continuing training of their employees as a critical factor for their long term market success. Consequently, optimising this investment through quality assurance arrangements both within the firm and with the providers is becoming a necessity.

Thus, there are many factors which, together, call for greater attention to quality in education and training – and hence for mechanisms, methods and systems which can help ensure quality all the time. The above list shows that, while some of these factors are internal to the education structures, most are external. Education and training systems are – increasingly – interwoven with the rest of society, and therefore subject to similar pressures and trends.

1.3 Total Quality Management (TQM)

1.3.1 Introduction

Simply stated, Total Quality Management (TQM) is an organisational strategy and a management approach which involves all employees and is aimed at continuously improving the organisation's effectiveness in achieving customer satisfaction. In this definition, the customer notion is interpreted broadly, and includes also the 'internal customers' within an organisation itself. The concepts and application of TQM originate from Japan, where the systematic use of TQM principles and methods in the production sector made



the country an economic superpower in just a few decades. Since the early 80s, the value of TQM has increasingly been recognised by business in the United States and subsequently Europe.

While originally limited to the private sector, and more particularly manufacturing enterprises, the early successes of TQM contributed to the spreading of its principles and methods across other sectors, including public services and non-profit organisations. The main 'logic' behind the adoption of TQM methods is as follows:

- the increasing competition requires continuous improvement of quality and productivity;
- customers determine purchasing choices based on their perception of quality, and hence determine the competitive position of the suppliers;
- an organisation must adopt the TQM paradigm to achieve this customer focus, and to be able to generate high quality all the time at the lowest possible cost.

Empirical evidence confirms, by and large, the validity of these hypotheses. Because of this, TQM has become the predominant approach to quality in the private sector, and increasingly in other areas of economic activity as well.

Although some official definitions of TQM exist, it must be recognised that the scope of 'Total Quality Management' is not very well defined. Each of the leading advocates and 'gurus' of TQM puts emphasis on particular aspects. Since the economy is in constant movement, and needs evolve, the balance between the different components of TQM tends to change over time. Aspects which were central to TQM some decades ago in Japan, may have lost their prominent place today. Thus, TQM is more a catalyst for existing trends and tendencies, than a standard or a precise catalogue of methods and prescriptions.

Over the last years some generic 'models' for Total Quality Management have been developed. In the United States, the first models were based upon the ideas of some quality gurus like Deming, Juran and Crosby. The more elaborated system which has been developed out of this is the set of criteria for the Malcolm Baldridge Award. This has become the standard reference model for TQM in the United States, with adaptations for specific sectors (including education) being piloted.

In Europe, the most important model is that developed for the European Quality Award (EQA), promoted by the industry-led European Foundation for Quality Management (EFQM). The EQA-model covers nine different areas: Leadership, People management, Policy and Strategy, Resources, Processes, People Satisfaction, Customer Satisfaction, Impact on Society, and Business Results. For each area, there are a number of subcriteria with specific questions for self-assessment. The premise behind the EQA-model is that "customer satisfaction, people satisfaction and impact on society are achieved through leadership, driving the policy and strategy, people management, resources and processes, leading ultimately to excellence in business results". The EQA model has also been at the basis of the criteria of national quality prizes in many European countries. Adaptations of the EQA-model have been made for the education sector, including a sophisticated one for Flemish (vocational) higher education (the so-called 'PROZA'-system).

In addition to such standard models, many specific quality systems have been developed, both at the level of (larger) companies, and for specific sectors. Over the last years, traditional 'inspection based' quality approaches in many European countries have been converted into assessment models which reflect more the TQM paradigm. Some excellent assessment and audit instruments have been developed, particularly in the Anglo-



Saxon world (e.g. the SQMS-system in Scotland). It should be noted that most quality control systems in education are predominantly based on quality assurance principles – which implies that they reflect some of the spirit of ISO 9000 (see next chapter).

1.3.2 Key concepts, principles and characteristics of TQM

When analysing the literature, one will find a broad consensus amongst quality experts about the key characteristics of TQM – although they may not agree as to their relative importance. There are also a number of elements which are considered as part of TQM by some, but not by all specialists. What follows is the author's own synthesis and positioning in these matters.

One may identify a 'TQM-organisation' through four different series of characteristics:

- underlying concepts;
- operational principles;
- implementation characteristics;
- typical results.

These series of characteristics are briefly described below.

First of all, there are the five underlying concepts – almost dogmas – of TQM:

- A clear customer focus. The first priority of an organisationand its staff must be to understand and to satisfy the needs and expectations of the customers and the chosen target group. Furthermore, this customer orientation also embraces the concept of the 'internal customer' within an organisation. For TQM to become effective, a clarification of the internal customer/supplier relationships within an organisation is needed, to ensure that all internal customers are satisfied by the internal suppliers.
- TQM requires *continuous improvement* in everything an organisation does. Quality is never-ending, and an organisation-wide attitude should prevail whereby the search for improvement always continues, by everyone in the organisation. Such improvements should be guided by a clear understanding of the mission of the organisation, and a vision of what it wants to become.
- Quality assurance of internal processes. This implies that standards are set, procedures to achieve them are defined, and adherence to these is guaranteed; problems occurring are remedied in a systematic way. Effective quality assurance contains the seed for continuous improvement (one cannot improve quality without controlling the current situation) but inefficient quality assurance leads to bureaucracy.
- Process orientation. The final quality of a product or service depends on all preceding processes. This principle points to the need for process thinking, for an integrated approach to the whole development and delivery chain, and for an optimisation of the internal customer/supplier interfaces. It is essential to identify all the critical processes in the organisation.
- Prevention instead of inspection to achieve quality. Through adequate preventive measures (appropriate design, decent planning, targeted training, adequate equipment, effective communication, etc.) Fewer quality errors will occur, customers will be more satisfied, and less inspection and control will be needed. The theory of 'quality costs' shows that the costs and time spent on preventive measures at the planning and design phase will eventually be lower than thecosts of inspection, defects, waste and repair.



Secondly, five important *operational principles* distinguish TQM-oriented organisations from the other ones:

- Management leadership and commitment. The driving force behind any TQM approach must be the commitment, vision and exemplary leadership of the senior management, which should be carried down to every level of management within the organisation. The leadership of management regarding TQM is essential for building a strong consensus within an organisation regarding quality. Managers should continuously communicate the TQM-message towards their employees.
- TQM requires *teamwork*. The benefits of TQM reside to a large extent in effective teamwork, particularly when involving people from different departments. Effective, cross-functional teamwork needs to be a key operational characteristic of any TQM effort. Such teamwork is both needed by, and the result of, effective TQM strategies. Teamwork also extends in working together with suppliers and possibly customers in addressing issues for improvement.
- Quality is everyone's job. Quality is not a functional or departmental responsibility which should be left to inspectors, or quality controllers. Achieving quality adherence to agreements, meeting customer expectations, avoiding waste should be the norm in all processes of the whole organisation. This goal requires the involvement of all employees at all levels and in all departments. It must be supported through human resource strategies that recognize and reward quality efforts.
- Focus on facts. Discussions and decisions on activities and resource allocations should be based on reliable and relevant information. This applies in particular to everything which relates to the customers, both external and internal (what are their needs, how do they feel about our products/services, etc.?). Of special importance for the development of a more quantitative thinking approach is the use of appropriate data gathering and analysis methods (including statistical ones) in order to constitute relevant information in a timely and efficient manner.
- TQM is based on *systematic problem-solving*. This requires the use of appropriate tools and methods for identifying weaknesses and areas for improvement, analysing them, tracing the sources of problems, seeking improvements and, finally, implementing them. In this context, the term 'problem' should be read as 'anything which should be improved'. While some of these tools may be quite sophisticated, most of them are very easy to apply, yet often remarkably powerful.

An organisation which supports these underlying concepts and adheres to the operational principles listed above, is likely to become a TQM-organisation. Whilst there are, in principle, many ways to implement all these principles, the reality shows that most TQM-organisations display the following *implementation characteristics*:

- The formulation of a clear *mission and vision statement*. These are communicated and accepted throughout the organisation, and serve as an anchor for the quality policy and strategy. In larger organisations, such statements also exist at department and unit level. Mission and vision statements include a clear identification of the customers to be served, and are complemented by the shared values of the organisation.
- The establishment of a *quality manual*, which describes the organisation, its policy, its key processes and the responsibility and authority of staff. In general, such an overall quality manual is complemented by procedural manuals at department level, describing the critical processes in sufficient detail. Access to up-to-date information in quality and procedural manuals is a necessity for quality assurance.
- Systematic training of staff throughout the organisation. Training needs are analysed and remedied where necessary, in order to ensure that all employees from the sen



ior manager until the shop-floor worker – are qualified for the activities they undertake and for the positions and responsibilities they might assume in the future. Training is complemented by human resource policies that reward improvement, professional development and quality achievements.

- Decision-making is delegated to the lowest possible level. The increased responsibility and authority of employees increases commitment and motivation and encourages mutual trust and support throughout the organisation. In line with this development, the organisation becomes flatter (fewer management layers) and managers become coachers rather than controllers.
- Customers are asked for feedback all the time, often through regular customer surveys aimed at understanding needs and expectations, checking satisfaction with the quality provided, and detecting new trends. Such surveys are systematically analysed in order to implement changes where necessary. Linked to this, there is an effective system for rapid and effective complaint handling. This applies to all kinds of complaints and problems, including those inside the organisation. External customers are often urged to indicate their complaints. Within the organisation, the complaint system should gradually be replaced by a 'suggestion scheme' whereby any individual in an organisation may suggest improvements.

Finally, an organisation which has been working alongside the TQM principles is likely to be able to show the following typical *results*:

- better and more consistent *quality* of products and services provided;
- considerable reduction of defects, waste, problems, complaints, delays, etc.;
- regular and timely innovations in products and services;
- cost effective and efficient processes throughout the organisation;
- a highly motivated, qualified and self-confident workforce.

Obviously, the nature and the extent of such results will differ strongly across organisations. The extent and speed of achievement depends not only on how the concepts, principles and characteristics listed are put into practice, but also on other factors such as the economic and legal environment, the size of the organisation, and the quality expectations of customers.

1.3.3 What can TQM achieve in education and training?

The fact that possibly a million or so European organisations have adopted some kind of quality or TQM approach (including ISO 9000), implies they must have seen some benefit in doing so. This argument also applies to the education and training organisations who have so far embraced the TQM concepts. One might imagine that the primary reason to move in this direction would have been to improve the quality of their education and training provision. The reality is more complex, however. Based on reported experience and documented case studies, the main rationale behind the TQM approach of the pioneering institutions appears to have been:

- an improved external quality perception and image, thanks to clearer internal policy choices, better customer orientation and more effective marketing;
- a more efficient internal organisation, with more effective management, better motivated staff, and more successful internal communication;
- achievement of professionalism in non-educational services, i.e. the services and activities provided by the institution in addition to the delivery of the course programme (registration, administration,...);



■ raising the quality of the education and training services and products themselves: the relevance of the course programme, the didactic quality of instruction, the effectiveness of the needs analysis,...

This is, actually, also the order in which the benefits of TQM seem to materialize. The fact that the quality of education and training itself is not the central or priority issue may be explained by the strong position of the early adopters in this regard. But it may also suggest that adopting TQM (and ISO 9000, as we will see) does not make a lot of sense if some basic quality level is absent.



2. The ISO 9000 standards for quality assurance

2.1 Introduction

Given the abundance of books available on ISO 9000, there is little added value in providing an extensive description of the norms, their origin and their applicability. This chapter, therefore, only highlights some key issues which are important for a good understanding of the more in-depth discussion on the application of the standards in education and training (Chapters 3, 4 and 5).

"ISO 9000" is the common name for a family of international standards for quality assurance within organisations. Their origin dates back to the development of the American military industry in the early 50s, when the significant expansion of production capacity, increased security requirements, and large-scale use of suppliers made the establishment of 'military' quality standards for suppliers mandatory. In the 50s and 60s similar developments could be observed across the developed world in other areas, such as the nuclear sector, the pharmaceutical industry, and car manufacturing. In general, the system was one whereby the customer (e.g. an automobile plant) audited the suppliers or potential suppliers against the specific standard.

Because this practice led to a proliferation of standards and audits – e.g. a supplier might be subject to several different audits by its major clients – the authorities in the United Kingdom published a 'generic' standard, BS 5750 in 1979. This adoption of this standard provided two main advantages:

- suppliers from a wide range of sectors needed only to comply with one standard, rather than with multiple ones;
- only one audit was required, by an independent body, who subsequently *certified* the supplier company.

Thus, several customer companies no longer needed to undertake different audits themselves, but could rely on the audit and certification by a third party. In addition to the obvious cost-effectiveness of this approach, it also solved the problem of confidentiality regarding the know-how of the suppliers.

The success of this new approach was rapidly observed outside the United Kingdom. In 1987, therefore, the International Standards Organisation (ISO) issued a set of norms, "ISO 9000", which were almost a direct copy of the British standard. Over the following years, the family was expanded with additional norms and guidelines. ISO 9000 became a European standard series for quality assurance (first called EN 29000, now called EN ISO 9000) which was gradually adopted in most European countries. At international level, a first revision of the norms was finalised in 1994, but the modifications were relatively minor. However, a major revision is underway, expected to be ready around 2000. This new version of the standard will be structured differently, will reflect more a TQM than a quality assurance approach, and will use a terminology which is applicable to virtually all economic sectors.

Indeed, with the publication of ISO 9000 a somewhat unexpected phenomenon had occurred. Although initially designed for use in contractual supplier-customer relationships only – and predominantly in the production sector – many different types of organisations started to show interest in the norms as well – first in the UK, subsequently in several other European countries, and finally in the USA and Asia. It was realised that the generic nature of the specifications made the norms applicable to most types of organisations, even non-profit ones – despite the fact that interpretation of some of the sections in a non-manufacturing context was not straightforward. The main reasons for this interest were: (1) the easy recognition by the market of the certificates; (2) the official 'proof' by an



independent authority of the quality level of the organisation; and (3) the absence of any better or more suitable international norm.

Thus, ISO 9000 has become the *de facto* industry standard for quality assurance for any type of organisation. By 1996 more than 100 000 organisations (predominantly in Europe) had already been certified, and the numbers keep increasing, albeit it at different speeds in different countries. The typical ISO 9000 "waves" in a country are:

- firstly, the manufacturing sector show interest;
- then the service firms directly related to production companies come into the picture;
- and finally organisations from other service sectors get involved, including public services and non-profit organisations.

It should be noted most European countries (with the exception of the United Kingdom) are still in the first wave, although the beginning of the second and pioneers of the third wave can be observed. This – simplified – "wave" model explains partially why providers of continuing education and training are typically the pioneers in the education sector.

2.2 The relation between quality assurance, ISO 9000 and TQM

In the official denomination of the different ISO 9000 standards, there are explicit references to "quality assurance". We have seen before (cf. Section 1.3.2) that quality assurance is an important part of a TQM approach. So, how do ISO 9000, TQM and quality assurance relate to each other?

The term "quality assurance" is defined in ISO 8402 – a standard containing official definition of key quality terms – as follows: "All the planned and systematic activities implemented within the quality system, and demonstrated as needed, to provide adequate confidence that an entity will fulfil requirements for quality".

In practical terms, quality assurance requires an organisation to ensure that:

- quality standards are defined for all activities to which quality assurance applies (a quality standard is a well defined set of minimum criteria which a particular output, product, service or process should meet);
- suitable procedures are available for ensuring that quality standards are met;
- procedures are systematically monitored for conformance, using statistical methods where appropriate;
- causes of the non-conformances identified are analysed;
- causes of problems are eliminated through appropriate corrective action, in order to avoid or at least minimise the probability of those problems reoccurring.

When all of this is undertaken – in whatever way – for a set of activities, processes or outputs, it can safely be stated that 'quality assurance' is taking place. It should be noted that quality assurance is mainly concerned with the *quality of conformance*, as opposed to *quality of design*, i.e. the quality assurance principle assumes that quality standards have been adequately designed, and that only conformity with these standards needs to be ensured.

Although all of this looks like common sense, the reality in many organisations is often quite different:

- quality standards are vague or ill-defined or even entirely implicit;
- the existing procedures are incomplete, conflicting or not used;



- inspection or verification of performance is irregular, or happens only when major problems have occurred;
- the real causes of problems are left untouched;
- only short term corrective measures are taken when problems arise, with the risk of the problem reoccurring.

If most of such features apply well to a particular organisation, then it will have to conduct a long and painful exercise to implement a quality assurance system. It should be noted that putting quality assurance mechanisms in place may require several iterations before the whole system becomes stable and normalized.

Although 'quality assurance' is only one of the underlying concepts of a TQM approach (cf. Section 1.3.2), it should be recognised that it is closely linked to the other components of TQM (all of which are, in fact, closely linked to each other). For instance:

- the customer orientation should be reflected in the definition of standards;
- continuing improvement is only possible when the current system is well known and under control;
- the effective application of quality assurance requires a process orientation;
- a well functioning quality assurance system reflects a preventive attitude, and will prevent problems.

Thus, effective quality assurance is an important milestone towards a TQM-based organisation; similarly, implementing TQM without adequate attention to quality assurance concepts will eventually fail. On the other hand, it must also be recognised that excessive focus on quality assurance may yield counterproductive effects, in particular when too much attention is paid to prescribing standards and excessive inspection. Such a "reductive" interpretation — whereby the suitability of the procedures, the analysis of problems and the eradication of problem causes are neglected — would quickly result in a bureaucratic system.

What then about ISO 9000? At first sight, it would appear that the ISO 9000 standards are concerned with quality assurance alone. This spirit penetrates the whole document, particularly through the numerous requirements for documented procedures, document control, quality records, inspection and verification, and corrective measures. However, a closer look shows that the standards cover more than quality assurance alone, and pay attention to other aspects of TQM, such as management leadership, process orientation, and quality improvement mechanisms (through complaint handling, internal audits). In a small enquiry amongst certified education and training providers, conducted for this study, several respondents confirmed that the type of management style needed for a successful TQM approach, does follow almost automatically from an effective and fully operational ISO 9000 system. But the respondents also recognised that ISO 9000 alone is not enough to cover all elements of TQM.

2.3 Main principles of ISO 9000

2.3.1 Structure of the ISO 9000 family of norms

ISO 9000 is really a *family* of norms, consisting of different components. The most important ones to quote here are:

- ISO 9000-1 (version 1994): Quality management and quality assurance standards Part 1: Guidelines for selection and use;
- ISO 9001 (version 1994): Quality systems Model for quality assurance in design, development, production, installation and servicing;



- ISO 9002 (version 1994): Quality systems Model for quality assurance in production, installation and servicing;
- ISO 9003 (version 1994): Quality systems Model for quality assurance in final inspection and test;
- ISO 9004-1 (version 1994): Quality management and quality system elements Part 1: Guidelines.

The other norms of the series concern further guidelines on the use of the standards in specific situations.

Some related standards, such as those on definitions (ISO 8402), on auditing quality systems (the series ISO 10011) and on quality manuals (ISO 10013), are often considered to be part of the ISO 9000 "family".

Certification (see below) of organisations is only possible on the basis of the norms ISO 9001, ISO 9002, or ISO 9003. All the other 'norms' consist of definitions, guidelines or interpretations. For instance, no certification is possible on the basis of the guidelines of ISO 9004-1, a document which reflects to a large extent the characteristics of TQM. Whenever reference is made to "ISO 9000" certification, it actually means certified conformity with the requirements of either ISO 9001, 9002 or 9003.

It is important to understand that the ISO requirements do not concern product specifications, but rather aspects of management, organisation and processes. A product or service can never be 'ISO 9000' certified, but the organisation behind the product may. Where product criteria exist (e.g. impurity levels for food), ISO 9000 certification actually means that a system is in place which ensures that these product criteria are always met.

2.3.2 Main concepts and principles

Thus, the ISO 9000 norms can be considered as generalised specifications for quality assurance, with additional elements borrowed from TQM. The norms do not prescribe how production and management should be organised, but require, simply stated, that an organisation:

- defines and plans its processes,
- documents them properly,
- checks their capability, and
- ensures that they are controlled and reviewed.

Although the ISO 9000 requirements go much more into detail, when these simple principles are adopted in all processes of an organisation, it will be more than 90% "ISO ready".

The ISO 9001-9003 standards consist essentially of a number of management principles expressed in production terminology. The norms' terminology reflects their original focus on situations where suppliers needed to demonstrate to their industrial clients their capability of guaranteeing the conformity to the product specifications. The ISO 9001-9003 norms are heavily process-oriented, with emphasis on thorough planning, documentation and control. The requirements are described in general terms, but the terminology is derived from manufacturing. The principles of the standards can be summarized as follows:

- the organisation has clear quality objectives;
- there are clear agreements between everyone involved;



- the organisation has the resources to achieve the required quality level (the notion of 'capability');
- the organisation defines itself which processes and resources are needed for quality;
- all processes and systems are under control, with evaluation and modification when appropriate;
- everything needed for quality assurance is documented;
- quality registrations allow verification and 'proof' of quality assurance

Given the historical development of the standards, some activities receive more attention than others. Some paragraphs are phrased in terms which makes them applicable to virtually any type of organisation, while others use a 'manufacturing' language. Thus, certain parts may need interpretation – or may not be applicable at all. It is recalled that this need for a European interpretation of ISO 9000 in education and training was one of the main rationales for writing this report.

In general, the norms indicate which type of activities need to be documented and controlled (contract, design & development, production, purchasing, delivery, etc.), but do not prescribe exactly how that needs to be done. The standards are more prescriptive regarding the management aspects (quality policy, management review, management representative, ...) as well as for some specific quality assurance measures, such as document control, test and inspection, quality records, corrective measures and internal audits. Implementing all these requirements results in a so-called "quality system". Such a quality system will extend beyond the minimal requirements for a quality assurance system.

2.3.3 The content of ISO 9001

ISO 9001 is the most complete standard, and embraces ISO 9002 and 9003. The latest release of ISO 9001 dates from July 1994 and was accepted as a European norm (EN ISO 9001) somewhat later. All Member States of the European Union have since adopted the standard also as a national norm, sometimes with some delay because of the translation. The document consists of 4 'chapters', of which the first three ('Scope', 'Normative Reference', 'Definitions') are very short – together less than a page. Chapter 4, called 'Quality system requirements' is what the norm is all about. It consists of 20 parts, often referred to as 'sections', 'paragraphs' or 'clauses':

4.1	Management responsibility	4.12	Inspection and test status
4.2	Quality system	4.13	Control of nonconforming product
4.3	Contract review	4.14	Corrective and preventive action
4.4	Design control	4.15	Handling, storage, packaging, preserva-
4.5	Document and data control		tion and delivery
4.6	Purchasing	4.16	Control of quality records
4.7	Control of customer-supplied product	4.17	Internal quality audits
4.8	Product identification and traceability	4.18	Training
4.9	Process control	4.19	Servicing
4.10	Inspection and testing	4.20	Statistical techniques
4.11	Control of inspection, measuring and test		
	equipment		

Some of the paragraphs are very long and contain several sub-sections; others consist of a few sentences only. Since Chapter 4 examines and discusses each of these requirements, only some general comments are given at this stage.



Both the length of the different paragraphs, their level of detail and the sequence of the different requirements reflect the historical development of the norms. At present, work is going on within ISO, the International Standards Organisation, for a major revision of the standards, scheduled to be ready around the year 2000. In the new standard, the various sections will be rearranged in a different and more logical order, the terminology used will be less production oriented, and redundancies in the text will be eliminated. On the content side, the process logic will be even more pronounced, and there will be more emphasis on TQM principles (in addition to quality assurance).

At this stage, the requirements of the standards can be grouped into three sets:

- general requirements of a quality system (management responsibility, quality manual and procedures, appointment of a quality manager, availability of qualified resources and staff, ...);
- the need to maintain documented procedures on the key processes of the organisation (design, development, purchase, delivery, etc.) and implement activities according to the procedures;
- specific quality assurance mechanisms, including test and inspection, keeping quality records, dealing with non-conformities, keeping documents up-to-date, conducting internal audits and holding regular management reviews.

Some of these requirements can easily be implemented in a well run organisation. In an effective, high performance organisation, often the only thing required is to write down, in a formalized manner, the way one is currently operating. However, the specific quality assurance requirements almost inevitably require more work, including the introduction of new activities and processes (particularly document control and internal audits) which often did not exist before.

It should also be mentioned that the ISO 9000 standards are no easy reading. Especially some of the longer paragraphs, and the overlap between some sections, do not facilitate understanding. Moreover, the meaning of certain sentences may not be obvious in a particular context. Consequently, the standards *always* need some interpretation. This has both advantages and disadvantages. The main disadvantage is that it creates an additional hurdle and may reduce confidence in the development of the quality system; it may also lead to excessive control mechanisms. It puts the burden on the organisation of justifying its interpretation of the norm (hopefully, this report may relieve some of this). But this necessity of interpretation also has advantages. For instance, certain sections can be declared inapplicable, whilst for others it may be shown that the requirements are impossible to implement or even illegal, if taken to the letter. In such cases, it is important to demonstrate that, nevertheless, the 'spirit' of the standard is being followed as far as possible.

2.3.4 The certification process

Organisations can be *certified* by an independent body for compliance with either ISO 9001, 9002 or 9003. Several steps are involved in that process. Since education and training institutions are no different in this respect from any other type of organisation, and the implementation process is well described in the literature, only the main steps are highlighted here.

First, the most appropriate standard has to be chosen. Since few organisations get certified on the basis of ISO 9003, the real choice is often only between ISO 9001 and 9002. ISO 9001 is the most complete standard, covering most of the activities of organisations. ISO 9002 is almost identical with ISO 9001, except for the exclusion of design activities. Thus whether an organisation interested in certification should go for 9001 or 9002 really



depends on whether it has design and original development activities which are to be part of the quality system. For the education and training sector this implies that ISO 9002 is most suitable for primary and secondary schools (since the programme if often prescribed by the government), while ISO 9001 is more appropriate for higher education and continuing education and training providers (since they design their programmes themselves).

A second important initial consideration is the 'scope' of the certificate desired. Indeed, an organisation can be certified for only part of its activities (e.g. for all activities necessary to produce certain types of products and services). Larger organisations often start with a pilot department before they engage in full certification. Education institutions may wish to exclude ancillary activities from their quality system, because they do not see the added value – or do not have the resources – for setting up quality assurance mechanisms for such activities.

Once the appropriate standard and scope are selected, the organisation will gradually build up the quality system. This may take between 6 and 24 months, depending on factors like the complexity of the organisation, the availability of human resources, the use of external consultancy, and the number of components of the quality system which are already available. The first part of the work consists in establishing the quality and procedure manuals. Once this activity comes to an end, the 'systems' requirements of ISO 9001 need to be set in operation: internal audits, document control, management reviews, corrective and preventive action, etc.

Most time is likely to be spent on the writing of the quality and procedure manuals. Other time-consuming activities may include the analysis of the different processes, the standardisation of processes, the effective use of new procedures such as document control, internal audits and maintaining quality records, and the training of staff. The time involved in such activities varies greatly from one organisation to the other.

When the whole quality system is fully operational, the organisation should get in touch with an officially recognized certification body (in most European countries, there are several of such organisations). Each certification body uses a slightly different approach, but generally the stages are as follows:

- [sometimes] a pre-audit by the certification body, for identifying gaps in the quality system (such a pre-audit may be undertaken during the preparation process; it is not compulsory and may only be necessary if the organisation has not used a specialised consultant);
- submission of the quality manual and procedures to the certifying body for analysis of conformance with the selected standard (the certifying body will not go any further if it discovers at this stage any major non-compliance);
- audit of the organisation by the qualified auditors of the certifying body; the main purpose if to verify that the quality system is indeed implemented as described in the documentation, and is conform with the requirements of the standard:
- approval of the quality system and award of the certificate;
- interim audits by the certifying body, typically 3 or 4 times over a period of 3 years;
- after 3 years, a new full audit (possibly by another certifying body) is required.

The organisation which wants to be certified has to bear the full costs of the audit and certification; it actually enters into a three year contract with the certifying body. It goes without saying that the certifying body should be independent from the organisation and also from the consultants who may have assisted the organisation.



3. ISO 9000 in education and training

3.1 The rationale for adopting ISO 9000 in an education or training organisation

3.1.1 Main reasons for seeking certification

Improving or maintaining the quality of education or training provision is not the only reason why providers seek to be certified under ISO 9001 or ISO 9002 (ISO 9003 is not relevant). Most of these reasons are similar to those of other organisations. Overall, four types of arguments are put forward:

- promotion of a high quality *image*, high visibility and credibility;
- a response to external factors;
- develop a full quality assurance system;
- improvement of *specific activities* of the organisation.

The importance of these arguments is likely to vary strongly depending on the nature of the organisation and their external environment. Let us consider each of the main reasons in turn.

A quality image

Image is important for all types of education and training organisations, because it influences the numbers of pupils, students or trainees – and hence in the long term the survival of the organisation. This reasoning applies whether the organisation is a school, an adult training provider or an internal training department in a company. The image perceived by the outside world depends on the quality of services delivered, as well as on how effective that quality performance is communicated. Being certified under ISO 9001 or 9002 is an easily communicated signal to the outside world that the organisation is committed to quality and that it has been subject to independent external scrutiny.

Circumstances will determine whether ISO 9000 certification on such grounds is justified. The more competitive the environment is, and the more ISO 9000 is understood by the target group of the provider, the more an ISO 9000 label will support or reinforce the reputation. Some of the more specific image-related arguments for ISO 9000 which have been reported in the literature or have been communicated to the author are:

- the need to demonstrate or confirm a leading or pioneering role in a particular market sector (ISO 9000 as a differentiating factor);
- a way to make the quality performance of the organisation visible and easily appreciated by customers (who cannot assess the value of other labels or certificates which may only be known in the training world);
- the willingness of traditional, public education institutions to be considered as an effective organisation which is capable of meeting professional private sector standards (getting rid of a sloppy 'public sector' image) this applies in particular to public institutions operating in competitive training markets;
- a contribution to maintaining or improving market share;
- credibility of institutions and training providers which teach management, TQM, and ISO 9000;
- the need for a quality label which is easily recognized at international level.



External factors

External demands may not have been the prime reason for launching an ISO 9000 exercise, but they are likely to become the driving factor for certification over the next years. This is because the concern for quality and performance is growing in society, with all organisations being expected to become more accountable to the shareholders or the citizens – depending on whether they are private or public. In addition, clients of education increasingly desire 'proof' of the quality promised. Certification is a way of responding to such demands. It is admitted that we are still far away from a situation whereby an ISO 9000 certificate becomes an almost compulsory requirement. But many signs suggest that over the next years an increasing number of European education and training providers (or at least some of their departments) will face the pressure of certification. An example of such developments is the requirement of Vetron, a Dutch association of training providers, which has restricted membership since January 1997 to certified training organisations only.

Examples of the type of external factors which have already forced certain education and training organisations to start the certification process are:

- the explicit demand from a client organisation, often related to the requirements of that client's own quality assurance system (which requests quality guarantees from suppliers);
- the current or anticipated demand from governmental bodies (e.g. for public tendering);
- the implicit or announced expectations from major client organisations;
- the pressures because of certification of main competitors;
- the demand from a sectoral training organisation (entry request);
- for training departments in firms: the extension of existing certification from other departments (e.g. production); or a requirement by the mother company;
- a compulsory requirement for being granted the right to deliver certified or accredited training:
- a mechanism to avoid state regulation or accreditation (e.g. in continuing training or higher education).

In the private, commercial education and training sector, such external factors still lead to positive differentiation with competitors. However, over time, they will become 'negative' motivators: merely being certified will no longer provide any particular advantage, but not meeting the external demand might have drastic implications in terms of market share. It is as yet unclear whether such arguments might also apply to public institutions — since in any foreseeable scenario, the number of public education institutions who may become certified over the next decade will remain a small minority. One factor — not very probable though — which may spur ISO certification in this area, would be the desire for achieving some kind of European compatibility of quality management and quality assurance arrangements.

Quality assurance and internal organisation

An obvious argument for going the ISO 9000 route is the need and willingness to establish a quality assurance system which covers all critical areas of the organisation. One might have expected this to be the first and most important reason for certification, but so far the 'image' and 'external demands' arguments seem to prevail. An exception may have been the University of Wolverhampton (the first university to be certified), where the unifying and systemic arguments appear to have been the predominant driving force. But from the information gathered for this study, this does not seem to apply as yet to many



other organisations. Maybe this is a temporary situation: most of the education and training organisations which pioneered certification were convinced that they had a good system already – this may be less the case in the future.

If an education or training organisation is looking for a comprehensive quality assurance approach, then the principles and requirements of ISO 9001 or 9002 are to be considered seriously. Indeed, most traditional quality assurance methods in education and training are based on a classical typology of a provider who delivers particular courses and programmes to students. This facilitates the analogy with industrial production and mass services – and hence the quality philosophy adopted there. One might even argue that the ISO 9000 standards can more easily be applied to education and training activities than to other services, such as health care, personal care, consulting or research.

Additional, related arguments which have been put forward by certified education and training organisations are:

- ISO 9000 provides a visible, understandable and verifiable focus for the internal quality improvement efforts of an organisation;
- the perspective of a certificate and formal recognition of the efforts undertaken is important for the motivation of staff;
- unlike traditional quality approaches in education and training, the ISO 9000 standards consider education or training activities not as isolated processes, but in the context of the organisation's quality objectives:
- ISO 9000 obliges the organisation to formalise its quality level, and develop procedures to avoid deterioration in the future;
- the certificate also helps as regards internal recognition of quality efforts, and provides reassurance and pride about the current performance;
- the compulsory requirements avoid the use of a 'TQM à la carte' with no clear milestones;
- independent of the rhetoric on 'quality assurance', education and training organisations may just seek the benefit of being sure that their internal organisation becomes more effective, efficient and consistent;
- documenting and operating a quality system is seen as an excellent means for maintaining quality levels in organisations with a high turn-over of staff (e.g. some higher education institutes) or who use external teachers and trainers.

Specific quality improvements

Some organisations have sought to introduce ISO 9000 to improve specific functions or activities. This may be related to a general concern of ensuring high customer satisfaction, or to the willingness to attain specific quality levels. Some examples related to education and training institutions are:

- improvement of the logistical and support processes;
- better control of external trainers;
- development of a thorough procedure for design and development of training and education programmes and courses;
- quality assurance of the examination and evaluation of students.

Putting the arguments into context

The previous paragraphs listed many arguments which an education or training provider may examine when considering ISO 9000 certification. Although many items are listed, it should be pointed out that, in general, most certified education and training organisations singled out only a few of these. Indeed, the belief in the importance of only one or two



main reasons for certification, gives a clear focus and may hence yield the necessary motivation and commitment to bring the whole exercise to an end; this may not be achievable if the goal of certification is diluted across a wide range of objectives.

Several of the possible arguments in favour of ISO 9000 also apply to other quality approaches and certification, accreditation or recognition arrangements. There is no doubt that several of these approaches are better tailored to the education and training world. Why then, still opt for ISO 9000? The main arguments which have been put forward in this regard are:

- ISO 9000 has high visibility outside the education and training sector, and also at international level which a national quality scheme seldom has;
- much external expertise, advice and training on ISO 9000 is readily available; this is often lacking for other quality approaches;
- last but not least, ISO 9000 should not be seen as competing but rather complementing other quality control arrangements: existing standards, evaluations, documents and methodologies can relatively easily be integrated into a quality system that meets ISO 9000 requirements.

It should also be mentioned that the reasons for seeking certification in the education and training world, do not differ fundamentally from those of other organisations. An in-depth UK survey "ISO 9000 – Does it work?" conducted in 1995 by the Manchester Business School on behalf of SGS (see Bibliography), found eight reasons for seeking certification which were each listed by at least half of the respondents to the survey (in decreasing order of importance):

- future customers likely demand for ISO 9000;
- increase consistency of operations;
- maintain/improve market share;
- improve service quality;
- customer pressure;
- use as a good promotional tool;
- make operations more efficient;
- improve product quality.

The survey also found that "small companies principally sought the standard to improve market share and for promotional purposes. (...) The larger the organisation, the more likely it was to cite customer pressure as a reason for certification. The service sector emphasised the importance of increasing market share and the need to improve consistency of operations and quality of service (...)". These results, when interpreted in an education and training context, are consistent with our findings regarding education and training providers.

3.1.2 Problems and disadvantages

Arguments in favour of certification should, of course, be balanced against the counter-arguments and disadvantages. These are numerous as well and – in the light of the limited number of certified education or training institutes – still outweigh the positive arguments. Again, several groups of possible disadvantages can be listed:

- interpretation problems;
- insufficient relevance;
- inappropriate standardisation;
- time consumption and cost;
- specific problems.



Interpretation problems

Application of the ISO 9000 standards is not straightforward. The requirements and even the underlying concepts have to be 'translated' or 'interpreted' into a language which an education or training provider can understand. It is this very problem which was one of the main reasons for writing this report.

Since Section 3.2 and the whole of Chapter 4 will cover this issue in depth, only the main categories of interpretation problems are listed here:

- the terminology used in the standards;
- the sequence and interrelationships of the different clauses;
- the methods and extent of quality assurance required;
- the product definition to be used;
- the mapping of the education and training processes onto the requirements of the standard.

Thus, anyone in education and training interested in ISO 9000 needs to take this first, considerable hurdle. There are many who have stopped considering using ISO 9000 after reading the norms, wondering about how these could be applicable to their situation. While the overall philosophy behind the norms may become clear after some time, the interpretation of some clauses might remain obscure. Even 'specialists' – i.e. people and organisations who have published interpretations or guidelines on ISO 9000 in education and training – do not agree entirely on all interpretation issues.

Insufficient relevance

Many education or training experts, particularly from the public education sector, criticize ISO 9000 at its core, as being of insufficient relevance for education and training institutions. It is true that the norms were neither designed nor optimised with education or training institutions in mind. The following fundamental arguments are often heard:

- there is no obligation in the ISO 9000 norms to adhere to any minimum standards; in fact, organisations set themselves the quality levels for which they are satisfied (with the risk, of course, that they set low standards);
- ISO 9000 is not the best way towards TQM in an education and training institute; there are other approaches which facilitate the introduction of TQM;
- many critical aspects of education and training institutions and their services are not explicitly listed in the standard (and may hence not be covered);
- the complexity and multiplicity of objectives and purposes of education do not fit very well with the standardisation resulting from ISO 9000;
- ISO 9000 imposes mechanisms on education and training institutions (e.g. document control, quality records,...) with little or no added value.

It would lead us too far here to discuss each of these criticisms in depth. There is some truth in all of them, but some objections are based on a misconception of the purpose and usage of the standards. Whether the risks pointed out might materialize or not, will depend on how the quality system is implemented. As of today, and in as far as the available information makes a full assessment possible, none of these potential weaknesses seems to have occurred with the first education and training organisations which have been certified. For instance, none of these institutions has set low standards. It is recalled that certified organisations will set their standards at the level required by the clients, and that the dynamic of the quality system (with its audits and corrective measures) will ensure that standards are set at an appropriate and sustainable level.



However, when ISO 9000 spreads across an increasing number of education and training organisations, the risk will become real that inappropriate and insufficiently relevant quality systems might be created.

Inappropriate standardisation

Since the ISO 9000 standards were originally conceived for medium-sized to large companies in the manufacturing sector, it is legitimate to query its appropriateness for education and training organisations, which are in many ways similar to small SMEs from the service sector (the main exception to this analogy are higher education institutions, but even these are service organisations).

Because of their less standardized operations, service organisations encounter problems in meeting the ISO 9000 requirements, when these are taken literally. Every service operation is unique and no service activity can be 100% controlled because of the involvement of external, unpredictable factors – in the first place the reaction of the customer, who is in general actively involved in the delivery of the service itself. This is quite obvious in an education and training environment, where it is well known that the quality of teaching and learning depends to a large extent on the interest, commitment and initial qualifications of the learners.

When writing procedures for education and training organisations, it is often not straightforward to find the appropriate level of abstraction. On the one hand, procedures should demonstrate compliance with the requirements of the Standard, but on the other hand they should allow for a certain level of variability. Education and training institutions should avoid excessive paperwork and prescriptive rules – but at the same time they ought to be able to control the quality of delivery. For that reason, they need to clarify for themselves what are the critical aspects are of their delivery – which need rigorous control – and those elements where more freedom and variability will not lower quality standards, or are indeed a real necessity for assuring quality all the time. This is even more important since the staff involved has in general high qualifications, and excessive prescription and standardisation might be neither desirable nor productive.

When there is a lack of understanding of the desirable degree of standardisation, there is a serious risk of creating a bureaucracy with a focus on procedures, records and control. Education and training organisations should therefore seek appropriate advice in order to minimise such risks.

Time consumption and cost

The cost and time-consuming nature of the implementation of ISO 9000 requirements are probably the most serious obstacle to the generalised use of the standards in education and training institutions – and this is likely to remain so for the coming years. There is always an important direct cost, consisting of the cost of certification. In most cases, there are additional expenses for guidance and training by specialised consultants. Both cost factors vary in relation to different parameters, such as:

- the size of the organisation;
- the number of fundamentally different services and their underlying processes (this has an impact on the number of procedures);
- the 'quality culture' which prevails;
- the appropriateness and completeness of existing documentation (procedures, forms, checklists,...);
- the existing evaluation and monitoring arrangements;
- the experience with quality assurance.



The cost of certification is, in general, mainly related to the size of the organisation and the number of procedures. The extent of need for training and external guidance will, however, depend on all the factors mentioned. Typical cost ranges (excluding very large or very small organisations) for certification are in the range of 3 000 to 10 000 ECU, while consultancy (training, advice, pre-audits,...) may vary from 2 000 to 20 000 ECU. To such figures should be added the salary costs and overheads of the staff working internally on the set-up of the system. Again this will vary considerably, but it is likely to represent the equivalent of between 4 manmonths (for a small, very well and focused training organisation) and several manyears. Typical implementation schedules – from initial decision to certification – last between 9 and 24 months.

Once certification is obtained, there is also the cost of the 3-4 interim-audits over a three year period (1 000 to 2 000 ECU/audit) and the maintenance of the system – which may require about 30% of the staff resources needed for the initial implementation.

Obviously, ISO 9000 comes at a price. This may be an insurmountable barrier, particular for small training providers and public education institutions. Few of these seem to be willing to take such a financial risk, unless they feel strong pressure to do so. Nevertheless, the cost issue should also be put in its proper context:

- other quality assurance arrangements (e.g. government based inspections) or Total Quality Management approaches also require considerable resources – with a cost structure which is often hidden (e.g. the real cost for inspecting a school may be higher than that for an ISO 9000 audit);
- expenditure for training, advice and pre-audits should diminish over time, as more experience becomes available;
- the cost of certification should be considered as a long term investment, to be depreciated over several years;
- while cost is the most visible barrier, experience to date suggest that the time consumption is actually the biggest obstacle: many organisations cannot release full-time staff for such an exercise, which implies that implementation happens in phases and is often interrupted because of other priorities.

Specific problems

Some other problems which have been reported include:

- the certification process does not lead to the publication of a public audit report governments are therefore reluctant to encourage the use of the system in public institutions for accountability purposes;
- the audit rhythm which follows the certification may fit badly with the annual school cycle or the academic year;
- the length of the implementation process up to two years requires a formidable commitment from both management and staff in particular since it is almost certain that unexpected problems will arise and that the external environment might change;
- cost and cost-effectiveness issues are not considered in ISO 9000.

Overall assessment of the importance of problems and disadvantages

Some of the above mentioned problems are likely to continue to persist in the future – unless the standards themselves change. Thus, it must be recognised that the ISO 9000 approach has some inherent weaknesses for education and training, which require skill and creativity to address. The cost and time implications are a real hurdle, and the risk of creating a bureaucratic system is always present. Besides, in the author's opinion there seems to be no justification for the objection, expressed by some education and training experts, that ISO 9000 would be fundamentally inappropriate for education and training



institutions. The reality is that most institutes have not yet reached a level of quality assurance and control which would make the implementation of the standard feasible.

By way of comparison, we list the most important problems with certification as reported by the already mentioned survey by the Manchester Business School:

- time required to write the manual;
- high volume of paperwork;
- high cost of implementation;
- time required to complete implementation;
- high cost of maintaining the standard;
- lack of free advice:
- lack of consistency between auditors;
- time spent checking paperwork prior to audits.

Only the first of these drawbacks was mentioned by over 30% of the respondents; the last item by 16%. The survey concluded on this point that "The high cost of implementation – in terms of time, volume of paperwork and money – were seen as the major problems related to ISO 9000 across all groups. Small organisations generally considered drawbacks to be more significant, when compared to the benefits, than did large organisations. The same pattern was seen in concerns with ongoing maintenance of the standard (...)".

3.1.3 Differences among education and training organisations

For practical reasons, the previous paragraphs did not list the advantages and draw-backs for each different type of education or training provider. In a European report like this one, it would have proven to be an impossible exercise anyway, since categories of education and training organisations display different characteristics across countries. However, it is useful to compare briefly the relevance and applicability of ISO 9000 by grouping education and training providers in different ways:

- Compared to the suppliers of 'initial' education (schools and higher education institutions), the providers of continuing education and training (CET) are more likely candidates for ISO 9000. This is because CET provision is much more subject to competition and the laws of supply and demand than initial education. The first education and training institutions to be certified and by far still the majority were all CET providers (including training departments of international concerns, e.g. IBM Germany).
- When comparing 'general' education with vocational education and training (VET), it is obvious that VET providers are more suitable candidates for ISO 9000. Indeed, VET institutions whether public or private need to have a clear client focus (the professions and sectors for which they educate) and they are more subject to external pressures regarding the relevance of what they offer. It should be no surprise that the first ISO 9000 certified schools in the different European countries were technical and vocational schools [e.g. Sandwell College (UK), Hillerød Technical College (DK), Technisch Instituut Glorieux (BE)].
- ISO 9000 is more likely to be appropriate for 'larger' than for 'small' institutes. The main arguments are that the larger education and training organisations: may have sufficient resources available, may already have or need a lot of standardized operations, can benefit most from a TQM approach for internal efficiency improvement, and are forced to have a medium and long term perspective. Often, it is the larger organisations in their sector who are the first to obtain a certificate [e.g. DIEU (DK), Technische Akademie Esslingen (DE), NCATB-Elsevier Opleidingen (NL)].



- The more varied and customised the provision of education and training is, to more time it will take to obtain an ISO 9000 certificate simply because the volume of documented procedures per person is much higher. Thus, organisations who only provide training "à la carte" face more hurdles than institutes using essentially the same approach for most of their education and training delivery. Thus, surprisingly maybe, while ISO 9000 standards may be less relevant and needed for a primary school than for a provider of customised training, the school would proceed much faster in implementing the quality system requirements in particular when it operates in a heavily controlled environment. In many schools much of the required quality documentation already exists, but lacks consistency and control.
- As to the higher education sector, it is difficult to identify clear patterns. Only a few full higher education institutions have been certified (University of Wolverhampton (UK), Luchtvaarthogeschool (NL), EHSAL (BE), ESIG (CH),...). There have also been some cases where units and departments of universities were certified typically in the management, engineering, facility management or continuing education areas [e.g. Department Industrieel Beleid of the KULeuven (BE)]. It seems likely that the number of such certified departments and units will increase over the next years, but whether this trend might also apply to full universities and other higher education institutions remains to be seen. ISO 9000 is met with much scepticism in academic circles, and the standards also face competition with existing or new audit, quality assurance or accreditation arrangements.
- A special category which merits attention are the training providers which depend directly of other organisations, such as training departments or semi-independent training units of large firms. Here we can see a mixed bag of developments. In some cases such training centres have taken the lead in ISO 9000 (e.g. the training department of the Dutch Railways), but in other cases they have followed, with some delay, the corporate company policy on ISO 9000 certification (e.g. different departments of CESI in France). Whether it makes sense to certify separately the training departments within companies, will to a large extent depend on their operational, financial and decision-making autonomy. If autonomy and scope are limited, then the quality system of the training department ought to be integrated in that of the firm as a whole.

This brief overview underlines the fact that the choice for ISO 9000 is not just a matter of balancing potential benefits and drawbacks. A decision also depends to a large extent on the particular environment in which the organisation operates.

3.1.4 Conclusion

The ISO 9001 and 9002 standards are suitable frameworks for the implementation of a quality assurance system with providers of education and training. The tangible and often compulsory requirements (quality policy, quality manual and procedures, regular audits,...) provide an overall, measurable framework for quality efforts, as opposed to less systematic and more qualitative improvement approaches. The main advantages of the ISO 9000 arrangements – compared to other quality assurance approaches are that:

- ISO 9000 certification combines four aspects at once: a comprehensive quality system, a credible assessment of it, formal certification and recognition, and a guarantee of ongoing maintenance and improvement;
- the value of ISO 9000 is appreciated by the different customers of education and training;
- ISO 9000 is easily recognized internationally.



There is, at present, no other international system with the same characteristics. At the same time, it must be recognised that the ISO 9000 norms are certainly not the best imaginable quality standards for education and training, and need to be complemented by content-related criteria. But the biggest drawback remains the considerable cost – in terms of both direct expenditure and input of human resources – of implementation. As of today, only education and training organisations with high quality standards and sound financial management can safely engage in ISO 9000 certification.

3.2 Terminology and interpretation issues

3.2.1 Main interpretation differences of the ISO 9001 requirements

The requirements of ISO 9000 always need some interpretation. In most service sectors, many of the specifications laid down in the standards need careful analysis and adequate interpretation before they can be applied in a particular education or training context. This interpretation relates to both the terminology — which is relatively easy — and to the processes concerned — which is often not straightforward. Also, the order of the different paragraphs will appear 'unnatural' or even 'illogical' to many people from the education and training world.

This section highlights some general interpretation issues, preparing the ground for the more detailed discussion of the clauses of ISO 9001 which is provided in Chapter 4. When interpreting the standard, the author had to make some choices. These are believed to reflect the most logical and consistent approach for a wide variety of education and training environments. It is recognised, however, that for certain aspects other interpretations can be defended — or that in a particular context another interpretation might be more suitable.

Maybe the most fundamental difference in interpretations of ISO 9000 standards for education and training is the definition of the 'product': is it the 'learning' or the 'learning process', or rather the education or training programme which is offered? This is not just an academic problem, but one which has implications throughout the standard. Based on comparisons with other service sectors, the real difficulty of controlling the learning process, and the choices made by many certified education and training organisations across Europe, the author has opted for the 'product' as being "the education and training services offered by the organisation, including associated products, tools and services".

This choice has many implications when interpreting a number of Clauses of ISO 9001. For instance, when 'learning' is taken as the 'product', then the 'testing and inspection' requirements concern assessment and evaluation of students and trainees. However, when 'the course (programme)' or 'training' is considered as the product — as is done in this report — then the 'testing and inspection' requirements refer to the evaluation of a course or training session by students, trainees and/or their employers.

The main interpretation differences – as derived from an analysis of available documents – are summarised in the next table; the column 'Interpretation adopted' is the one that is the basis for comment in Chapter 4. The full titles and contents of the clauses can be found in Chapter 4 as well.



Clause	Interpretation adopted	Other possible interpretations
General	Product = the education and training services offered, including associated tools and services	Product = the acquisition of learning Product = the student, trainee, pupil Product = the degree, qualification obtained
4.3 (Contract)	Contract review only covers agreements with customers	Contract review also covers contracts with suppliers, sub-contractors or external trainers
4.4 (Design)	Design does not include (or only marginally) development of education and training (designs stops at the specification of programme content)	Design includes substantial elements of development, up to close before delivery stage
4.4 (Design)	Design validation of a course happens just before it is released	Validation of the course design happens after the first run of the course
4.5 (Document)	External documents need minimal document control, unless when they critically affect the quality of the education or training	Most external documents should be con- trolled much the same way as internal documents
4.6, 4.10 + others	Part-time teachers should be considered as other staff (unless their input is very limited or irregular)	Part-time teachers are to be considered as sub-contractors
4.7 (Customer's product)	Supervision of students, welfare, care is not to be considered under this Clause (where relevant, this falls under Clause 4.9)	Supervision of students, welfare, care should be considered under this Clause
4.8 (Identification)	Identification and traceability refers to the education and training services	Identification and traceability applies to the student/learner
4.9, 4.11 (Calibration)	Calibration is only required for the equip- ment of which the accuracy is critical for the learning process, and for any equip- ment needed for testing the performance of trainees	Calibration is needed for all equipment used for the purpose of developing and delivering education or training
4.10 (Inspection)	Test and inspection concerns the monitoring and evaluation of the education and training services offered, including tangible items produced for it (assessment of students is covered under 4.9)	Test and inspection cover all aspects of assessment which take place, including those concerning students. Test and inspection only refer to the tangible items used for the education and training service
4.10 (Inspection)	Receiving inspection is impossible for services; these aspects may only be applicable for certain goods purchased	Receiving inspection and test cover the need to establish the level and needs of incoming students
4.12 (Test status)	Inspection and test status refers to the stage at which education/training programmes (and all their components) are	Inspection and test status only applies to equipment Inspection and test status also applies to learners
4.13 (Noncon- formities)	The definition of non-conforming 'product' should be extended to include the services as well	The control of non-conforming products only refers to faulty tangible items
4.15 (Handling)	Only applicable to tangible goods (in as far as these are used)	This clause should also be interpreted for services, and thus cover items like student care, counselling etc.
4.20 (Statistics)	For simple statistics (sums, averages, tables) evidence of the correctness of the figures is OK; for more advanced computations evidence is needed of understanding of the techniques adopted	This clause applies to all stages where at least sums or averages are computed



At first sight, this list may appear excessively long. Surprisingly however, many of these interpretation differences do not necessarily lead to conisderable variations in the implementation of quality systems. This is a direct result of the overlap and redundancy of the requirements of ISO 9001. This ensures that certain quality assurance arrangements which are, in a particular interpretation, not covered by a specific clause, need to be undertaken anyway because of the requirements of another clause. In particular the general clauses 4.2 (Quality system) and 4.9 (Process control) are formulated in such a way that quality arrangements must be in place for *all* critical processes, whether these are covered by a specific clause of the norm or not.

There is a second, more challenging type of interpretation problem. This concerns the assessment of how the requirements can be addressed effectively with minimum overhead. Indeed, in many situations it is often not straightforward to say whether or not the requirement of ISO 9001 is entirely fulfilled. Consider, for instance, the need to define and analyse 'design input factors' during the design process (Clause 4.4). It will be a matter of judgement by the institution and the auditor of the certifying body to decide whether all critical input factors are being considered. Another example: the requirements for purchasing goods and services from suppliers, where the organisation has to decide on the appropriate level of quality control. It is not obvious on what basis such specifications are acceptable or not. A third example: are slides and overhead transparencies used by trainers to be considered as 'controlled documents' or not?

The answer to such questions has major consequences for the implementation and maintenance of the quality system. The ISO 9000 standards contain many parts which need subjective assessment for education and training providers. Chapter 4 contains guidelines about what is necessary or desirable for most types of institutions, but it is stressed that every case is different and needs its specific interpretation. This is, actually, both a strength and a weakness of ISO 9001.

A third, related type of interpretation difficulty concerns the rigour and extent to which the requirements have to be followed, such as:

- the level of detail needed for documents (particularly procedures and work instructions) which has considerable implications for document control;
- the nature and quantity of quality records often the biggest stumbling block in the effective maintenance of the quality system, and the seeds for a bureaucratic, paper based system;
- the specificity of the quality policy and objectives;
- the frequency of internal audits and management reviews;
- the scientific validity of the evaluation and assessment methods used.

No authorative guidance exists on such issues. Some auditors, consultants and quality managers go much further than others in these matters. It is risky to generalize here, since what needs to be done depends on the complexity of the organisation, the demands from customers, and the educational attainment of the staff. In the report, a plea is made for minimal bureaucracy, but one ought to check that the certifying body agrees with the interpretation adopted.

3.2.2 Using ISO 9000 terminology in an education and training context

In line with the interpretation choices made in this report, it is possible to draw a table which 'translates' some of the key terms found in the ISO 9000 standards into more common education and training terminology.



ISO 9000 term	Interpretation for an education or training organisation
Supplier	Developer or provider of education or training
Customers	Students/pupils for education; employers or trainees for training
Product	Course, education or training programme; education or training material
Executive Management	General Director/Headmaster/Rector/Committee of Directors/
Contract	All types agreements with customers – also implicit ones – from the registration of students for a course, to a negotiated agreement for custom-made training
Design	Definition of education/training specifications; design of course programmes and curricula; specification of the content of education/training material; design of assessment and evaluation instruments for students/pupils/trainees
Purchasing	Acquisition of necessary goods and services, including the use of temporary, free-lance or external teachers/trainers/lecturers
Processes	Refer to the development, planning, delivery of education and training, including any assessment of students/pupils/trainees
Inspection and test	The assessment and evaluation of the courses/programmes/ materials by the students/pupils/trainees/employers
Calibration	Refers mainly to the assessment of the validity of the assessment and evaluation tools used
Nonconformities	Any problem occurring during the development and delivery of education and training
Training	Training and professional development of the staff of the education or training organisation

The next chapter provides more details on these terms. It should be noted that several words used in ISO 9001, e.g. 'procedure', 'document and data control', 'internal quality audits', or 'quality records' have a particular meaning in the Standard. For that reason, they are not included in the Table, but their meaning will become clear in Chapter 4.

Finally, a terminology problem is recalled which is often neglected: the translation of the norms into the different European languages. Although these translations are authorized, they have often been made with a particular type of organisation in mind. This translation may, in turn, facilitate or not the 'interpretation' for education and training purposes. Some key terms of ISO 9000 are actually not easy to translate, such as:

■ management	■ review	■ purchasing
■ control	■ contract	■ data
■ procedure	■ design	■ inspection and testing
■ audit	■ system	■ training
■ records	■ verification	■ servicing

The author of this report has encountered cases were the translation of the terms 'inspection and testing' led people to believe that the clause concerned the testing of students or an external inspectorate system.



3.2.3 Rearrangement of the requirements of ISO 9001

When reading the different requirements of ISO 9001 or 9002, one is likely to be puzzled by the particular sequence of the different requirements (which are grouped in 20 sections or Clauses). The order does not appear logical – which presents an additional hurdle for education and training organisations in finding their way through the norm. Moreover, the length of the clauses varies considerably from a few lines to a whole page. Some clauses have been subdivided in many sub-clauses, while others consist of a single paragraph. Surprisingly, moreover, the length of the Clauses or the number of subclauses are not correlated with their importance – in particular not in an education and training context. The varying degree of detail in the specifications and the overlap between certain sections may further add to the confusion of those who are not yet familiar with the standards.

One way to resolve this problem, is to read and analyse the requirements of the standards in a different sequence. For education and training organisation, this will be particularly useful if they read at the same time the interpretation provided in Chapter 4 (since that chapter was meant as a reference document, the sequence of the different clauses and sub-clauses has been rigorously followed). The next paragraphs suggest a different sequence, which may be more appropriate for education and training organisations. The titles used are those of the norms. The figures (e.g. 4.2) refer to the corresponding (sub-)clause.

General principles of the quality system

- 4.2.1 Quality system general
- 4.1.1 Quality policy
- 4.1.2.3 Management representative
- 4.1.2.2 Resources
- 4.1.2.1 Responsibility and authority

These clauses list the general requirements of a quality system, including a quality manual, a clear quality policy, the appointment of a quality manager, allocation of sufficient resources and the definition of responsibility and authority of staff.

General requirement for the implementation of the quality system

- 4.2.2 Quality system procedures
- 4.2.3 Quality planning
- 4.18 Training
- 4.5 Document and data control
- 4.6 Control of quality records

These clauses specify the general requirements for the implementation of the quality system: document processes in procedures, undertake quality planning where required, make sure that all staff is appropriately trained, and control documents, data and records.

General principles of process control

- 4.9 Process control
- 4.10 Inspection and testing
- 4.20 Statistical techniques
- 4.7 Control of customer-supplied product
- 4.8 Product identification and traceability
- 4.12 Inspection and test status



These are general quality assurance requirements which apply to all critical processes, in order to ensure that these processes are under control and will yield the required quality outputs. These requirements are stated in fairly general terms; some aspects are not applicable to education and training. For several processes (see below), there are more specific additional requirements

Specific requirements for particular processes and activities

- 4.3 Contract review
- 4.4 Design control (does only apply to ISO 9001)
- 4.6 Purchasing
- 4.11 Control of inspection, measuring and test equipment
- 4.15 Handling, storage, packaging, preservation and delivery
- 4.19 Servicing

These clauses elaborate, for some processes, further the general quality assurance requirements. The applicability of the different paragraphs varies considerably (see Chapter 4 for more details).

Maintaining and improving the quality system

- 4.13 Control of nonconforming product
- 4.14 Corrective and preventive action
- 4.17 Internal quality audits
- 4.1.3 Management review

Together, these clauses create a feedback loop which ensures that problems are solved, causes of problems are removed, improvements take place continuously, and the whole quality system is regularly and thoroughly reviewed.

Reading the interpretations of the clauses of ISO 9001 in this sequence might help an education or training organisation to understand better the real implications and requirements of the standard.



4. Interpretation of ISO 9001 requirements for education and training organisations

Introduction

This chapter is the core of this report. It contains a detailed discussion of all requirements of ISO 9001 and their interpretation in an education and training context. Since ISO 9002 is a subset of ISO 9001, no separate discussion is provided on ISO 9002. An attempt has been made to pay attention to all types of education and training organisations – from schools, through higher education institutions, to providers of continuing education and training. Thus, for any particular organisation, some parts of the discussion will not be applicable; in general this will be obvious from the text.

An important limitation of the chapter should be mentioned. Many education and training organisations undertake activities which may not be directly linked to education, such as child care, accommodation of students, consultancy, research, etc. Although such activities are occasionally mentioned, no specific interpretation or guidance is provided on such matters.

The starting base for writing this interpretation was the author's own expertise in assisting education and training organisations with regard to ISO 9000. This was considerably enriched through the analysis of a large number of documents from all over Europe, produced by standards organisations, certifying bodies, researchers and certified providers of education and training themselves (see Bibliography). The author was moreover given the possibility to analyse internal documents of certified organisations and certifying bodies.

For some general aspects of interpretation and the terminology used, please see Section 3.2. Please note that certain concepts, such as 'customer' need to be interpreted differently for different types of organisations. In order to avoid continuous reference to "ISO 9001 and ISO 9002", the text uses most of the time the term 'the Standard'. The word "Clause" is used to refer to one of the twenty numbered requirements of the Standard, or sometimes to a well-defined subsection of one of the requirements. Although it may appear cumbersome, the term 'education or training organisation' is used fairly systematically – it may refer to a school, a university, an adult education institute, a training department within a firm, a private training provider, etc. Only when no misunder-standing may occur, more specific terms (e.g. 'school') have been used.

In the text, the interpretation choices which were mentioned in Section 3.2 have been followed consistently. Occasionally, however, some reference is made, *in italicised text*, to the implications of other possible interpretations.

A final warning when using these guidelines: every organisation has different characteristics – and thus the interpretation of the ISO 9001 requirements will need to follow suit. There are also many different ways in which the requirements can be met. Any quality system is unique, and the guidelines which follow should never be followed blindly. There is no substitute for a proper analysis of the activities and operation, before any ISO 9000 exercise takes off.



39 - 4 4

4.1 Management responsibility

4.1.1 Quality Policy

The education or training organisation must have a Quality Policy statement, in order to comply with the requirements of this first clause of the Standard. The statement should clearly demonstrate the organisation's commitment to quality towards its customers. The quality policy of publicly regulated education institutions should also be relevant to the legal, social and national context.

The quality policy should include or refer to specific quality objectives to which the organisation is committed. Such objectives should be relevant to the needs and expectations of customers (employers, students, trainees,...). They should give substance to the main messages of the overall quality policy, which is often phrased in general terms (e.g. 'We provide high quality education to students'. 'We deliver an excellent service to all our clients'. 'Our mission is to develop customized training which meets the needs of our customers'). Some examples of specific quality objectives of an education and training organisation are:

- 90% of trainees are satisfied or very satisfied with the training courses;
- all governmental standards and regulations are met or exceeded;
- failure rates of students will be below 15% by a certain date;
- success in obtaining a particular certificate, prize, award,...

The quality objectives can also be related to organisation and performance of the quality assurance system itself, for instance:

- all agreed deadlines are met;
- each process is subject to at least one internal audit per year;
- all course programmes are reviewed every three years by independent experts;
- all trainers have certain content-related and didactic qualifications:
- verifiable learning outcomes will exist for all courses by a certain date;
- all criteria for a quality prize or award will be met by a given date.

When defining quality objectives for substantiating the overall quality policy, one may also identify goals which implicitly refer to the achievement of a particular quality level, such as: increasing market share, low turn-over of staff, the amount of internal training, commitment of parents and students, spontaneous congratulations, winning of tenders, employability of graduates, etc.

The definition of the Quality policy is the responsibility of the senior manager(s) with "executive responsibility" – what is meant is the chief executive/head teacher/ director/general manager/rector etc. This responsibility should not be delegated. The quality policy should be issued, communicated and made known throughout the organisation. Indeed, the Standard requires this policy to be "understood, implemented and maintained at all levels". The quality policy and related objectives should be included in the Quality Manual and should be publicly accessible (e.g. on public display). In larger organisations, middle management has an important role in communicating and supporting the quality policy. All full- and part-time staff (including external experts and free-lance trainers) involved in both educational and support activities should receive the same quality message from management. Sub-contracts with third parties who may affect the quality of the service (in particular external teachers and trainers) should refer to the quality policy of the education or training institution.



It must be possible to show how the quality system of the organisation is consistent with the Quality Policy and the stated quality objectives and intents. Moreover, all statements relating to quality which are given in documents, training manuals, promotional material, etc. must be consistent with the quality policy.

There should be evidence of the intent to deliver against these objectives. This could be done by establishing an annual action plan for quality improvement which is linked to the Quality Policy and its specific objectives. Larger organisations need to have such plans and goals for each major department.

As the aims of the organisation change, and needs and expectations of customers evolve, the quality policy and objectives will require adaptation. These changes and the necessary implications for the quality system should be examined at management reviews (Clause 4.1.3) and documented. Changes must be issued in a controlled manner (see Clause 4.5) and properly communicated to all employees.

4.1.2 Organisation

4.1.2.1 Responsibility and authority

The responsibility and authority of all staff who can influence quality should be documented. In most organisations, including in the education and training sector, all staff have an effect on the quality of the service (including part-time or external teachers and trainers, support staff and subcontract staff). It is up to the management of the organisation to decide on the responsibility and authority for particular tasks and duties.

The requirement of this clause may be fulfilled by including a full organisational chart in the quality manual (Clause 4.2.1), complemented by other statements in the quality or procedure manuals (e.g. task allocations as part of procedures). The quality system documentation could also include references to existing descriptions of duties (e.g. in staff contracts or public regulations). Sometimes, responsibility might need to be assigned to a team, committee or council (such arrangements are often found in higher education). Descriptions of responsibility and authority should be clear, complete, and consistent across the organisation.

These documents, charts and descriptions of duties are to be considered as 'controlled documents' (see Clause 4.4). This implies that they should always be up-to-date, authorised and available to those who need access to them.

Special care should be given to defining the responsibility and authority for critical quality system aspects, particularly:

- the management review (Clause 4.1.3);
- the management function responsible for Quality (Clause 4.1.2.3);
- the responsibilities and authority for the maintenance of the Quality system (several Clauses);
- inspection and verification (Clause 4.10 and others);
- quality system non-conformances (several Clauses);
- corrective and preventative actions (Clause 4.14);
- quality records (Clause 4.16 and others);
- internal quality audits (Clause 4.17).

In an effective quality system, defining responsibilities for such activities is in general fairly straightforward.



4.1.2.2 Resources

This clause expects an education or training organisation to ensure that adequate resources are provided to allow the effective and efficient operation of the quality system. With 'resources' are meant: people, facilities, equipment, education and training materials, supplies, services, etc. This requirement covers all management, work performance, and verification activities which are part of the quality system. All resources needed must be adequately identified; this demand is restated for several specific processes in other clauses of the Standard.

In education and training, particular attention should be paid to the adequate definition of the necessary competences of teachers/trainers (see Clause 4.18). All staff need to be appropriately qualified for the tasks for which they are responsible. The Standard, in addition, underlines this need for those people undertaking the critical functions of verification of the quality of work and the internal auditing of the quality system (Clause 4.17). Where a division of responsibility exists between the performance of tasks and the assessment or verification of these tasks, this should be clear, and both functions should be appropriately resourced.

Appropriate training should be provided to ensure people are and remain qualified for their jobs (this applies in particular for verification and auditing functions). Training records should be maintained to provide evidence that suitable training has been provided (see Clause 4.18). Such training records are to be processed as Quality Records (see Clause 4.16).

4.1.2.3 Management representative

The education or training organisation needs to identify a member of the management team as the person who has a clearly defined responsibility and authority for implementing and maintaining the requirements of the Standard. This person, referred to in the Standard as the 'management representative', may be called 'Quality Manager' or 'Quality Coordinator'. Responsibility and authority must be fully documented. He or she may have other responsibilities dependent on the size of the organisation and the complexity of the quality system. In larger organisations, it may be useful to appoint a deputy quality manager, and to create a 'quality management unit' headed by the quality manager. In very small organisations, the quality manager is often the director/head of the institution.

Formal reporting mechanisms should exist between the Quality Manager and the management team regarding system, process or service failures. The Standard also suggests that the quality manager be a liaison person with external parties. He or she thus becomes the focal point for the quality system, and has a central role in the continuous improvement of the system. It is recalled, however, that all staff bear responsibility for the quality of their own activities, and that the ultimate responsibility for quality and performance is with the senior manager.

4.1.3 Management review

According to this clause, senior managers have to assess regularly the suitability and effectiveness of the quality system, so that they may identify areas where improvements can be made. The review process, called the 'management review', should be described in the quality manual and take place at defined intervals (at least once a year). It should



ensure that the quality system continues to meet the needs of the organisation, its customers, and the requirements of the Standard. Other staff may attend this review meeting when appropriate.

The management review is in general based on an overall report of the quality manager and may include such items on the agenda as:

- an analysis of the quality policy and its implementation;
- the state of development of the quality system;
- the results of internal quality audits;
- an analysis of customer complaints:
- the status of documented systems;
- outcomes of evaluations and customer satisfaction surveys;
- the effectiveness of corrective and preventive measures taken;
- evaluation of improvement programmes.

The review must at least consider the quality policy and the key characteristics of the quality system, and determine whether quality objectives are being achieved. This requires the preliminary specification of objective targets for quality system improvements (e.g. the specific quality objectives mentioned in Section 4.1.1) and assessment at a later stage if these have been achieved.

Evidence should be available that reviews are planned and effective. Minutes of management review meetings must be maintained and be treated as quality records (Clause 4.16). Those records should show what decisions were made, what preventive, corrective or other follow-up action were to be undertaken, and who was responsible for their implementation. There should, of course, also be evidence that these decisions have effectively been implemented and/or re-assessed in a subsequent management review.

4.2 Quality system

4.2.1 General

A quality system consists of a coherent set of standards, procedures, activities, feedback mechanisms and management actions which together ensure the quality of the products and services delivered. Since most activities in an organisation, particularly in education and training, influence quality in one way or another, the quality system will need to encompass virtually all activities and processes of an organisation (with the possible exception of financial tasks, certain marketing and public relation activities, part of internal information and communication, personnel administration, environmental measures, etc.).

It is useful to recall here that an organisation can be certified for part of its services (e.g. a consultancy office only for its training activities; or a school only for its regular education programmes). In such cases, the quality system will only need to cover those activities, functions and factors which could affect the quality of the education and training for which certification is sought.

Not any type of system of work organisation is a quality system; to qualify as a 'quality system' under ISO 9000 all requirements of the Standard need to be fulfilled. Clause 4.2 lists some of the main principles; more explicit requirements are developed in the other clauses. Overall, the clauses provide a framework for a quality system (e.g. with a quality policy, documented procedures, internal audits, quality records, etc.), with strong empha-



sis on effective quality assurance mechanisms. The detailed implementation of the quality system is not prescribed in the Standard. There are, for instance, no detailed instructions on how procedures have to be written, on the format of quality records, or on the frequency of internal audits. What matters is that the quality system is appropriate to the type and size of the organisation.

The establishment of a quality manual is a mandatory requirement of the Standard. The manual should include at least:

- the organisation's quality policy and objectives;
- a short description of the organisation and its activities;
- the main responsibilities and functions of staff;
- an outline of the structure of the quality system documentation;
- a cross-reference between the procedures (and other elements of the quality documentation) and the requirements of the Standard.

The relation between the organisation's policies and the documented procedures should also be obvious from the quality manual.

In a small education and training organisation it may be possible to include the full quality documentation as part of the quality manual; if not, at least references to the procedures should be made. In larger organisations, these documents are typically compiled into one or more procedure manuals (there is no formal obligation to do so).

It may be useful for education and training bodies to explain in the manual how the 'production terminology' has been interpreted in the education/training/ learning environment, and to justify on what grounds certain clauses (or part of them) were not considered applicable, or interpreted in a particular way.

4.2.2 Quality systems procedures

A 'procedure' is a formalized and often standardized description of a process. A process consists of a series of tasks undertaken by two or more persons. Most of these tasks are sequential, but some of these could occur in parallel and/or be repeated within the same process. The description of one or more sequential task undertaken by a single person is often called a 'work instruction'; a procedure may contain or refer to one or more work instructions. In practice, there is no strict boundary line between both concepts. There are also no strict guidelines on the complexity and length of procedures; for maintenance and document control reasons, it is often appropriate to keep procedures relatively short.

It should be noted that the Standard uses the term 'procedure' fairly often. In fact, the term in ISO 9000 corresponds with both the concept of a procedure as defined above, and with a work instruction. In this report, the term 'procedure' is only used for the description of a process.

Sub-clause 4.2.2 of the Standard requires that 'procedures' exist for describing the processes within the quality system. These procedures should be consistent with the Quality Policy. Some of the procedures required concern the core processes which are common to any type of education or training organisation, e.g. delivery of courses, recruitment of staff, registration of students/trainees, etc. Other procedures will be specific to certain organisations, e.g. assessment of students or development of learning material. A third group of procedures is related to the specific requirements of the Standard, e.g. internal auditing, document control.



The number, length, and level of detail of procedures should reflect the complexity and variety of the processes within the organisation. Another important factor is the competence level of staff. With highly qualified staff (which is often the case in education and training institutes) detailed procedures and work instructions are not mandatory. In such circumstances, the organisation should be able to provide evidence (e.g. through quality records showing few non-conformities) that more details are not necessary. For lower qualified administrative and support staff more detailed procedures and work instructions may be necessary. Thus, documented procedures should include sufficient detail to enable qualified staff to understand what has to be done and by whom. But procedures must be written in such a way that the quality system can be audited effectively and efficiently. This excludes procedures written at a high level of abstraction.

Writing procedures, work instructions and other associated documents (e.g. forms) can be a tedious task. Education and training bodies should, whenever possible, develop these on the basis of existing documents, but ensure that these are suitably and consistently identified and referenced. Most of the documents which publicly funded education institutions are legally required to have can often be 'recycled' or simply be used as such. While it is still common practice to document the quality system on paper, the procedures, work instructions, forms or other related documents might also be available (possibly exclusively) electronically – or even in any other accessible form, e.g. on audio or video cassettes.

4.2.3 Quality planning

The term 'quality planning' may be misleading, and should not be confused with 'quality plans'. 'Planning' refers to a method, a 'plan' to the result. This clause requires that quality planning is carried out to ensure that quality policy and objectives (Clause 4.1.1) are met, and to define methods of implementation for meeting the customer needs. Basically, an organisation should undertake such planning in a systematic way. Since most agreements with customers include the execution of standard sequences of tasks, such tasks will already be documented in procedures. For many education institutions with a regular provision of the same types of programmes, all aspects of quality planning may be covered by standard procedures included in the documented quality system itself. This would exclude, however, that a non-standard activity (e.g. a new, customized course) to be part of the quality system covered by the certificate.

For most other education and training organisations, the Standard provides in this clause a list of activities which should be considered for quality planning, and which may be integrated in the quality planning for 'non-regular' activities. This process may be limited to creating a sheet of paper showing the specific requirements of the customer, the appropriate (parts) of procedures which are to be applied (by whom and when), possible exceptions, and any other extra information needed to ensure that the needs of the customer are met. There should be evidence that quality planning occurs and that it is effective.

4.3 Contract review

4.3.1 General

The word 'contract' is used within ISO 9000 in a general way, meaning any type of agreement between the education or training provider and its customers (students, trainees, employers, etc.: anyone who receives services from the organisation). Thus, a 'contract' refers to an agreement to deliver education, training or other services; it may be



actual or implied, written or verbal. In principle, the 'contract' may cover all activities at the customer/supplier interface. In education and training, these contacts are likely to be frequent and interwoven in the other processes (design, delivery, evaluation,...). Since such processes are well covered by the requirements of other clauses of the Standard, one can limit the scope of this Clause to the initial 'agreements' made with the customers.

This clause does not prescribe what an (initial) agreement should contain. Rather, it states that every agreement with a customer should be 'reviewed' – which means that an assessment is made on whether one can deliver what is promised. The bottom line is that a provider of education and training services should have verified that the organisation is capable of fulfilling the 'contractual' requirements.

Many education and training providers offer pre-designed courses or programmes, to which students or trainees can enrol. The registration establishes a 'contract' between the provider and customer; often, many features are implicit. This 'contract' is usually not negotiable: the student, trainee or employer has only the choice to accept or reject the offer made. In all such cases, 'contract review' is not really applicable to the individual registration of students/trainees, but applies to the general conditions and promises of the provider. 'Contract review' comes down to a regular review by the provider of:

- the suitability of standard conditions offered to students, employers,...;
- the accuracy of the promotional information (e.g. a course prospectus);
- the up-to-dateness and appropriateness of the information provided in advance of a course (including pre-enrolment guidance);
- the implicit expectations generated (e.g. what a parent expects from a school; or the quality of the facilities in an expensive top management course).

Where custom-made training is to be delivered – in particular for continuing education and training – this Clause should be interpreted like in most other business sectors: both customer and provider must know what they agree, and there must be clear evidence of a review of terms and conditions. There will be need for evidence that every custom-made agreement has been reviewed.

Contract review also applies to the conditions imposed by grant-awarding bodies: no grant for an education or training project should be accepted if the provider is incapable of meeting those requirements. If legal constraints apply to certain types of education or training provision, customers should be informed of these.

Please note that some experts also include the agreements with sub-contractors, particularly external trainers, under this clause. This line of thought is not followed here, since this aspect is sufficiently covered by Clause 4.6 and the control mechanisms included elsewhere in the Standard.

4.3.2 Review

Before a 'contract' is concluded, both customer and supplier need to understand what education and training services are being offered. They should be deliverable by the education or training organisation, and evidence should be available that the resource implications have been taken into account. This clause requires that such aspects have been formally considered before a final contract is agreed (this is what is meant by 'review'). Examples of elements which may need to be reviewed are:

- the suitability of existing training courses for re-use or adaptation;
- the availability of know-how (content related, didactic, methodological,...);



- the availability of qualified staff (at the required moment);
- the feasibility of the proposed planning and the time schedule;
- the technical feasibility and the availability of resources and equipment.

Particular attention should be paid to those situations where the final agreement is different to what had been offered or required initially.

An almost inevitable implication of this Clause is that verbally made agreements imply some positive confirmation in a written or electronic form. This may be achieved using fax confirmation messages, confirmation by letter, or logging verbal orders and confirming them back to customers over the phone.

For many education and training organisations – particularly those using standard conditions for their courses – the procedure or work instructions to be established to meet the requirements of this clause will be fairly simple. In more complex situations, there might be the need to have separate procedures for:

- preparing a quotation or a draft contract;
- reviewing and approving a quotation and/or a contract.

4.3.3 Amendments to contracts

Between the start and the end of the contractual period, the agreements made initially may have to be amended. This applies in particular to long education and training programmes, or to the development of comprehensive education or training materials, where changing circumstances may require modification of the programme. Such changes would normally be covered by other clauses of the Standard, particularly Design (4.4), Process Control (4.9), and Control of non-conforming product (4.13). Thus, it is possible to limit the scope of this clause to modifications which occur just after the agreement has been concluded, and before any design or development work has begun. A typical example is a sudden programme modification because of unavailability of a trainer. Another example is the rescheduling or cancellation of course. Such events could be documented in a separate procedure or work instruction (if they occur regularly), as an explicit part of other procedures (e.g. on the conduct of courses), or through the generic arrangements for dealing with non-conformities (see Clause 4.13).

What matters is that both the provider and all customers concerned are aware of the changes in the agreements and their implications. Thus, the provider should have good communication channels with all customers affected, as well as a mechanism for internal communication of all contract amendments and their implications. This is particularly important for larger education and training organisations, where several departments may need to be informed and to act accordingly.

4.3.4 Records

A written contract is to be treated as a quality record in itself (see Clause 4.16), but there should also be evidence of the contract review process itself (e.g. minutes of meetings, list of changes made). Such evidence should also be kept and maintained as quality records.



4.4 Design control

4.4.1 General

The correct understanding and application of this clause on 'design control' is crucial for the application of the Standard to education and training programmes. The main purpose of the design parts of the quality system is to develop education and training services which start from identified needs and objectives. The first paragraph of this Clause is an introductory one, highlighting the need to have appropriate procedures for the design and the verification of the design, and to ensure that all design requirements are satisfied.

Clause 4.4 is only appropriate if the programmes or courses developed are to some extent unique or customised, in e.g. subject matter, delivery method or assessment method. When there is, within the scope of the quality system, only marginally original design activity, then this Clause does not apply and ISO 9002 is applicable instead of ISO 9001. This applies for instance when

- the provision of education or training is based upon standard syllabi
- education or training is predefined within explicit requirements (either requested by customers or prescribed by the state, like in schools)
- when the programme is compiled by assembling modules of standard programmes or courses.

Any marginal design activities left could then be covered under Process Control (Clause 4.9). Practice to date indeed confirms that schools and highly regulated adult education bodies, subject to external prescription of educational content, are indeed certified on the basis of ISO 9002, while continuing education and training providers and higher education institutions aim for ISO 9001, since their capacity to design their programmes themselves is a critical quality characteristic.

Anything which requires an original design is covered by this clause. For education and training providers this may include:

- curriculum design and development;
- course design and development;
- design of course syllabi;
- design of learning materials (including software based materials);
- design of assessment materials and methods;
- design of assessment/examination questions;
- definition of work placements;
- preparation of guidance and tutoring of students;

and possibly:

- preparation of course visits;
- planning of teaching/training schedules;
- design of extra-curricular activities;
- design of other services delivered and developed by the provider (e.g. consultancy, research, training needs analysis,...).

Whether activities such as preparation of course visits fall under this clause or rather Process Control (Clause 4.9) depends on the 'design originality'.



For education and training activities, the interpretation of the different paragraphs of this clause is often not straightforward, for several reasons:

- There may be several, but related design processes: programme design, course design, development of syllabi, guidance, etc. For each of them there should be documented design methodologies which comply with the different requirements of the clause which may be difficult or even impossible.
- The design process in education is often based on subjective grounds and/or its quality may be difficult to assess.
- For education and training courses, particularly the relatively short ones, it is often impossible to distinguish between elements of design, development and planning. For longer programmes, (re-)design might even take place during the period of delivery.
- The appropriateness of the design can often only be assessed after the education or training has been delivered; for a course which is only run once, it implies that certain verification and review steps required by the Standard are either impossible to implement or do not give full guarantee of quality of delivery (this characteristic also applies to many other services).

It should also be noted that 'design' of education may have two components:

- The first is the definition of education and training goals and objectives. When these are predefined by the state or the customer, this aspect does not need to be considered. However, when the provider has a certain degree of freedom in setting the goals and objectives, the processes of identifying these learning objectives could be so important (e.g. on the basis of training needs analysis, market research, comparisons with other programmes) that they merit a procedure in its own right.
- The second, more 'traditional' type of design in education and training is when identified learning goals serve as the basis for the development of programmes, courses and learning materials. The focus of the discussion below will be on such forms of design.

The different sub-clauses of Clause 4.4 define different stages of the design process. A common requirement is the clear definition of responsibility for each aspect of the design process. This should be documented in procedure(s).

It is recalled that some education and training bodies also provide 'non-educational' services and products (e.g. research). Such design activities are not considered here.

4.4.2 Design and development planning

This second paragraph of the clause basically states that all design and related development activities (for curricula, courses, training materials,...) should be carried out in a planned way. One needs to specify:

- all the critical tasks which need to be done;
- the output of each task;
- the staff who need to carry out the activities.

Staff assigned to design should be qualified to do so; adequate resources need to be available. It goes without saying that standard procedures should be available to avoid that a new plan has to be established for every new design activity.



4.4.3 Organisation and technical interfaces

This sub-clause applies in particular to larger education and training providers where different people and/or different departments jointly contribute to the design (of a curriculum, a course syllabus,...). The responsibility and authority of all the staff involved in the design process (including lecturers, course directors, programme managers, tutors, external trainers), and their interfaces, should be clearly defined and documented. Appropriate communication channels and coordination mechanisms should exist to provide all relevant parties with correct information at the right time.

4.4.4 Design input

'Design input' refers to all the information elements which are necessary (and may reasonably be expected to be obtainable) for the design of a particular education/training output. The final goal of this first step is to arrive at a sufficiently detailed specification which will guide the design process itself.

Design input elements may vary significantly from one type of design to another; they may include:

customer and course related elements

- the overall learning goals, objectives and outcomes to be achieved;
- the requirements for assessment of learning (e.g. the type of examinations);
- how and to what extent the quality of the output should be evaluated;
- the desirable period and the number of hours available for learning;
- the financial possibilities of the customer.

learner related factors

- educational profile, existing qualifications, entry-level of learners;
- other characteristics of learners (age, sex, working environment,...);
- familiarity of learners with particular types of learning (face-to-face, distance learning, etc...);
- types of support available to learners (e.g. tutor, line manager, additional learning support materials, self-assessment methods, library, equipment,...);
- restrictions on place and time for the learning.

internal factors

- internal rules resulting from the quality policy and the quality system;
- commitments made in programme prospectuses and course announcements;
- rights and promises given to students/trainees;
- availability of resources (staff, place, equipment,...) at the required moment;
- competence and qualifications of available trainers/teachers;
- the design or content of similar, existing outputs;
- outcomes of previous design and programme reviews;
- financial or resource constraints of the provider.

external factors

- legal requirements and constraints which are applicable;
- external quality standards or criteria which are adhered to by the provider;
- requirements of external examining or certification bodies;
- requirements needed for a particular certificate or external award;
- outcomes of appropriate market research;
- opinions of particular bodies representing the customers (e.g. professional associations suggesting requirements for certain graduates).



Furthermore, the design input information should include the results of any contract review activities which are not included in the above list. The input requirements considered have to be documented, since they will be used for review and comparison with the design output (see below). The design input elements have to be reviewed; incomplete, ambiguous or incompatible requirements must be resolved with those who have issued them (this may even require an amendment of the contract).

In order to obtain some of this input information, it may be necessary to carry out market research, analyse appropriate literature, discuss practical issues with customers, or conduct a training needs analysis. An efficient way to deal with all these requirements is to work on the basis of one or more checklists where all the possible input requirements for a particular design process are listed.

4.4.5 Design output

The 'design output' information contains the final outcome of the design process. For many education and training development processes it is difficult to define where 'design', 'development' and 'delivery' stops. If any 'development' is excluded from 'design', then the 'design output' data might not differ much from the 'design input' data, the main difference being that some choices have been made between different options available (e.g. choice of a particular training method, an assessment methodology, a particular trainer). When, on the contrary, 'design' includes most of the development process, then the design output might almost resemble the final education product (finalised courses, syllabi, programme schedule, etc.).

When putting the 'design output' stage in the early phase of the development process, one may identify the following types of design output elements:

- detailed objectives of every module;
- a definition of the knowledge, skills and attitudes the learners should acquire:
- the didactic approach to be adopted;
- resources and staff to be used for development and delivery:
- scheduling of modules or lessons:
- teacher and trainer manuals (or outlines of these):
- overview of the format and content of the learning support materials:
- external or internal validation criteria of the courses or materials;
- the evaluation and assessment methods and tools to be used:
- planning and guidance for the development, highlighting critical aspects.

The distinction between design 'input' and 'output' is not always clear-cut, particularly for relatively modest designs following a standardized approach. In education and training there is also a tendency to specify design input requirements in output terms. Once the type of design output has been defined for a particular design process, one should check that the requirements this sub-Clause states are met. The design output should:

- be documented properly;
- conform to all appropriate regulatory requirements;
- be formulated in such a way that a comparison with design input is possible;
- be verified and validated on the basis of the design input requirements (see 4.4.7 and 4.4.8 below);
- contain or reference criteria for the acceptance of the final outcome of the service (development or delivery);
- highlight critical factors for development and delivery of outputs.



Although, in the end, most of these rather technical requirements boil down to common sense, this step is a formal requirement and should take place in every design process (could be at the verification stage).

4.4.6 Design review

The sub-clauses 4.4.6 (design review), 4.4.7 (design verification) and 4.4.8 (design validation) are closely interrelated. For short and simple design processes in education and training, the requirements will almost entirely overlap. For more significant designs the stages will differ, but remain closely interrelated.

The main purpose of the 'design review' is to the evaluate suitability of the design process itself. This requires an education or training institute to:

- check whether the design input elements are still relevant;
- review the capability of the design process to meet those requirements.

Representatives from all units of departments involved in the design process (or the particular design stage under scrutiny) should be involved in this review. Other competent staff should join where this is felt appropriate (e.g. an expert might be asked to give his independent opinion).

A design review can, in principle, be conducted at any stage of the design process, preferably towards the end of every major stage. It is up to the education and training organisation to determine the frequency, timing, format and thoroughness of the design review. This will mainly depend on the complexity of the design, the number of people involved and the uniqueness of the design methodology under review. For instance, if a continuing education institute always uses the same design methodology for short courses, then the design review can be 'light' and often been subsumed in the verification stage (see 4.4.7). On the other hand, for major design processes such as the creation of a complete curriculum or the production of comprehensive learning materials, several thorough reviews might be necessary. Such reviews are likely to take place as part of progress meetings attended by all parties concerned.

Whatever approach is taken, the design review activities should be planned in advance. Problems identified through the review activities should be given appropriate solutions. Records of the design review process (e.g. minutes of meetings) should be kept.

4.4.7 Design verification

'Design verification' is a necessary step to ensure that an acceptable design is obtained. In essence, verification activities should compare the output of the design process with the design input requirements. Given the intangible nature of education and training, defining proper and affordable verification mechanisms may prove to be difficult. Possibilities include:

- pilot courses/programmes (trial runs), if a course is run more than once;
- comparing the design with similar designs;
- external comments from examination bodies, experts,...;
- feedback and/or agreement from customers;
- self-assessment on the basis of an appropriate checklist.

Records of the verifications should be kept.



In principle, the verification of a course design is not the same as the verification of the course itself. In reality, it may only be possible to verify fully the appropriateness of the design after the course has taken place (covered by the requirements of Clause 4/10, Inspection and Testing). On the other hand, verification of learning materials before finalisation and delivery, is often feasible — and indeed desirable.

When a design is complex and consists of different phases (e.g. the adaptation of a new curriculum), the verification should be conducted at each stage. On the other hand, design verification for a typical short custom training course may require no more than asking the customer for confirmation that the design corresponds with what he needs.

4.4.8 Design validation

'Design validation' is the formal approval that the design process has been 'completed and that development and/or delivery can go ahead. In principle, validation should follow successful design verification. When designing education and training programmes, it may not be possible to undertake validation at all, or only at earlier stages in the design process. This applies in particular when full verification is only possible after delivery of the service. In such cases, 'validating a course' course implies then that at least the 'controllable' aspects of the course design have been reviewed and verified; validation then coincides with the verification.

This clause is only fully applicable to the design of tangible items (learning materials, assessment tools, assessment questions,... – where the validation requirement can easily be implemented) and to the development of course programmes or other educational services which will be undertaken several times. In such cases, it would appear most appropriate to validate the course after the first trial run: only then one is in a position to check whether all user requirements have been adequately addressed. Of course, all precautions have to be taken to ensure that such a trial run also meets the requirements of the customer.

4.4.9 Design changes

There should be documented mechanisms for dealing with changes which inevitably arise during the design processes, particularly longer ones, and for programmes which are run more than once. One should have a mechanism for identifying when changes are needed, for implementing and recording them. These changes could affect the overall objectives, and be either customer driven or resulting from internal evaluations or voluntary improvements.

For courses which are only run once, for occasional handouts, this clause does not really apply, except for longer programmes where a mid-term review could indicate the need for changes.

4.5 Document and data control

4.5.1 General

In education and training organisations, many documents which are needed for a quality system exist already. Often lacking are, however, standardisation of documents, descriptions of some processes, and a system to 'control' such documents. The purpose of this clause on 'Document and data control' is to make sure that everyone in the organisation



has – or has access to – up-to-date copies or electronic versions of the documents and data they need. Clause 4.5.1 states that there should be documented procedures for document and data control. There are two main processes to be controlled: approval and issue of documents (Clause 4.5.2) and modifications and changes of documents (Clause 4.5.3).

The requirements of this clause apply, in principle, only to the documents and data which are used within the quality system itself; for education and training organisations this means that relatively few documents are excluded. Documents to be considered include

- the quality manual;
- procedures and procedural manuals;
- work instructions, forms, lists whether these are printed, in electronic form, handwritten, etc.;
- training materials, course syllabi (in particular when they are used more than once).

It could be argued that agreements, monitoring and evaluation results, reports, project and course files, minutes of meetings, filled-in forms, letters, etc. also fall under this definition. However, such documents or data are either specifically covered by other Clauses of the norm, or by the general requirements of the 'Quality records' clause (4.16). For this reason, we will exclude these from the discussion and focus on documents and data which serve as an input to, or description of, the quality system. To illustrate the difference: the design and adaptation of a form falls under this clause, while a filled-in form falls under Clause 4.16, 'Quality records'.

Document and data control also concerns externally generated documents which have an impact on the quality of education and training, such as:

- all external documents referred to in the quality and procedural manuals;
- information and prescriptions from customers which are necessary for producing the required education and training;
- legislation, government circulars affecting the delivery of education and training;
- requirements of external certification, accreditation or other relevant bodies;
- external course syllabi or training materials used as part of the provision (however, this requirement is also covered by other clauses in the Standard);
- relevant handbooks and user manuals (e.g. on software).

Of course, the organisation cannot itself modify such external documents (as required in Clause 4.5.3), but they still need to be appropriately codified, issued and distributed to those who need them. If external documents are not automatically forwarded to the organisation (such as government circulars), arrangements should be in place to ensure that the latest edition of such documents are immediately acquired (e.g. through a subscription to an appropriate publication service). If an external document is modified, obsolete copies have to be removed in a similar way as internally controlled documents (Clause 4.5.2).

If the document and data control system is partially or totally electronic, then:

- sufficient hardware should be available to facilitate access
- security measures should be in place to avoid unauthorized modification
- backup measures should be appropriate to avoid unintended loss
- printing should be restricted to avoid spreading of obsolete documents.



The essence of Clause 4.5 is that anyone who issues a 'controlled document':

- is authorized to do so;
- has approved the latest version;
- knows who has access to the document:
- is confident that obsolete copies have been removed;

and that any user of a document

- knows which version he or she is using;
- is confident that it is the latest, authorised version.

4.5.2 Document and data approval and issue

This sub-clause requires the following to be part of the procedure(s) for document control:

- identification of all documents which are to be 'controlled documents' (this includes course material, when it is likely to be re-used);
- instructions on how and by whom controlled documents may be prepared;
- instructions on how controlled documents are to be uniquely referenced;
- mechanisms to ensure that all controlled documents have been checked by approved personnel before they are issued and released;
- distributions lists of the persons who need to receive controlled documents;
- mechanisms for making controlled documents available to the people on the distribution lists;
- a master list or equivalent set of documents identifying the current revision status of all controlled documents.

Moreover, there should be an effective mechanism to ensure that obsolete documents:

- are either immediately deleted
- or, when archiving certain obsolete documents is needed for legal or other justified reasons, that they are removed and protected from unintended use.

Please note that 'access' or 'availability' does not require that everyone has a personal copy; rather that people know the controlled documents which may affect them and can consult these at an appropriate location. For instance, education and training organisations may make available only one set of documents for a whole department, a unit or for (groups of) teachers and trainers. Although such arrangements facilitate document management, they also require extra attention for communicating to all staff all changes which have occurred in the documents that affect them (Clause 4.5.3).

4.5.3 Document and data changes

The following elements need to be in place:

- all persons who are authorized to make changes to controlled documents should be identified;
- if these persons are not the same as the original issuers, then they should have adequate access to all relevant background information;
- the nature of change should be indicated (on the document itself, on a separate list, in the communication attached to the revision,...);
- there should be a system for recording the changes which have been made to documents.



Please note that these principles apply, not only to procedures and work instructions, but to all controlled documents, except for external documents, where only the nature of the changes can be recorded.

It is common practice to indicate a small revision by a version number increase like 1.1, 1.2, 1.3 etc., while major revisions are indicated by starting at the next integer number, e.g. 2.0, 3.0, etc.

4.6 Purchasing

4.6.1 General

This clause covers the 'purchase' of all goods and services which are critical to the quality of education and training services offered. Purchasing includes:

- acquisition of goods;
- leasing or renting equipment, services;
- sub-contracting of services;
- use of external trainers or teachers (see below);
- recruitment of interim or temporary (short-term) staff;
- external professional assistance (consultancy, translation,...).

All such products and, particularly, services are to be 'purchased' to a defined specification and from reviewed source.

This clause deserves particular attention for all education and training organisations which use external teachers and trainers, since the institute-trainer interface is critical to the delivery of quality education and training. Examples of such 'externals' are: 'free-lace trainers', 'independent trainers', 'interim' or temporary teachers, invited academic or professional experts, etc. When such 'external' trainers deliver substantial amounts of education or training for an organisation on a regular basis, it seems more suitable to consider them – for the application of the Standard – on the same basis as regular staff, whatever their employment status might be.

Other sub-contractors for which the requirements of this Clause may apply are:

- external experts who advise on education and training issues (e.g. on course content, multimedia development, quality assurance,...);
- external tutors, guidance providers, assessors or examiners;
- other education or training organisations to which whole or part of the design, development or delivery of programmes and courses is sub-contracted;
- persons and organisations who develop tools, forms or other documents used (e.g. for training needs analysis, evaluation,...);
- temporary staff for administrative support;
- desktop publishing and translation services;
- interpretation services;
- transport services;
- mailing and courier services;
- providers of accommodation (particularly classrooms) or catering;
- providers of relevant equipment, software and/or audio-visual material;
- publishers or editors of books and learning materials; bookshops;
- printing or copy services of promotional and course material;



- companies and organisations providing work placements, equipment or facilities which are an integral part of a course programme;
- certification, awarding or accreditation body services.

This list includes possible sub-contractors for education and training related services. When education and training organisations develop other services (e.g. research, supervision of pupils, sport and cultural activities,...), other types of sub-contractors might also need to be considered.

When the number and type of such sub-contractors is limited, then it is often possible to cover all the requirements of Clause 4.6 in one single procedure (with different annexes per type of sub-contractor) or include parts or all of the requirements within other procedures (for instance, provision of training rooms or catering could be embedded in a procedure on the delivery of a training course). For use of external trainers, a separate procedure is recommended.

To summarize, the purchasing process can vary between being fairly important — especially for continuing education providers who use external trainers, and for higher education institutions offering many services — to rather trivial — e.g. with schools offering regular education programmes in their own premises, exclusively with own staff.

4.6.2 Evaluation of subcontractors

For each pertinent type of sub-contracting, the principles of quality assurance should be applied. The first step in this is to assess potential sub-contractors on their capability to deliver the quality level expected.

For all external trainers and teachers (and by extension, for organisations who provide such persons or provide training services on a sub-contract basis), the following needs to be present:

- a clear definition of competence requirements, related to the nature and type of teaching and training to be delivered;
- evidence that the persons selected meet those requirements (based on track records, academic qualifications, professional experience, previous performance, trainer certificates and awards, evaluation by trainees, recommendations by peers or clients, etc.);
- evidence that the persons selected understand the operation and conditions of the part of the quality system which affects them;
- a controlled list of teachers and trainers who have proved to be suitable for certain types of education and training.

It is strongly recommended that a special 'recruitment' procedure is established which deals with such 'purchasing of teachers/trainers' and which includes the above elements. Even education and training organisations with a long history of collaboration with external trainers, are likely to require a higher degree of formalisation in the selection of trainers and monitoring their performance. These people should be monitored (on the basis of classroom observation, feedback from trainees, regular discussion,...) to establish whether they should be re-used in the future, and for what purposes.

For the other types of pertinent sub-contracting the following needs to be in place:

identification of the relevant quality standards, specifications or other requirements; the level and degree of detail will depend on the nature of the sub-contract and the extent of impact on the education/training service;



- sub-contractors should only be selected if they can be shown to be able to meet those requirements (for existing sub-contractors this might be on the basis of documenting previous experience; for new ones, on the basis of a documented formal assessment or audit on the basis of the identified requirements);
- sub-contractors need to understand the requirements of the quality system, be able to demonstrate adherence with requirements and keep whatever records needed to prove conformity with the quality system;
- the performance of selected sub-contractors should be regularly reviewed (possibly with audits);
- quality records (Clause 4.16) should be kept regarding the performance and monitoring of sub-contractors;
- a list of approved sub-contractors should be maintained, and available where such purchasing may be initiated;
- only approved subcontractors should be used, except in emergency situations (where there should be strict authorisation).

Through such mechanisms, it is ensured that all sub-contractors have the capability of supplying materials or services of the required quality.

Monitoring of subcontractors may involve some of the following activities:

- recording of any problems, defects, incorrect delivery, poor service, errors, etc.; such records are to be reviewed regularly;
- reviewing of performance on a regular basis;
- carrying out audits.

The procedure(s) should identify how action will be taken if a subcontractor does not meet requirements. This may eventually include removal from the approved subcontractors list.

4.6.3 Purchasing data

Essentially, this sub-Clause requires that all 'purchasing documents' (order information, agreements, letters and related documents) are fully explicit as to what the organisation wants to purchase and to what standard. Such documents should be reviewed and approved by the organisation before they are released.

Agreements with external trainers should include the specific contributions required for a particular course, but also (possibly under the form of a reference to other documents) information about the general requirements of the quality system for trainers.

For the other sub-contractors, this sub-Clause requires making sure that the order information is unambiguous and sufficiently detailed. Orders could in principle be verbal, but then at least a documented record should be created.

4.6.4 Verification of purchased products

This sub-Clause may apply to situations where the education or training organisation and/or its customer wish to verify the suitability of e.g. external training facilities, residential accommodation, work placements, etc., after the sub-contractor has been selected. A customer may also want to meet or observe a particular trainer before accepting him or her. Such aspects should be suitably arranged with the sub-contractor, and do not modify the organisation's overall responsibility for ensuring the quality of service.



4.7 Control of customer supplied product

In manufacturing the external customer may sometimes provide materials for incorporation in products they ultimately buy back. In services, particularly education and training, this is rarely the case. Examples include:

- the provision of accommodation by the customer;
- the provision of equipment and materials (manuals, software, video tapes,...) to be used for learning purposes;
- background information, reports, specific know-how;
- case studies, examples to be integrated in a course.

In schools, particular attention should be given to tangible items pupils bring with them: these should be fit for purpose, safeguarded against loss (if feasible) and protected from damage (in as far as the school sees this as its responsibility). Something particular which may be considered under this clause are educational credits and exemptions from courses which students carry with them.

Essentially, this Clause states that where such situations occur and may influence the quality of the service, the organisation should:

- check the quality and the suitability of what has been supplied;
- store and maintain it carefully (record it in the case of credits);
- report to the customer any loss or damage.

Where such processes apply, they should be documented. This could be done as part of other procedures.

4.8 Product identification and traceability

There are three more or less tangible 'products' of education and training providers to which identification and traceability applies:

- learning materials, in particular books and syllabi;
- course programme content and structure;
- awards given to the learner.

This clause requires the appropriate *identification* of such 'products'. This boils down to keeping records of courses which are being or have been taught (course name, by who, for who, when, where, how,...) and linking these with the written specifications and agreements. It requires every course and associated elements (syllabi,...) to have a unique name and/or coding. When courses, syllabi or awards vary over time, then identification needs to extend to include the version, the year or anything else which makes it possible to identify each different course offering. Of course, these codification and naming conventions should be used consistently within the organisation.

Traceability refers to the need to be able to tie each learner to the particular education and training programme followed. Traceability is not mandatory, only needed when agreed at contract stage. However, it is something which most education and training organisations can already provide based on existing learner records.

Note: interpretation of this clause is different if the product is taken to be 'the learning' or 'the student'. Such an interpretation would put more emphasis on student/ trainee records, in particular their development of attainment. However, even with the adopted interpretation (the product is the provision of education and training), there is need to con-



trol student/trainee records, at least because of the requirements of Process Control (Clause 4.9). In either interpretation, maintaining staff records is necessary, but does not need to be covered under this Clause, since it results from the combined requirements of other clauses (including 4.1, 4.6, 4.9 and 4.18).

4.9 Process control

This Clause is the heart of ISO 9000. It encompasses all education and training related processes which are not covered explicitly somewhere else in the Standard. Depending on the type of organisation and the education and training services offered, this may include one or more processes from the following categories:

development and preparation

- conduct of training needs analysis of trainees (could also be covered by Clause 4.4);
- development of a particular course module (may fall partially under 'design');
- development and finalisation of a learning material;
- educational software development;
- translation, adaptation and editing of learning materials;
- desktop publishing and finalisation of course materials;
- printing/copying of education and training materials;
- planning and scheduling of the course provision [see also the clauses on 'design' (4.4) and 'quality planning' (4.2.3)];
- cancelling and rescheduling of courses; replacements of teachers.

administration and support

- individualised and/or mass mailing of information, course prospectus, etc.;
- development and maintenance of the mailing list and address data files;
- management of the computer services (could be several procedures);
- distribution of appropriate information and materials to teachers;
- provision of practical information to students/trainees/employers on course schedules, classrooms, requirements, etc.;
- distribution of course materials to students/trainees/pupils;
- registration of students/trainees.

learning related activities

- teaching and delivery of classroom training;
- provider-led conduct of practical exercises, software simulations, laboratory work, workshops;
- tutoring, monitoring and providing feedback to students (for courses, project work, exercises, theses,...);
- provision of educational guidance to students (at several stages throughout the programme);
- supervision of on-the-job training and work experience placements;
- provision and use of library resources (may include several procedures);
- adjusting courses and programmes during delivery;
- educational-oriented visits and tours.

organisation of education and training sessions

- facility management during a course;
- provision of appropriate equipment and didactic material;
- catering and accommodation;
- recording attendance of students/trainees;
- supervision of pupils (in schools);
- reward or discipline measures (in schools).



assessment and evaluation of learners

- verification of student/trainee ability and entry qualifications;
- intake discussions of students or trainees;
- development of test and examination questions;
- assessment of students (tests, examinations);
- awarding of degrees and certificates.

Note: most education and training organisations (particularly schools and universities) provide other services to students (accommodation, social services, transport, administrative services...) or to customers (e.g. research, consultancy,...). Although these are not discussed in this report, such activities are in general to be documented according to the requirements of this Clause. This also holds for 'generic' services such as health care availability, career guidance counselling, safety and security arrangements, transport, etc. - whenever these can be considered as part of the (sometimes implicit) agreements with the customers.

In principle, all such processes need documented procedures. This may be a daunting task, but the reality is less frightening. First of all, in most education and training organisations, only a subset of these processes exist and/or are critical to the quality of the service (for instance, most schools do not develop their own materials, and most private training providers do not formally assess trainees). Thus, part of the list can immediately be abandoned. Secondly, some processes may be so simple and straightforward that it makes sense to group several of them within a single procedure - or subsume the activities in other procedures. Finally, some of the processes listed may be sub-contracted. Thus, the number of procedures to be documented in order to comply with the requirements of this clause, will in most cases be far shorter than the long list above may suggest.

In principle, Clause 4.9 need not apply to those education and training related activities covered elsewhere in the Standard, such as:

- contract negotiation (4.3);
- design and development of courses and learning materials (4.4);
- purchasing of goods/services and use of external staff (4.6);
- monitoring and evaluation of the education and training programme (4.10-4.13);
- training of staff (4.18);
- servicing after the course (4.19).

However, in an education or training context the requirements of these clauses often reflect the same principles of quality assurance as laid down in Clause 4.9. This implies that procedures, work instructions, document control, quality records, etc. can be approached from the same perspective. One can, therefore, give a consistent look to all procedures and related documents which are part of the quality system. There is no need to give the procedures which have 'their own Clause' a 'special place' in the quality and procedure manuals. Rather, the structure of the various processes documented in the manuals should reflect the internal logic of the organisation. The only processes which may merit a special place - since they affect virtually everyone and are specific to ISO 9000 - are the 'generic' quality assurance procedures, i.e. those concerning the diffusion of the quality policy and management review (4.1), document and data control (4.5), corrective and preventive action (4.14), control of quality records (4.16), and internal quality audits (4.17).

Clause 4.9 lists several requirements which need to fulfilled, some of which translate badly to education and training processes (and to service processes in general). However, clause 4.9 states in its penultimate paragraph "where the results of processes can-



not be fully verified by subsequent inspection and testing (...) the processes shall be carried out by qualified operators and/or shall require continuous monitoring and control of process parameters to ensure that the specified requirements are met. This applies to most of the processes to be considered. Thus, evidence should be available of the incorporation – where feasible – of the principles of 'controlled processes' to any process falling under this Clause:

- documented procedures and/or work instructions, including sufficiently detailed criteria or guidelines for the operation of the process;
- suitable, approved and well-maintained equipment and an appropriate working environment;
- availability of qualified staff, with specifications of the required competences (in particular for critical activities);
- compliance with any applicable norm, standard, quality plan, or other procedures;
- monitoring of progress (if possible through measurable indicators); records of these monitoring activities should be kept for a reasonable period.

This is really common sense for anyone concerned with quality in education and training. The challenge is, however, to manage all of this in a systematic way for all pertinent processes.

Since these processes are to be controlled, the organisation should be able to provide evidence that the procedures have been followed. A system of records (many of which often exist already) and regular checks (through internal audits, lack of complaints about non-delivery or defects) may suffice for most of the processes. Test and examination results may provide further evidence. What matters is that an organisation can demonstrate that adequate standards and procedures have been defined, and that these are in general followed (mechanisms for identifying and correcting problems are discussed later).

In most of the processes listed, the quality of competence of staff is critical. This reinforces the importance of setting standards for staff and making sure to recruit qualified staff and external teachers/trainers who meet such standards. That is a far more productive approach then defining a process control procedure with endless checks and routines. In as far as staff recruitment has not already been covered by complying with the requirements under one of the other Clauses (e.g. 4.1, 4.2, 4.6 or 4.18) it makes sense to have a procedure for staff appointment which respects the requirements of Process control, i.e. includes:

- standards for different types of staff (qualifications, experience, skills, attitudes,...);
- how potential new staff members are assessed against those standards before recruitment;
- how the continuing relevance of staff competence is monitored (see also 4.18).

Process control is also the area where most of the work instructions are to be written down, since this is the core operation of education and training organisation. Caution is necessary, since this is not common practice in many education and training organisations. Once appointed, most education institutions leave teaching staff a high degree of freedom to get on with the job. For ISO 9000 certification this is clearly insufficient, but one should equally avoid the temptation to overprescribe how teaching and other educational activities are to be carried out. One approach to this difficult choice is to consider what type of promises are given to customers, or what expectations are raised (or might be expected) with students/trainees. Procedures and work instructions should contain enough information, guidelines and monitoring mechanisms to make sure that these promises and expectations can be fulfilled.



Whether a particular work instruction is needed or not, also depends to a large extent on the qualification of the staff member. The higher qualified, the less need there is for detailed work instructions. However, through the system of internal audits (Clause 4.17) it should become quickly obvious in which areas work instructions are lacking, and in which ones they are superfluous.

To summarize, process steps and minimal standards need to be identified and documented for all critical processes. The process must be carried out in a controlled way, so that plans, procedures and instructions are available, staff and equipment used are adequate, and that the general conditions are appropriate to achieve the plans.

4.10 Inspection and testing

4.10.1 General

There are a number of interpretation and applicability issues concerning this Clause. When course programmes and other learning opportunities offered to students and trainees are considered as the 'product' in the context of the Standard, 'inspection and testing' refers to the monitoring and evaluation of the development and delivery of those services. Logically, under this interpretation, assessment of learners (tests, examinations,...) is not covered by this Clause, but by the requirements of Process Control (Clause 4.9.).

This view is not shared by all experts involved in the certification of education and training establishments. In practice, however, whether assessment of learners is related to Clause 4.9 or 4.10 does not make much difference for the quality manual, procedures and quality records.

A second problem is that the different stages at which, according to the Standard, inspection and testing are supposed to take place (receiving, in-process, final stages) may either coincide or simply not exist. Moreover, it is often not possible to use 100% reliable inspection and test methods, given the intangible nature of services. In such situations it will be important for the organisation to show that the spirit and the purpose of this Clause are adhered to as much as possible. The problem of applicability of the Standard also holds for Clause 4.12 and even more for Clause 4.11.

Overall, the Standard requires under this Clause to inspect all incoming 'products' (if these are critical for the end-quality), to check training and education products and services while they are under development, and to check the output for conformance to specifications (if possible) before they are released for use. Other clauses (notably 4.13) also require to set aside non-conforming 'products' so as to prevent their accidental use.

Education and training services consist of tangible items such as learning materials, and the non-tangible teaching and assessment processes and related activities. Some experts have argued that teaching and assessment is already fully covered under process control, so that Clause 4.10 need only be applied to the tangible items which are used or produced for the education or training service. In such a narrow interpretation, the requirements of this Clause could apply to the following types of products:

- purchased learning materials (textbooks, software, equipment, audio/visual items,...),
- assessment and other tools,
- supplies, specific material used in courses,

always, of course, in as far as these are critical for the quality of the output.



In a broader interpretation the Clause "inspection and test" also covers teaching and related activities, to the extent that the specific requirements are applicable (e.g. 'incoming product'-inspection is in general impossible) and/or are not yet covered by other clauses. This broader interpretation is adhered to in the sequel. Simplified, the spirit of the set of requirements laid down in this Clause is that an education or training provider can demonstrate that the development and delivery of education and training services is checked, monitored and reviewed at appropriate intervals and with suitable methods – all in order to ensure that what has been required and promised will be delivered.

A range of different methods and approaches can be used in order to comply with these 'test and inspection' requirements, such as:

- course and programme evaluations by learners during, at the end and after the education and training delivery;
- self-assessment by teachers and trainers;
- monitoring of effectiveness of the education and training by programme coordinators, customers, independent assessors;
- statistical analyses of student performance, including trend-analyses;
- indirect measures of quality (e.g. motivation of learners, complaints,...);
- opinions expressed by customers and people who have completed a course;
- comparisons with related programmes, benchmarking;
- comparison with existing standards.

What type of evaluation is most appropriate, which method is to be preferred, and at what intervals evaluations need to be conducted, will depend on the circumstances and the particular type of provision: the organisation must be able to show that the evaluation mechanisms are appropriate to ensure the continuous quality of the education and training services provided.

The first paragraph of this Clause 4.11 explicitly lists the requirement to have documented procedures for test and inspection in order to ensure that specifications are met. These inspection and test activities must be part of either standard procedures or specific quality plans: the results should be considered as quality records (Clause 4.16).

From a practical point of view, the assessment, evaluation or review of individual items of a programme (e.g. a particular course, a syllabus, a practical training period, project work,...) may be documented as part of the general procedure(s) which cover their development and delivery, i.e. those procedures that need to exist in order to comply with the requirements of Process Control (Clause 4.9). On the other hand, for comprehensive education and training services, for the overall didactic and pedagogical approach, and/or for a course programme as a whole, it may be more appropriate to document the evaluation mechanisms in a separate procedure.

Where education and training organisations deliver other types of services (e.g. guidance, research, supervision, etc.), it will also be necessary to comply with the requirements of this Clause. Although such processes are not discussed here, the interpretation to be given to the inspection requirements might often be similar, mutatis mutandis, to what applies to education and training. It is indeed a common characteristic of services that inspection can nottake place before delivery, but only during the service or after it.

4.10.2 Receiving inspection and testing

This sub-Clause requires to inspect or test all incoming goods and services before they are put to use (4.10.2.1), by taking into account the quality control mechanisms already



used by the supplier to avoid unnecessary double checking (4.10.2.2). In urgent cases, when no inspection has been possible, the products concerned need to be clearly marked and identified (4.10.2.3). In education and training, three cases can be considered:

- tangible products which are critical for the education or training provision;
- education and training services provided by external trainers or organisations;
- other services supplied by sub-contractors.

By their very nature, services (including education and training) cannot be 'inspected upon receipt'. One possible exception is when trial runs of courses have taken place: such trials could be used for 'receiving inspection' – but it seems more logical to consider this as part of the design requirements (Clause 4). Since formal assessment of sub-contractors (including external trainers) is already covered under Clause 4.6.2, the requirement of this sub-Clause 4.10.2 can be safely ignored for education and training services.

Please note that, when 'learning' or 'student' is considered as the 'product', there would be a compulsory requirement to test their entry level and qualifications — unless, of course, the customer does not want this to happen. In reality, whenever such action is to be taken, it will need to be documented; in the interpretation followed in this report, this would be covered by process control (4.9).

When critical, tangible products are acquired, they need to be inspected prior to use. Special care should always be taken when purchasing complete learning packages, courses and textbooks from external sources (this concerns both the publishers and the authors). Obviously, the inspection can be reduced if effective preventive techniques are employed, such as purchasing items from approved subcontractors where high confidence levels have been established. In such cases, all that may be required is that the teacher or trainer checks at the beginning of a course that all items used are OK (for long programmes, this may have to be repeated several times during the overall period of activity).

In any case, some kind of record has to be made of the inspection which has been carried out, demonstrating that the goods were judged to be suitable for their intended use.

4.10.3 In-process inspection

Purchased goods (e.g. a course handbook, an assessment tool) are most often used unaltered, so there is no need for any type of process-inspection. If they are modified, inspection of the adaptation may occasionally be necessary, to avoid problems jeopardizing the timely and effective development of the training or education service for which the product is being modified.

For education and training 'products' which are developed by providers themselves (e.g. a course syllabus), in-process inspection becomes relevant as soon the development time (the period between the initial design and the final delivery) is substantial. This should, in principle, already have been documented following the requirements of Design (Clause 4.4) and Process Control (Clause 4.9).

The same observations apply to the development of the services (educational and others). The typical example is the development of a new curriculum. But in addition, "inprocess" inspection also applies to the delivery itself, since that is part of the service process. For courses which last more than one day, it is often desirable and appropriate to have one or more interim evaluation. These should comprise at least the checks carried out to ensure compliance with the planned programme (e.g. on the basis of student



and staff records, student assessment, course evaluation,...). Whether other services delivered merit such type of in-process inspection, will have to be considered in the light of their contribution to the quality of the programme. Again, one would expect that appropriate "in-process" checking mechanisms have been built into the procedures established following the requirements of Process Control (4.9).

4.10.4 Final inspection and testing

Final inspection and testing is always applicable:

- for goods it concerns a final check before release;
- for services, particularly education and training itself, it may concern the verification of a test course or prototype (if applicable), but above all it applies to the evaluation of the services after they have been delivered.

The evaluation of education and training activities is not only important from the perspective of this Clause. It also serves as an input or verification for the requirements of many other Clauses of the Standard, including:

- Contract Review (4.3 Were specifications appropriate? Were they met?);
- Design (4.4 Was it appropriate?);
- Process Control (4.9 Was delivery as specified?);
- Corrective Measures (4.14 What should be changed?);
- Internal Audits (4.17 Where should we improve?).

The requirements of this crucial sub-clause refer, for education and training, to all types of assessment, monitoring, test or evaluation activities (including evaluations some time after the course) which are used to analyse the performance of the programme. All these together should provide an overall picture which makes it plausible that the quality requirements have been met.

A minimum requirement in higher education and continuing education and training is a quality assessment by students/trainees, preferably on standardized evaluation forms. Where relevant, this should be complemented by evaluations by teachers themselves, by programme coordinators or employers. In schools, assessment by pupils may not be possible or relevant, but it could be replaced by teacher self-assessment, opinions of other teachers (in particular those who 'receive' the pupils at a later stage), a review of the results of examinations and assessment of pupils, opinions of education institutions or employers who take on the pupils; etc.

In cases where evaluations can be conducted entirely before the final release of a course or material (e.g. trial runs of courses, review of the quality of course syllabi by independent experts), the evaluation results must be acceptable before the delivery can take place.

4.10.5 Inspection and test records

The procedures which cover all the checks, assessments, reviews and evaluations mentioned, should include:

- the requirement to maintain quality records of such evaluations;
- the records should show whether activities have been satisfactorily conducted;
- if records show problems or non-conformities, procedures for dealing with 'non-conforming products' (Clause 4.13) should be applied.



4.11 Control of inspection, measuring and test equipment

4.11.1 General

Although this is one of the longest clauses of the Standard, the applicability to education and training is fairly limited. This holds both for the services delivered, and for any education and training products supplied. It has already been highlighted that provision of education and training cannot easily be measured, and that the instruments to do so are not 100% reliable.

This clause states that documented procedures must exist which ensure that the evaluation, test, assessment and checking methods, software, equipment and tools are sufficiently reliable to ensure that the education and training provision conforms to specifications.

This Clause may therefore apply:

- in the first place, to any instrument, software, tool or method used to ensure that learning is taking place in the way which was planned and agreed (such approaches being required by Clause 4.10);
- but also when applicable to any inspection, measuring and test equipment used by learners as part of a learning process where such measuring is critical to the learning process (this is not a very frequent situation, mainly referring to learning situations where the use of calibrated equipment is important).

When an education or training organisation also undertakes other activities for which it also seeks certification (e.g. research, carrying out surveys, publishing,...), the full requirements of this Clause may apply.

The thrust of this Clause concerns predominantly the evaluation tools used during the development and delivery of education and training. These may include:

- forms for recording assessment information,
- reference standards and norms.
- checklists.
- questionnaires,
- specific software,
- audit methods (internal and external),
- benchmarking procedures,
- assessment of performance methods by externals,
- external awarding systems (for labels, certificates...).

Following the requirements of this Clause, an education or training provider must be able to show the reliability of the test methods and tools used:

- the outcomes are valid;
- evaluation results can be safely compared from one course (or other product) to another.
- the same method is applied consistently throughout the whole organisation.

Of course, this reliability has to be proven within reasonable limits (any judgement in education and training is likely to have some subjective component) and in as far as the application of the methods is under the full control of the provider (for instance, abuse of questionnaires by students can never be completely eliminated). The organisation has to demonstrate that reliability and effectiveness of evaluations are taken seriously and that all reasonable efforts have been made to obtain outcomes which can be trusted. No or-



ganisation is expected to take steps beyond those that are customarily accepted by the education and training sector as relevant. Often, the acquisition or modification of existing, commonly accepted external evaluation methods may suffice.

Interpretation of control requirements for tangible control equipment used within a learning context is briefly discussed under the next paragraphs.

If agreed with the customer, the evaluation, test or assessment results must be delivered to the customer.

The methods for assessment of students are not included under this Clause since these do not measure the 'product'; they fall however under Process Control. A well established examination system where the quality assurance principles of Process Control apply will include mechanisms to ensure that assessment is reliable and consistent to a reasonable degree. Therefore, the spirit of Clause 4.11 will be present.

4.11.2 Control procedures

As regards the evaluation methods of education and training activities, this sub-Clause states that these should be well justified, documented, and the outputs should be recorded. This does not add any new requirements to those already discussed above. Let it just be recalled that:

- the extent of evaluation should be determined in advance
- appropriate evaluation tools should be identified and used in a consistent manner
- evaluation methods and frequency should be documented and implemented
- evaluation should take place under suitable circumstances
- results of evaluations should be reliable
- quality records are to be kept of all evaluations.

There are cases – particularly in courses with a considerable technology content or those concerning metrology as a subject – when tangible measurement and test equipment needs to be used and whereby its accuracy impacts on the quality of the course (e.g. when low accuracy may provide students with wrong information). This equipment may concern, for instance:

- weighing scales,
- electrical and electronic meters, oscilloscopes.
- chemical measurement instruments,
- ECG equipment.
- any equipment used during assessment, where the calibration can affect the learner's score.

In such cases, the Standard requires the organisation to:

- identify all equipment where calibration is necessary;
- define the accuracy needed for such equipment;
- select appropriate equipment;
- check and calibrate the equipment at regular intervals and under suitable conditions;
- let qualified staff undertake the calibration;
- calibration itself must be based on predefined methods and on appropriate reference standards:
- keep quality records of these calibrations;



- recalibrate and re-assess non-calibrated results when appropriate;
- avoid deterioration of control and measurement equipment.

In addition, for equipment used for other services (e.g. commissioned research, surveys, consultancy,...) the case for calibration must be carefully thought through; procedures should be documented accordingly.

4.12 Inspection and test status

This clause expects an organisation to know what tests and evaluations of the education and training products and services (at development and delivery stage) have been applied and what the result was. The information available should indicate whether and to what extent the test results were satisfactory; if they were not, appropriate action has to be taken (see Clause 4.13.2). The clause applies to all services and goods which undergo one or more test or evaluation process.

For most education and training organisations, this requirement will be relatively easy to fulfill, provided all evaluations can easily be linked to the relevant programme activities – and vice-versa. Adequate documentation and filing of programme records will normally suffice to meet these requirements. A useful tool is a standard checklist which is filled in for every regular type of activity, where all evaluations scheduled can be ticked off and commented on as needed.

This clause is of particular relevance for:

- long programmes which consist of many modules and learning materials (in order to know for each component to what extent it has been evaluated);
- programmes with trial runs or which are run for the first time;
- provision of education and training by using learning materials which are not fully finalised (e.g. software in 'beta-version').

4.13 Control of nonconforming product

4.13.1 General

'Nonconformance' refers to any feature of the quality system where performance or outcomes are not in line with what had been planned and agreed. This clause requires an organisation to have procedures for undertaking immediate action to prevent such nonconformities adversely affecting the quality of the service.

More specifically, this sub-clause states that the following general requirements are to be met:

- (timely) identification and recording of non-conformities;
- assessment of the impact of the non-conformities on quality;
- isolation of non-conforming items to avoid inappropriate use (when feasible);
- notification to the appropriate people;
- immediate remedial action when needed.

Some of these requirements are further developed in the next sub-clause.

For education and training organisations, non-conformities may thus appear in such diverse processes as the development and delivery of the programme, the practical ar



rangements for students and trainees, the sub-contracted services, students/trainees performance (when due to the provider), learning materials, etc. Thus, this clause applies not only to faulty items and defects in tangible products used in teaching, such as:

- defective audio-visual equipment,
- damaged books and other learning materials,
- teaching materials (e.g. slides, OHPs,...) which are not consistent with the course syllabus,
- ill-functioning demonstration or simulation material

but also to problems, errors, omissions, etc. relating to the services provided, e.g.:

- late or non-arrival of a teacher/trainer,
- poor coverage of planned subject matter,
- improper didactic materials,
- scheduling errors,
- inappropriate accommodation,
- low validity of examination results,
- poor performance of teachers,
- irrelevant programme content,
- loss of documents.

Not all non-conformities have the same importance. For services, three typical cases can be considered:

- the non-conformity is of minor importance (e.g. will not affect quality and/ or is unlikely to be repeated): in such cases no further action is required;
- the non-conformity indeed affects quality, but cannot be remedied anymore (e.g. behaviour of teachers, incomplete coverage of subject matter): in such cases customers should be informed, and the information should be fed into the procedure for Corrective Action (Clause 4.14) to prevent it from re-occurring;
- the non-conformity indeed affects quality, but immediate remedial action is possible (e.g. reschedule a module; update the learning materials): this is covered by the next sub-clause.

4.13.2 Review and disposition of nonconforming product

This sub-Clause concerns the actions to be taken as a result of any type of non-conformance (where such action makes sense – cf above). Requirements for the documented procedures include:

- definition of responsibility and authority for dealing with nonconformities;
- assessment of the non-conformity and decision of action to be taken (rework, accepted as such, used in another context, completely rejected,...);
- seek agreements with customer for continued use of any nonconforming product;
- re-assess conformity of any reworked product;
- keep quality records of these reviews.

Such requirements are fairly obvious for tangible products, e.g.

- labelling them 'out of date', '1996 only', etc.;
- storing them in a separate location, marked e.g. as 'non-current learning materials'.

However, the requirements need suitable interpretation for the intangible education, training and related services. What matters is that an organisation always takes timely



and appropriate action whenever a problem occurs. Such action should, moreover, be recorded in one way or another to allow for continuous improvement [following from the requirements of Corrective and Preventive Action (Clause 4.14) and Internal Audits (4.17)].

One may consider having a separate procedure for 'Control of nonconforming product', but one could also include the mechanisms in place as part of the specific procedures.

4.14 Corrective and preventive action

4.14.1 General

This clause is, combined with Internal Audits (4.16) and Management Review (4.1.3), crucial for ensuring the continuous improvement of the quality system and the quality of the services delivered. While the 'Control of non-conforming product' (Clause 4.13) concerned the immediate steps to be taken when problems occur, this clause on 'corrective and preventive action' requires documented procedures for adapting the quality system in order to systematically prevent or reduce the non-conformities. Such measures will often lead to an amendment of procedures or related documents.

The difference between 'corrective' and 'preventive' action may not always clear-cut, in particular when the provision is subject to considerable change: a corrective measure to one course is at the same time a preventive measure for another. In as far as it is necessary to make the distinction, corrective measures are always a response to a problem or a non-conformity, while preventive actions are to be taken to avoid problems occurring, before any problem has arisen.

This Clause does not stand on its own, but is related to, and overlaps with, many other requirements of the Standard:

- corrective and preventive measures could be both an input to, and an output of the Internal Audit (4.17) process;
- modifications in procedures resulting from these measures have to be controlled (4.5);
- the data used should have been recorded as quality records (4.16) following the Control of nonconforming product (4.13);
- corrective measures may result from action undertaken as part of the design verification (4.4) and inspection and testing (4.10) activities.

It should be noted that this first sub-Clause states that the nature and extent of corrective or preventive action should be related to the importance of the problem and the risks involved – a common sense requirement.

4.14.2 Corrective action

This sub-Clause exclusively concerns corrective measures and requires that documented procedures cover:

- a system for handling complaints;
- the effective and timely dealing with reports and records concerning nonconformities;
- the analysis of the causes of nonconformities;
- recording the results of such analyses:
- decisions regarding the appropriate corrective action to eliminate the causes or reduce their impact (whenever feasible);
- monitoring of the implementation of corrective actions.



These requirements can be met by education and training organisations in different ways. As a minimum there should be a procedure for effectively dealing with all kinds of complaints from customers, students or trainees. It is often useful to extend such a procedure to include complaints from staff, as well as to broaden it to a 'suggestion scheme' for quality improvements.

One may, in addition, create either a 'generic' procedure for corrective action, or instead incorporate the requirements of this sub-clause in other procedures. If the quality system is modest in scope and the control of processes is well documented, some or all of this may be covered under the Internal Audits procedure (4.17), making it unnecessary to have a separate 'Corrective Action' procedure, except for the complaints handling. However, the quality manual should at least refer to the corrective action arrangements which are in place and where and how these are documented. What matters is that the organisation uses a systematic method of:

- investigating nonconformities or noncompliances and their causes;
- scanning the available data and records in such a way that worrying patterns or trends may be detected;
- defining methods for analysis (e.g. statistical techniques, quality improvement projects....);
- implementing and monitoring all corrective measures.

It is recalled that records should be kept of problems, defects, the investigation of their causes and any follow-up corrective actions implemented; these should be used as part of the internal audit process (4.17).

Education institutions which provide certificates and degrees when pupils/students have successfully passed examinations, will also need to foresee a specific appeal and complaint procedure concerning the examination process. In general, such a requirement is often already documented under the form of a 'student charter' or a document listing the 'examination rules and principles'.

4.14.3 Preventive action

The principles are similar to those of the previous sub-Clause. The Standard requires an organisation to use all relevant information sources in order to prevent problems and non-conformities occurring. Possible sources (in addition to those used for corrective action) are:

- market research and training analysis reports;
- progress reports;
- management reports;
- surveys, evaluations;
- internal and external audits and assessments;
- statistical analyses of student/pupil/trainee performance;
- trend analyses of data (e.g. on student performance, trainee complaints,...);
- feedback from former students.

Based on the timely analysis of such information, preventive actions should be implemented in similar ways as the corrective actions (see above). Preventive actions are closely linked to the outcomes of internal audits (4.17); the outcomes should also be discussed as part of the Management review (4.1.3). Of course, the effectiveness of the preventive measures should be monitored.



The organisation must be able to show that reasonable efforts are undertaken to prevent nonconformities occurring, and that these efforts are based on appropriate information and analysis, and the use of suitable methods. Again, it is important to be able to show that the analysis undertaken has led to suitable preventive measures. If the quality system is fully operational (including corrective measures, internal audits and regular management reviews) there may be no need to have a separate procedure for preventive measures – but the organisation will still have to be able to give evidence that preventive measures are systematically applied. Again, the quality manual should at least refer to the arrangements which are in place and to how and where they are documented.

4.15 Handling, storage, packaging, preservation & delivery

Since it is difficult to cross-refer the individual headings of this Clause to education and training activities, they are discussed together.

- 4.15.1 General
- 4.15.2 Handling
- 4.15.3 Storage
- 4.15.4 Packaging
- 4.15.5 Preservation
- 4.15.6 Delivery

This is a clause where the term 'product' has to be interpreted literally as a tangible item, excluding services. Most of what is stated in the clause about products is often not critical for education and training, and does, therefore, not justify elaborate procedures. Products where parts of this clause may apply to might be:

- learning materials, in particular open learning materials;
- books, journals, other documents and audiovisual materials in libraries;
- materials on which learners work (e.g. wood);
- computing equipment and software;
- language lab equipment;
- other tools and equipment used in the learning process;
- assessment and examination papers;
- informative documents, guidelines.

Overall, the requirements are straightforward:

- all critical aspects of handling, storage, packaging, preservation and delivery of products should be documented as part of procedures;
- items should be properly marked so that they can be identified if needed;
- the handling, location and storage method of critical products should be such as to prevent damage or deterioration even after final inspection (this may relate to cleanness, humidity, temperature, light, etc. of rooms);
- storage areas should be appropriately secure (this applies e.g. to exam papers, library documents);
- there should be appropriate methods for logging such materials in and out the storage areas;
- the quality and condition of the products in stock should be assessed at appropriate intervals;
- packaging should be implemented according to the agreed specifications;



78

- items waiting to be checked before use should be separated from items ready for use;
- when agreed with the customer, the protection of critical items should also include the delivery to destination (e.g. for open learning).

Often, for the education and training services provided, all that needs to be done is to include one or more paragraphs or work instructions within the procedures for purchasing, product development or delivery – and only for the products and stages where this is critical. Larger institutions, for instance with extensive library facilities, or those disseminating learning materials, will need to ensure that the requirements of this clause – particularly 4.15.1 (general), 4.15.3 (storage) and 4.15.6 (delivery) – are properly documented in separate procedures or integrated in the procedures required for Design (4.4.), Purchasing (4.6) and Process control (4.9).

If students are considered as the 'product', then the coverage of the Clause has to be extended. Under this interpretation, services like health care availability, career guidance counselling, safety and security arrangements, transport, also need to be covered. In the interpretation followed in this report, such activities are — if considered critical for the quality of the service — to be documented under Process control (4.9).

4.16 Control of quality records

The obligation to keep 'quality records' is stated at numerous places in the Standard. Minimum requirements listed explicitly in a particular clause of the Standard are quality records for:

- management review (4.1.3);
- contract review (4.3.4);
- design review (4.4.6);
- design verification (4.4.7);
- evaluation of sub-contractors (4.6.2);
- verification of purchaser supplied product (4.7);
- identification and traceability (4.8);
- qualified processes, equipment and staff (4.9);
- release of incoming products before verification (4.10.2.3);
- inspection and test records (4.10.5);
- checking of test hard- and software (4.11.1);
- calibration of inspection, measuring and test equipment (4.11.2e);
- accepted non-conformities and repairs (4.13.2);
- investigations of the cause of nonconformance (4.14.2);
- results of internal quality audits (4.17);
- follow-up activities of internal audits (4.17);
- records of training of staff (4.18).

The requirement to keep a record of particular information is also an explicit or implicit requirement of several other Clauses (e.g. for document control (4.4), corrective and preventive action (4.14), etc.), although the term 'quality record' is not used.

Clause 4.16 extends the definition of quality records to cover the full set of records which can "demonstrate conformance to specified requirements and the effective operation of the quality system". Thus, the minimum requirement is the establishment and maintenance of a set of records which together provide the documented and objective evidence of the smooth operation of the quality system at all stages, and of the conformance to the procedures and quality plans.



A number of general remarks apply:

- such 'quality records' may exist on paper, in electronic form or even otherwise (e.g. indications on a planning board or a wall);
- the output of a process itself can be a quality record: filled-in forms and checklists, letters, planning schedules, minutes of meetings, reports, internal memos, etc.;
- it follows from the above that in a well organised education or training organisation, many of the necessary quality records will already exist (although they may not be standardized and not maintained properly)

Thus, an education or training institution will need to identify which existing records could serve as quality records, and which additional ones need to be created in order to comply with the requirements of the standard. Some creativity is needed in order to minimize the number of records necessary, e.g. by trying to:

- draw together several types of quality records on one paper (e.g. by using checklists);
- modify existing outputs so that they may also serve as quality records (e.g. slightly adapt a form used for accountancy purposes).

Again, the prime purpose must be to be able to provide evidence of the proper functioning of the quality system – nothing more.

For quality records identified by the organisation, this clause requires to:

- identify and codify them appropriately;
- gather and collect them;
- have appropriate filing and storage arrangements;
- define access to them:
- maintain them during a predefined interval;
- dispose of them after the retention period.

None of these requirements is difficult to implement (although retention periods are often ill-defined); however the challenge is to apply the rules for all quality records consistently. Simplification can be obtained by treating several quality records together: for instance, all quality records concerning a particular student, or a particular training course, are kept in one file together: this will simplify storage, access, maintenance and disposal.

Retention periods could vary and should be defined in the light of the requirements of the law and the quality system. Some examples:

- records of non-conformities could be kept for one year or until the next internal audits;
- data on courses, students and staff could be kept for about 5 years;
- information on degrees awarded should be kept for much longer.

During the retention period, records should be protected from theft, damage or deterioration (including data corruption of electronic records).

The term 'quality record' may lead to the misconception that new, special records have to be created for the quality system. In reality many existing day-to-day records are suitable as quality records. Whether or not certain information is appropriate for that purpose depends on its critical contribution to the evidence that procedures are followed and the quality system operates well. The following is a checklist which may help in establishing a list of suitable records:

■ all types of documents (forms, reports, lists, minutes,...) used for evaluations, audits, tests, assessments, reviews,...;



- external or internal reports of reviews, tests, screenings, audits, benchmarks;
- letters, notes, memos;
- minutes of meetings;
- quotations, contracts, agreements;
- invoices (if sufficiently detailed);
- registers;
- complaint forms;
- a log-book of courses and events;
- student/trainee records (courses followed, results obtained,...);
- staff records (particularly qualifications and performance including for guest lecturers and external trainers);
- lists of (possible) suppliers or sub-contractors;
- internal or external reports, studies, analyses, surveys, statistics,...;
- filled-in forms and checklists;
- computerised files (appropriately protected);
- exam questions and papers;
- video recordings of training;
- computer logs;
- documents produced by quality improvement projects;
- documents produced in relation to staff training;
- lists of certificates awarded.

This list might provide some inspiration on how existing records could be used as 'quality records' – as such or after minimal adaptation (e.g. by making sure that minutes of meetings always contain certain standard information). Rarely, all these documents will be needed to provide evidence of the quality assurance system in operation. On the other hand, not all records are to be considered as quality records either: they are only quality records when:

- the requirement to keep the records is specified in a procedure; some of these records are a compulsory requirement of the Standard, and
- the records meet the aforementioned requirements for labelling, retrievability and retention.

It may be necessary to control records maintained by a subcontractor, if these are needed to demonstrate conformance to specified requirements or the effective operation of the quality system.

When keeping records on persons, or other sensitive data, legal requirements regarding the storage of data and privacy have to be respected.

4.17 Internal quality audits

This clause on internal quality audits does not need special interpretation for education and training organisations. Nevertheless, the requirements are fairly new for most of them. Even in well organised, prestigious and recognized education or training institutes, internal audits were rather uncommon before the arrival of ISO 9000. Fulfilling the requirements of this Clause, together with those on document control (4.5) and quality records (4.16), make the real difference between a good functioning organisation, and one with a quality system based on the Standard.



The requirements of this Clause are easy to understand. There need to be documented procedures which cover the following aspects:

- a schedule should exist which ensures that all activities which are part of the quality system are audited at appropriate, regular intervals;
- the audits should be carried out to schedule;
- internal auditors should be appointed;
- these auditors should have the skills necessary to undertake internal audits; appropriate training should be arranged;
- the audits should be carried out by persons who are independent from the activities being audited;
- the audits should verify the compliance with procedures and quality plans;
- the audits should also analyse and determine the effectiveness of the quality system;
- audit reports should state the deficiencies and problems observed and recommend improvements where necessary;
- audit results should (at least) be communicated to those who are responsible for the area audited;
- follow-up corrective actions should be carried out promptly;
- the implementation of corrective actions should be verified and recorded;
- audit records are to be treated as quality records;
- results of audits should serve as an input to the management reviews (4.1.3).

It is obvous that these requirements impact on the whole organisation.

In general, it suffices to write one procedure which specifies how the organisation is going to run its internal audits. Some requirements (training, quality records, corrective action, management review) could be incorporated into other procedures. Some general remarks and observations on the audit process:

- the frequency of auditing should be determined in the light of the importance of the processes; once or twice a year is often a good starting basis, to be adapted in the light of experience;
- the requirement that auditors are independent of the area being audited, may be difficult to comply with in a small training company; in such cases the auditor should pay extra attention to objective analysis;
- auditing of quality management systems is a skill: there should be appropriate training (1-2 days + some trial runs) to make sure that auditors are well prepared, undertake their job in the right mind set and are accepted by colleagues;
- auditing should focus on the process and the activities, not on the people; this approach helps to overcome resistance and hiding of problems;
- the auditor should have access to all relevant information (including quality records, corrective action and previous audit reports), and be able to ask any question considered relevant to knowledgeable staff;
- care should be taken to identify both conformity and nonconformity, to ensure auditing is a positive process; the better the quality system, the more internal audits will be welcomed as opportunities for improvement, rather than exam-type judgement processes;
- the need for and the type of corrective action is best agreed between the auditor and the auditee; this will facilitate implementation.

Given their size and limited resources there will be few education or training organisations with full-time internal auditors. It is best to create a small pool of motivated people. This implies that for most of them, auditing will only take a few days per year of their time. Since auditing can be an interesting learning experience, such an arrangement also facilitates the distribution of knowledge and experience across the organisation.



4.18 Training

Various sections of the Standard require that all activities which may affect the quality of the service are to be carried out by competent and qualified staff. This clause expands on this by requiring that training needs of staff – both current and new – should be analysed systematically, so that at all times they have the knowledge and skills they need. In particular, there should be documented procedures regarding the training of staff which include:

- analysing the training needs of staff;
- providing training in order to fill any training gap identified in the training needs analysis:
- ensuring, through appropriate education, training, learning or experience, that all staff members who undertake activities which affect quality, are qualified and competent to do so;
- keeping records on the training of staff and their qualifications.

Please note that these requirements may also apply to regularly used external trainers – in as far as such requirements have not yet been covered by Clause 4.6 (purchasing of training services).

Training needs may be assessed on the basis of regular surveys, as part of the annual staff appraisal, or following analysis of deficiencies in the quality system itself. Special care is needed for the analysis of training needs and the updating and re-training of teachers and trainers. Although the Standard allows an organisation to set its own requirements – corresponding to the quality levels expected by its customers – it is often useful to specify qualification requirements and re-training activities in three areas:

- didactic competences (which should be related to the typical target group of the organisation, e.g. adult learners);
- subject-related competences;
- programme related knowledge and skills (e.g. concerning the specific approach used in a particular course, the quality policy of the provider, etc.).

Evidence of current qualifications of staff could be provided by maintaining up-to-date CV's.

The concept 'training' should be interpreted broadly, including induction of new employees, guidance, and staff development. This affects all staff, including managers, teachers/trainers and support people. In order to develop an active training policy, it may be useful to specify general competence requirements, qualifications, and desirable experience for all functions and tasks within the organisation. A useful document could be a qualification matrix, showing which teacher or trainer is qualified to provide certain subjects or courses. One should be aware, however, that qualification requirements change over time and that such documents must be controlled.

An obvious case where all staff will need some additional training concerns the introduction of the quality system itself and the changes in the operation caused by it. Particular attention may need to be paid to:

- the understanding and use of procedures;
- implementation of corrective measures;
- quality records;
- internal audits.



78

For the more specific functions of the organisation (e.g. purchasing) no extra training may be necessary if the staff concerned have themselves been involved in the establishment and review of procedures. Whether training is sufficient or not, can often only be assessed after the training. If a quality system operates well, and few problems can be attributed to errors from staff, then there is an indirect proof that the staff is qualified and well trained.

In public education institutions, the qualifications required from teachers are often prescribed by law. From a legal point of view, these could be considered as sufficient. In such cases, the institution may not have the legal freedom to impose higher standards.

It is finally recalled that within education and training, teachers and trainers often carry out their tasks with minimal contact with the rest of the organisation. This situation requires well-established communication lines to make all staff aware of the (changing) requirements of the quality system, to detect new training needs in time, and to be confident that all staff will contribute actively to the maintenance of the quality system (e.g. by spotting deficiencies and suggesting improvements).

4.19 Servicing

'Servicing' refers to any activity undertaken after delivery of the product and service. This clause says that, if servicing is specified in the agreement with the customer, that arrangements must be in place to ensure that the organisation can carry out these activities according to requirements. If no servicing is agreed or needed, then an education or training organisation can safely ignore this clause.

Possible types of servicing are:

- support of students and trainees after completion of the programme (e.g. help line);
- mailing of learning materials, proceedings, corrected exercises, assessment results, etc. to students and trainees after a course has been conducted;
- providing a detailed review report of a training programme to a customer;
- conducting an additional assessment or effect measurement some time after a course has been delivered (e.g. to assess relevance for a job);
- providing job-seeking services, special facilities for retraining, social activities, etc. for alumni;
- providing refresher courses.

If such activities are part of the quality system and formally agreed with customers, then the general principles of quality assurance – as found throughout the Standard – have to be adopted.

4.20 Statistical techniques

4.20.1 Identification of need

The first sub-Clause asks an education or training organisation to identify which statistical techniques – if any – are needed for the proper function of the quality system. This applies in particular to the verification, analysis and evaluation activities.



In general, these needs are limited for education and training organisations. They could concern, for instance:

- processing of assessment and evaluation results;
- summarizing and categorizing data on complaints, deficiencies, defects,...;
- construction and analysis of performance indicators such as pass rates, cycle times, achievement records, customer satisfaction scores;
- trend analyses of data.

Such elements could be specified as part of the relevant procedures.

It is mentioned that statistical techniques are important for research and some other activities (e.g. surveys) which education or training organisations might conduct.

4.20.2 Procedures

The ultimate purpose of this clause is to guarantee that statistical information is reliable and comparable over a time period. Therefore, appropriate mechanisms should exist to ensure that the statistical techniques are used in an appropriate way. This implies that it is possible to demonstrate their suitability for the purpose and that qualified staff apply them.



5. Implementation of a quality system based on ISO 9000 standards

5.1 Introduction

5.1.1 Purpose of this chapter

Interpreting and analysing the requirements of ISO 9001 for an education or training organisation may intellectually be the most demanding task. The real challenge and most of the work, however, will be the implementation of the quality system itself. In this chapter we provide a short overview of some of the key issues which are to be considered in this regard. The text is essentially based on implementation stories reported in documented case studies, articles, some surveys, telephone interviews and the author's own experience. It does, therefore not claim to be complete, but rather provide an opportunity for those interested in ISO 9000 to learn from others.

5.1.2 Conditions to be met before launching a certification exercise

Section 3.1 highlighted the possible advantages and drawbacks of developing a quality system according to the ISO 9000 requirements. An organisation should consider such arguments before it decides to implement the Standard or not. This is, however, only part of the story. Some organisations benefit from much better 'starting position' than others, and will have far less difficulty in implementing the requirements. Thus, the relevance of the Standard should not only be considered in terms of advantages or drawbacks of the quality system, but also in the light of the complexity and risks of implementation. After all, establishing a quality system is not merely adding a few bells and whistles to an existing organisation, but is an important 'change process' which will have an impact on the whole organisation. It is well known from management consultancy experience that the implementation of change processes is always difficult and risky, and that resources are often underestimated. This also applies to the whole certification process, from the initial decision on adopting the Standard to the final award of the certificate. Interestingly, unlike many other types of change processes, few ISO 9000 exercises seem to be abandoned once they are initiated. One reason more to ensure that one is well prepared.

Although it is dangerous to generalize about the 'ideal' starting requirements for ISO 9000, we suggest to consider the following ten conditions:

- the organisation is already well organised;
- there is a (sometimes implicit) quality policy, with standards which are taken seriously;
- the organisation has been and is likely to remain fairly stable in terms of activities and personnel (no other important change, expansion or streamlining operations are going on);
- there is a good understanding of all internal processes;
- many standardized documents exist already (even more to be preferred: procedures and a quality manual);
- the organisation is financially sound;
- a qualified, motivated and credible (highly regarded) person is available to coordinate the implementation;
- the senior management believes in the value of certification and is committed to it;
- the number of significantly different types of customers, products and services is limited (note that this has little to do with the customised or individualized nature of the products and services);



the organisation is small with only a few departments and maximum a few dozen staff members.

If most of these conditions are met, the organisation can safely engage in an ISO 9000 exercise. But if none or only a few apply, than it is likely that the journey towards certification will be long and paved with obstacles. A bonus in both cases would be the easy access to professional advice and to the experience of similar organisations who have already implemented the requirements.

5.1.3 What is the minimum quality documentation?

The most visible aspect of the fulfilment of the ISO 9000 criteria, is the existence of the controlled 'quality documents' (quality manual, procedures, work instructions, forms,...). In a well-established quality system, the volume, structure and complexity of this documentation should be a true reflection of the complexity of the organisation. Therefore, no specific guidelines can be given as to the desirable size and volume of the quality documentation. It remains useful, however, to pinpoint those aspects which should always be documented as part of an ISO 9000 compliant quality system. These minimal requirements are:

- A specific quality policy and specific quality objectives, which are consistent with the quality system and with the ISO 9000 requirements.
- Description of responsibility and authority of all staff.
- Procedures (with corresponding work instructions, forms, criteria, lists,...) which cover the quality assurance requirements of the Standard:
 - controlling and maintaining of the quality documentation (including the modification of procedures and related documents);
 - conducting internal audits;
 - dealing with complaints and problems (including self-assessment); corrective measures:
 - training and professional development of staff;
 - holding regular management reviews;
 - controlling quality records (this could also be part of other procedures).
- Procedures (with corresponding work instructions, forms, criteria, lists,...) which cover the processes and activities which are explicitly mentioned in the Standard:
 - systematic review of all agreements with customers (including registration of students/trainees);
 - design and planning of courses, programmes, education and training materials (only for ISO 9001);
 - purchasing of goods and supplies;
 - recruitment of staff; use of external trainers (when applicable);
 - monitoring and evaluation of education and training programmes.
- Procedures (with corresponding work instructions, forms, criteria, lists,...) which cover the other processes not already addressed, such as:
 - development and adaptation of courses, programmes, education and training materials (when applicable);
 - delivery of education and training programmes (enrolment, teaching, practical organisation);
 - assessment and examination of trainees (when applicable).



For small organisations with a regular and standardized provision, this may just suffice; it may in some cases even be possible to group some of these processes into a single procedure, while certain items of this list may not apply. All the other requirements of the Standard (e.g. product identification and traceability) can easily be incorporated as part of the procedures mentioned. Thus, about 15 procedures would appear as the minimum for ISO 9001.

This minimalistic approach is, however, more the exception than the rule. For small education and training providers, which essentially always use a standardized approach, about 20 'procedures' seems a more realistic number — at least if the full scope of the activities is included in the quality system. The larger an organisation becomes, the more the number of procedures is likely to increase, up to 100 or more (for higher education institutions). The rate of growth is not so much dependent on the number of staff or the turn-over, but rather on the number of different processes which take place. A small higher education institution, with its many different activities and services to students, may need more procedures than a large continuing education provider.

It is recalled that providing numbers of procedures is always somewhat problematic. Quite different practices exist for the writing of procedures: in certain organisations they are very general, leaving much of the detail to separate work instructions and other related documents; with others, the procedures are much more detailed, with less reference to annexes. It goes without saying that, the more details a procedure contains, the more there will be a tendency to split it up into smaller procedures, in order to keep the maintenance practical. With such an approach, the number of procedures tends to rise. This should be balanced against the fact that usually fewer annexes are needed in such cases.

Another parameter in the numbers game is the required level of detail of procedures and work instructions, which is linked to the qualification level of the people concerned. The higher skilled and educated staff are, the less necessary it will be to use lengthy and very detailed procedures and instructions. Since most education and training organisations have fairly highly qualified staff, it is often possible to leave out details, since it can be safely assumed (and demonstrated by the running of the quality system) that the omission of such details will not negatively affect quality. The fact that fewer details are needed, often allows some streamlining and merging of procedures as well.

With such considerations, we have entered into the area of individual tailoring of a quality system to the needs of a particular organisation. It is not really possible to make any further generalisations on such matters in a report of this kind. What should be retained is that some creativity is needed for creating a quality system with minimal quality documentation, but which remains fully compliant with the Standard.

5.2 Implementation of the quality system

5.2.1 Generic issues

The implementation of a quality system (compliant with ISO 9001 or 9002) is different for every organisation. Nevertheless, the following stages can often be identified:

- preliminary investigations regarding the relevance and cost-benefit ratio of certification;
- decision by the senior management, possibly including the selection of an appropriate consultant:
- analysis of the existing situation: type of customers, outputs, processes, documents available; choice of the appropriate standard (ISO 9001 or 9002);



- awareness building, training (on the standards, writing of procedures), project planning;
- drafting the components of the quality documentation: procedures, quality manuals, instructions, forms, etc. (this is the most resource intensive phase);
- training on document control and internal audits (may start in the previous phase);
- revision of the quality documentation;
- start of operating the quality system (including document control, internal audits, corrective measures, management reviews, etc.);
- external audit and, if successful, certification;
- maintenance of the quality system with 3-5 external audits over a period of 3 years.

It is recalled that the whole process from decision to certification for a typical education or training organisation is often in the range of 12 to 18 months. The reader is also reminded of the fact that a certificate is only valid for 3 years.

5.2.2 Positive messages from implementation experiences

A rudimentary analysis of the experience of education and training organisations which have already been certified, shows the following positive results:

the ISO 9000 certification process

- assists in a better understanding of the outputs delivered and the internal operation of the organisation;
- contributes strongly to the generalisation of process thinking, and the acceptance that every process in the organisation contributes to the final quality;
- quickly introduces a culture where keeping to the procedures is seen as an asset, and not a bureaucratic prescription (this depends heavily on the common sense used when designing procedures);
- makes people aware that many of the requirements are already fulfilled and that many of the documents needed already existed;
- adds a new dimension and dynamic to the existing quality culture;
- provides a real aid to systematic procedural development;
- is often less difficult to implement than initially anticipated;
- quickly demonstrates its applicability and relevance.

These positive messages result from succesful, pioneering cases. Whether they will remain valid when more organisations become certified remains to be seen.

5.2.3 Negative messages from implementation experiences

Even ISO 9000 enthusiasts in education and training recognise that implementation of a full quality system is not an easy affair. Some negative experiences from those already certified in education are listed below:

- there is a considerable direct cost and also an indirect cost which is sometimes underestimated;
- some managers pay only lip-service (they see only the marketing potential) but do not commit sufficient resources;
- there is lack of official guidance on several interpretation issues, and the extent of acceptable variability (the norm is seen as 'education-unfriendly');
- there are no accepted references for the level of standards on which the quality system has to be built;



- work for certain requirements is underestimated (e.g. design, calibration,...);
- customers often have no explicit ideas of what they need and want;
- the implementation process starts slowly because of insufficient expertise;
- for many people, learning to think in process terms is challenging;
- most people lack the skills to write good procedures for their own activities;
- training and raising awareness of own staff takes more time than expected;
- it takes time before a culture of self-reflection and self-assessment emerges:
- the resistance to changing attitudes and way of thinking of staff is often greater than expected;
- people find it difficult to be both critical (even when things are OK) and constructive when problems arise;
- staff members' time constraints make it difficult to progress rapidly, keep on schedule, and ensure continuity in the implementation;
- differing internal attitudes amongst staff (active and less active people, drivers and lazy people) slow the implementation process down;
- there is an almost irresistible tendency to overload the quality and procedure manuals with details, which requires extensive streamlining at a later stage;
- the finishing touch for getting all processes under control often gets postponed;
- once the quality documentation is completed, getting used to the paperwork takes time and effort;
- initially, the quality system cannot keep up with the pace of change;
- the launching of the document control procedure quickly results in a bureaucratic system;
- there is insufficient consistency with regard to other quality requirements (e.g. from the inspectorate).

In order to put such negative experiences in context, it is recalled that most of these have been reported by certified education and training organisations. These are, despite their critical comments, in general positive or even extremely positive about their experience. When preparing this report, managers of about a dozen certified education or training organisations confirmed to the author that, with hindsight, they would do it again.

5.3 First results of ISO 9000 certification in education and training

5.3.1 A marginal phenomenon

At the time of writing of this report, only a few hundred education and training organisations had obtained an ISO 9000 certificate (of which over half in the UK). Even though this figure is likely to increase significantly over the next years, it remains small, both compared to the number of organisations already certified (over 100 000 now), and the several hundred thousand education and training organisations in Europe. So, by all standards, ISO 9000 certification is still a marginal phenomenon in the education and training world, and will remain so for the next years — with the possible exception of certain segments of the training market in some European countries (United Kingdom, Netherlands).

For that reason, the results of ISO 9000 certification which are presented hereafter, should not simply be generalized. Indeed, those organisations which were the first to obtain certification can hardly be considered to be representative for the whole sector. Overall, such organisations report favourably about their experience, but in the author's impression, many of them already had a well developed quality assurance or TQM cul-



ture before they engaged in certification. Thus, more research over a sufficient time scale would be required in order to examine whether the benefits reported continue to outweigh the drawbacks, and under what assumptions the experience may be transferable to other education and training organisations. More analysis is also needed on:

- the cost-effectiveness of certification, and the period over which the investment should be depreciated;
- methods and tools to keep bureacracy at a minimum;
- the real impact of ISO 9000 standards on the improvement of the content of education and trainig;
- alternative quality assurance models, systems and strategies which might be more appropriate and more cost-effective than ISO 9000.

The education and training sector is not unique in facing such issues. Similar questions also apply to many other areas in which the long term implications of ISO 9000 certification remain unexplored.

Notwithstanding these critical remarks, it is useful to list the main results which have been reported by a whole range of certified education and training organisations across Europe. Below follows an overview of the key messages, resulting from an analayis of the literature available, the author's own experience, and some telephone interviews conducted for this study. It goes without saying that all results do not apply to all organisations.

5.3.2 Benefits of certification for education and training

The main benefits of ISO 9000 based quality systems, as reported by certified education and training providers, can be grouped into four categories: external recognition of quality level, an improved quality culture amongst staff, streamlining of the internal organisation, a dynamic of continuous improvement, and more consistent quality outputs for specific processes.

External recognition of quality level

In the cases where ISO 9000 certification is a mandatorty requirement, its acquisition is of course a major benefit. Apart from this, still a rare requirement for education and training institutes, other reported external benefits include:

- an easy external appreciation of the certificate as a real quality label, which helps in marketing and instilling confidence;
- improved access to large companies (applies particularly to continuing education and training providers);
- recognition of leadership in a particular sector by peers (will apply less in the future);
- enhanced credibility of the quality claims made (in particular because of the independent auditing);
- recognition by competitors in the training world.

An improved quality culture amongst staff

Many of the reported benefits relate to an improved quality awareness and commitment of staff:

- the development and maintenance of the quality system provides a clear focus for internal quality improvement by all staff;
- staff have a better understanding of their duties and responsibilities;



- TQM-concepts like "internal clients" and Aprocess orientation' are more easily understood and applied;
- certification provides greater internal visibility of quality efforts and reassurance for staff;
- quality issues are better understood and more efficiently and effectively discussed by staff;
- there is a higher quality awareness when purchasing goods and services from third parties;
- staff have more confidence in themselves;
- staff become more critical of themselves and more constructive in seeking performance improvements;
- staff is fully prepared and committed to continuous improvement.

An interesting – but not unexpected – observation is that the improved quality culture is found more amongst managers and support staff than with teachers and trainers.

Improvement and streamlining of the internal organisation

Here there are two series of benefits. On the one hand, the new quality assurance mechanisms, like document control, internal audits, corrective measures and management reviews are quickly seen as contributing to more efficiency, effectiveness and quality improvement (even when these new arrangements initially meet some resistance).

On the other hand, the quality system contributes to a tangible organisational change, in particular a more streamlined organisation:

- overall, a more rational and transparant organisation emerges;
- all policies and procedures become more in line with each other;
- responsibilities of staff are more clearly defined, with no grey zones left;
- management practice is improved at all levels (leadership is encouraged);
- most processes are conducted in a standardized manner, resulting in more efficient and effective operations;
- internal communication throughout the organisation is greatly enhanced.

Some cases even mention a reduction of bureacracy – but it is not obvious that such findings could be generalized.

A dynamic of continuous improvement

Certified education and training organisations, in particular those who have been certified for a number of years, report positively about the emergence of a dynamic of continuous improvement throughout the whole organisation, noticeable through:

- regular changes and improvements across the whole quality system;
- a more flexible and dynamic organisation (one called it an 'oiled' system);
- a high value added from the internal audits, assisting in the move towards a learning organisation;
- early and better identification of problems and weakness, followed by a fast and adequate response;
- emergence of various spin-offs, such as generalisation of team work and inter-departmental coordination.



More consistent quality outputs

Most certified organisations report on a more consistent meeting of process standards and strongly reduced variability of outputs, including:

- the design and development of curricula and training specifications;
- the development and finalisation of education and training materials;
- the use of external trainers and teachers;
- the running of the administrative support functions;
- the different interface points with clients (students, trainees, employers,...);
- more focus on the needs of internal clients throughout the different people;
- streamlining (up to re-engineering) of specific processes (e.g. purchasing of goods and services).

Overall, most certified organisations are able to show increased satisfaction among students, trainees or employers.

Comparison with other sectors

In the already mentioned study by the Manchester Business School for SGS (see Bibliography), the UK organisations surveyed listed the following as the top ten benefits of ISO certification (in decreasing order of occurrence):

- better management control.
- improving awareness of procedural problems,
- using the standard as 'a promotional tool',
- improving customer service,
- facilitating elimination of procedural problems,
- improving efficiency,
- keeping existing customers,
- increasing customer satisfaction,
- aiding inducation of new staff,
- improving market share.

All of these benefits were reported by at least one out of two organisations. One can easily verify that these benefits are broadly similar to those listed for education and training organisations. The study reported also that "69% of respondents felt that their expectations had been met or exceeded. (...) The majority were also satisfied with the impact of ISO 9000 on their organisation. (...) Comments made by respondents suggest a clear link between management commitment to improvement and satisfaction with ISO 9000."

5.3.3 Drawbacks of certification

A mixed bag of problems

Even education and training organisations who are overall very positive about ISO 9000, recognize a number of problems and drawbacks in the operation of the system. In addition to numerous implementation hurdles (see Section 5.2.3) the problems most frequently mentioned are:

- the continuous volume of paperwork involved;
- the cost of certification and the ongoing cost of maintenance (this is particularly a concern of publicly financed education institutions);
- the risk of evolving towards a bureaucracy focused on procedures and registrations;
- the difficulty of implementing changes fast.



Other criticisms mentioned include:

- ISO 9000 certification does not yield a quality assessment report (which makes it unsuitable for public accountability purposes);
- at least part of the staff may not be motivated to support the quality system;
- there is a danger of thinking one is perfect once the certificate is awarded;
- some risk of 'forgetting' the requirements of the quality system until the next external audit arrives;
- certification does not yield an increase in market share.

These and other problems should not be generalised, since they are highly dependent on particular circumstances.

Some education and training institutions mentioned some 'problems' of an entirely different nature:

- jealousy and even straight rejection of the value of the certificate by peers;
- lack of interest by the press (which has got so used by 'ISO 9000 News' that they see no news value in reporting about another certificate).

Putting the drawbacks into context

Quoting again from the SGS study, the main issues raised by certified UK organisations across all sectors of the economy as 'being a major problem' were:

- the time required to write the manual and to complete implementation.
- the high volume of paperwork, and
- the cost of implementation and maintaining the standards.

Other complaints concerned the lack of free advice, inconsistency between auditors (raised by larger companies), the time spent on checking paperwork prior to audits, the vagueness of the standard, and the difficulty of interpreting the standard. The study also found that "those least satisfied with the standard had been motivated much more by presssure to keep existing customers and to use ISO 9000 as a promotional tool. (...) Small organisations generally considered drawbacks to be more significant, when compared to benefits, than did large organisations. The same pattern was seen in concerns with ongoing maintenance of the standard'.

5.3.4 10 tips and conditions for success

ISO 9000 is not an appropriate goal for any organisation in any circumstance. Neither can anyone engage in the certification process with a reasonable chance of success. Below are 10 tips for successful implementation of ISO 9000 certification by education and training organisations. Most of these are necessary conditions for successful certification:

An education or training organisation interested in ISO 9000 should:

- know very well why it wants to be certified;
- have fully committed managers who are prepared to implement a quality system;
- be convinced of the relevance of the standard (not only of the certificate);
- take all (interpreted) requirements of ISO 9000 serious (unlike TQM, you cannot pick what you like!);
- have a long term vision on quality issues;
- have already high quality standards for its key services;



- be well organised;
- not be in the middle of another major change project (e.g. restructuring, take-over,...);
- be capable of committing sufficient staff time;
- start from a sound financial situation.

If an organisation fails to meet more than half of these conditions, it is unlikely to benefit quickly from its ISO 9000 journey.



6. Conclusion

The research underlying this report has shown that certification on the basis of ISO 9000 standards can provide a valuable framework for the quality efforts of education and training organisations. Experience so far indicates its capability as a mechanism for improved customer service, high levels of quality assurance and a dynamic of continuous quality improvements – the three cornerstones of TQM.

ISO 9000 is not in contradiction with any sound educational standards or practice, and can easily complement other quality approaches (in particular those focussing on input-or output-factors). Obtaining a certificate enhances the organisation's quality image and underpins its quality claims in an increasingly competitive environment. It may enable an education or training provider to fulfill or exceed externally imposed quality criteria.

These positive outcomes come at a price. First of all, obtaining an ISO 9000 certificate requires considerable commitment and resources, in particular when the starting conditions are not optimal. Only a fraction of education and training providers seem currently in a position to engage in such an effort – if they are convinced of its cost-effectiveness. Secondly, ISO 9000 is in itself not sufficient for achieving quality of the content of education and training programmes – in particular in situations where the customers are not in a position to specify or influence the content (including learning objectives, quality criteria, indicators, and evaluation mechanisms). This situation applies to most regular education provision, and to some areas of continuing education and training. From that perspective, ISO 9000 is merely a tool for assuring quality of education and training provision, while the definition of content will have to be derived elsewhere.

It should be mentioned that many of the arguments which have been put forward against ISO 9000 certification of education and training providers – such as the possibility to set low content quality standards, unnecessary paperwork, and the irrelevance of several of the standards' requirements – have so far not proven to be justified. Indeed, such issues appear to have been anticipated and been addressed creatively. But it must also be recognised that the first certified education and training organisations were not necessarily representative either.

What remains unresolved is the question of cost-effectiveness of the certification process and the maintenance of the quality system. More research is needed on the impact of ISO 9000 certification, its relevance, its cost-effectiveness, and the coherence with traditional quality concepts and mechanisms used in education and training.

An international ISO 9000-'like' standard, which would be specifically developed for education and training organisations, would seem to be an attractive option. No international movements in this direction are taking place. There are, however, some interesting developments going on in other corners of the quality world: an 'education' adaptation of the critera for the European Quality Award, internationalisation of (higher) education assessments, international accreditation schemes (also mainly for higher education). It is unclear whether any such initiatives will gain sufficient momentum. In contrast with ISO 9000, they suffer from a narrow focus on particular types of education and training institutions, and from low visibility externally.

Maybe the education and training world will wait until it sees the fully revised version of ISO 9001, scheduled to be available by 2000. This new release will be structured better, use a more generic terminology and will be based more on TQM concepts. That sounds



91

like a more attractive option for education and training organisations – and in any case one which is likely to meet less resistance.

In conclusion, even the increasing number of ISO 9000 certificates is unlikely to put an end to the vivid debate about quality assurance and quality management in education and training institutions. Eventually, it will be the market which will decide whether the cost of certification is worthwhile, whether its benefits will outweigh the drawbacks, and whether any other national or international quality scheme is more appropriate.



Bibliography

Most of the following list of books, reports and articles deal specifically with the application and implementation of ISO 9000 in education and training environments: guidelines, interpretations, case studies, critical comments, etc. One should be aware that many of these documents concern a particular type of education or training organisation; some documents are also somewhat biased by the national context or the particular experience of the organisations reported on. The bibliography further includes references to several ISO 9000 standards, as well as to some general books on ISO 9000. It may be useful to note that the ISO 9000 series exists also as European norms (series EN ISO 9000, previously EN 29000) and that authorized translations of these norms exist in many European languages, issued by the national normalisation bodies.

In addition to the documents listed in the bibliography below, the author had access to several internal documents of certified organisations, certifying bodies and experts. Given the confidential nature of such documents and the fact that they are not publicly available, they have not been included.

- 1. AFNOR, 1995: Norme FD X 50-764. L'assurance qualité en formation professionnelle. Mode d'emploi. AFNOR, Paris
- 2. Bartz, W. J., Keuper, R. & Mesenholl, H.-J., 1993: Quality in Continuing Engineering Education. In: Proceedings of the 1993 SEFI Conference on Quality in Engineering Education (Luleå). SEFI, Brussel
- 3. Bartz, W. J., Mesenholl, H.-J. & Keuper, R., 1996: Aspekte der Umsetzung der Qualitätsansprüche in die Praxis. [Vortrag] Technische Akademie Esslingen, Ostfildern
- 4. Bentz, S., 1995: Kvalitetsstyring af Ingeniøruddannelser. DTU, Lyngby
- 5. Blackwell, C., Hector, P. & Sarfaty, D., 1995: Assurance of Quality in Continuing Education. A Handbook (Project Aquaforce Part II). FEANI, Paris
- 6. Bonamy, J., 1994: La qualité dans la formation continue en Europe. Rapport National Français. Document produced for a European Community survey on quality in continuing training. CNRS-IRPEACS, Ecully
- 7. Box, J. & Smeets S., 1996: Continue kwaliteitsverbetering in het hoger onderwijs. Verder met ISO en EFQM. In: Thema, nr 1
- 8. Børrild, E. e.a. (Ed.), 1995: *Erfaringer med kvalitetsudviklung på erhversskoler*. DEL, København
- 9. Brehm, K.-H., 1995: Zum Qualitätsmanagement bei Siemens-Weiterbildungsveranstaltungen Über die DIN ISO 9000 ff. hinaus. In: Qualitätsmanagement und Qualitätssicherung in der Weiterbildung (Ed. Feuchthofen, J.E. & Severing E., Luchterhand, Neuwied
- BSI Quality Assurance, 1994: Managment Systems of Schools. Guidance notes for the application of BS EN ISO 9002 for the management systems of schools. BSI Quality Assurance, Milton Keynes
- 11. BSI Quality Assurance, 1995: *The Guide to Quality Assurance in Education 1995*. Stevenson Publications, London



- 12. BSI Quality Assurance, 1995: Education and Training. Guidance notes on the application of BS EN ISO 9001 for quality management systems in Education and Training. BSI Quality Assurance, Milton Keynes
- 13. BSI Register of Licenses, 1996: Extract covering the sectors "Higher Education", "School Education" and "Education not elsewhere specified". BSI Quality Assurance, Milton Keynes
- 14. Certiked, 1994: Criteria voor kwaliteitsborging. Certiked (Stichting certificatie kennisintensieve dienstverlening), Badhoevedorp
- 15. CERTQUA, 1994: Qualitätssicherungssysteme in Unternehmen und Einrichtungen der Beruflichen Bildung. Certqua, Bonn
- 16. Collins, D., Cockburn, M. & MacRobert, I., 1991: The Applicability of BS 5750 to College Operations (II). Sandwell College
- 17. de Heer, A. & Ahaus, C.T., 1992: ISO 9000-serie en Kwaliteitshandboek. Kluwer, Deventer
- 18. de Jonge, H. & van der Vlist, L., 1996: Kwaliteitsverbetering op basis van kwaliteitsborging. De invoering van ISO-systematiek in een MBO-instelling. In: Thema, nr 1
- 19. Dembski, M. & Lorenz, T., 1995: Zertifizierung von Qualitätsmanagement-systemen bei Bildungsträgern. Expert Verlag, Renningen
- 20. Doherty, G.D., 1995: BS 5750 parts 1 and 2/ISO 9000 (series); 1987 and education do they fit and is it worth it? In: Quality Assurance in Education, Vol 3, Number 3, 1995. MCB University Press
- 21. Drevon, F. e.a., 1993: Les normes qualité applicables à la formation. CRCI des Pays de la Loire, Nantes
- 22. EFQM European Foundation for Quality Management, 1996: Self-assessment based on the European Model for Total Quality Management 1996. EFQM, Brussels
- 23. Feuchthofen, J.E. & Severing, E. (Ed.), 1995: *Qualitätsmanagement und Qualitätssicherung in der Weiterbildung*, Luchterhand Verlag, Neuwied
- 24. Freeman, R.,1993: Quality Assurance in Training and Education. How to Apply BS 5750 (ISO 9000) Standards. Kogan Page, London
- 25. Gelders, L., Proost A. & Van der Heyde, C., 1993: Kwaliteitszorg in een universitaire afdeling: een gevallenstudie. In: Onze Alma Mater. Leuvense Perspectieven (mei 1993). Vlaamse Leergangen Leuven, Leuven
- 26. Gnahs, D., 1996: *Qualitätsmanagementhandbuch für Weiterbildungseinrichtungen.*Max-Traeger-Stiftung der GEW
- 27. Harrewijn, J., 1994: *De nieuwe ISO 9000-familie. De ontwikkeling en implementatie van dynamische kwaliteitssystemen.* Roularta Books, Roeselare
- 28. Højdal L., 1995: Quality management in a Danish Technical College. In: The Quest for Quality Towards Joint European Quality Norms, LDC, Leeuwarden
- 29. ISO International Organization for Standardization, 1991: ISO 9004-2:1991, Quality management and quality system elements Part 2: Guidelines for services. ISO, Genève
- 30. ISO International Organization for Standardization, 1993: ISO 9000-2:1993, Quality management and quality assurance standards Part 2: Generic guidelines for the application of ISO 9001, ISO 9002 and ISO 9003. ISO, Genève



- 31. ISO International Organization for Standardization, 1994: ISO 8402: 1994, Quality management and quality assurance Vocabulary. ISO, Genève
- 32. ISO International Organization for Standardization, 1994: ISO 9000-1:1994, Quality management and quality assurance standards Part 1: Guidelines for selection and use. ISO, Genève
- 33. ISO International Organization for Standardization, 1994: ISO 9001: 1994, Model for quality assurance in design, development, production, installation and servicing. ISO, Genève
- 34. ISO International Organization for Standardization, 1994: ISO 9002: 1994, Model for quality assurance in production, installation and servicing. ISO, Genève
- 35. Jensen, P.B., 1992: ISO 9000. A Guide and Commentary. Buch Jensen Quality Management [DK]
- 36. Johnson, P. L., 1993: *ISO 900. Meeting the new international standards.* McGraw-Hill, New York
- 37. Kerklaan, L.A., 1992: Certificatieplan MKB. Kluwer, Deventer
- 38. Keuper, R. e.a., 1995: Qualität in der beruflichen Bildung. Tips aus des Praxis für eine Qualitätsverbesserung in der beruflichen Weiterbildung. Landesgewerbeamt Baden-Württemberg, Stuttgart
- 39. Kirk, H. J. & Lange, E., 1995: *Tietgenskole og ISO 9001*. Undervisningsministeriet, København
- 40. Kuwan, H. & Waschbüsch, E., 1996: Zertifizierung und Qualitätssicherung in der beruflichen Weiterbildung Kurzfassung. Infratest Burke Sozialforschung, München for the Bundesinstitut für Berufsbildung, Berlin
- 41. Lamprecht, J.L., 1994: ISO 9000 and the Service Sector. Marcel Deker [USA]
- 42. Mills, D., 1993: Quality auditing. Chapman & Hall, London
- 43. Manchester Business School, 1995: ISO 9000 Does it work? SGS Yarsley International, East Grinstead
- 44. Nielsen, A. B., 1995: *En ervervsskoles kvalitetsprojekt*. Undervisningsministeriet, København
- 45. Nielsen, S. P. & Børrild, E., 1996: *Quality debate in initial vocational education in Denmark*. Report commissioned by CEDEFOP, Thessaloniki
- 46. Nijs, R., 1995-96: *ISO-Certificatie voor hogescholen.* [Deel 1 & 2] In: *Persoon en Gemeenschap 95/95,* nr 3-5
- 47. Peek, G. & van der Wiele, T., 1995: Richtlijn voor het auditen van onderwijsinstellingen. Kwaliteitsborging, certificatie in onderwijs een feit! NEN-ISO 9001: Onderwijs met kwaliteitsgarantie. SGS European Quality Certification Institute
- 48. Persson, P., Thorman, J. & Larsen, A., 1995: Kan man lege med ISO? Hillerød Tekniske Centralskoles kvalitetsudviklungsprojekt. In: Erfaringer med Kvalitetsudviklung på Erhvervsskoler. DEL, København
- 49. Raad voor de Certificatie, 1994: *De ISO-9001 Certificatiecriteria voor opleidingsin-stellingen*. CEDEO, Den Haag
- 50. Rooney, M., 1994: Guidelines on the application of the ISO 9000 series to further education and training. NACCB (National Accreditation of Certification Bodies), London



95 106

- 51. Severing, E., Stahl, T., 1995: Qualitätssicherung in der betrieblichen Weiterbildung Fallstudien aus Europa. Synthesis study of the European Community survey on quality in continuing training. ISOB (Institut für sozialwissenschaftliche Beratung), Regensburg
- 52. Todorov, B., 1994 (2nd ed.): *ISO 9000, un passeport mondial pour le management de la qualité*. Gaëtan Morin, Montréal
- 53. Undervisningsministeriet, Erhvervsskoleafdelingen, August 1995: Strategioplæg om kvalitetsudvikling og resultatvurdering i erhvervsskolesektoren
- 54. van den Berg, A. & van de Minkelis, A., 1994: *ISO-audit vragenlijst Trainings- en Opleidingsinstituten*. DNV Industry, Rotterdam
- 55. Van den Berghe, W., 1995: Achieving Quality in Training. European Guide for collaborative training projects. Tilkon, Wetteren (B)
- 56. Van den Berghe, W., 1995: The ISO 9000 norms and the certification of training providers. Contribution to a European Seminar Quality in Continuing Education and Training an in-depth examination (Brussels), organized by Technologisch Instituut, Antwerpen
- 57. Van den Berghe, W., 1995: De markt heeft weer eens het laatste woord. Onderwijsinstellingen overwegen ISO-certificaat: er bestaat geen alternatief met dezelfde herkenbaarheid. In Intermediair (BelgiN), nr 37
- 58. Van den Berghe, W., 1996: *Quality Issues and Trends in Vocational Education and Training in Europe*. CEDEFOP, Thessaloniki
- 59. Van den Berghe, W. (Ed.), 1997: *PROZA Een instrument voor zelfanalyse en kwaliteitszorg in hogescholen*. [To be published]
- 60. Wilson, G., 1994: *Quality Assurance in Higher Education*. Ulster Business School, Newtonabbey [TPOK/Juran Institute]

About the author

Wouter Van den Berghe is managing director of Tilkon, an independent research and consultancy office specialised in questions of quality, education and training, innovation, international project cooperation, and organisational change. He is the author of several books, reports and articles on these issues. Wouter Van den Berghe undertakes study, training and consultancy assignments for public authorities, international organisations, research institutes, consultancy offices, educational institutions, training providers and firms in many European countries.

Comments and feedback on this report are welcomed on the following address:

Tilkon, Kerkwegel 12a, B-9230 Wetteren (Belgium);

fax: +32-9-366.28.35

e-mail: info@tilkon.eunet.be



96 101

Application of ISO 9000 standards to education and training Interpretation and guidelines in a European perspective

Wouter Van den Berghe

CEDEFOP Document

Luxembourg: Office for Official Publications of the European Communities

1998 — V, 96 pp. — 21 x 29.7 cm

ISBN 92-828-2822-0

Price (excluding VAT) in Luxembourg: ECU 8.50

Venta • Salg • Verkauf • Πωλήσεις • Sales • Vente • Vendita • Verkoop • Venda • Myynti • Försäljning

BELGIQUE/BELGIÉ
Moniteur betge/Belgiach Stantsblad
Noniteur betge/Belgiach Stantsblad
Rue de Lourain 40-42/Leuvenseweg 40-42
6-1000 Bruxellee/Brussel
Fax (32-2) 551 01 84
Jean De Lannoy
Avenue du Roi 202/Konsiel
Tél: (32-2) 553 51 89
Fax (32-2) 553 69 61
E-mail: jean.de.lannoy/@infoboard.be
URL: http://www.jean-de-lannoy.be

Librairie européenne/Europeae Boekhande Rue de la Loi 244/Wetstraal 244 B-1040 Bruxelee/Brussel Tdi. (32-2) 295 26 39 Fax (32-2) 735 08 60

DANMARK
J. H. Schultz Information A/S
Herstedvarg 10-12
DK-2620 Albertslund
TII. (45) 43 63 23 00
E-mail: schultz @ schultz.dk
URL: http://www.schultz.dk

DEUTSCHLAND
Bundesanzeiger Vertag
Bretts Straße 78-80
Postlach 10 05 34
D-50687 Kbn
Tel, (49-221) 20 29-0
E-mail: vertrieb 8 bundesanzeiger.de
URL: http://www.bundesanzeiger.de

EAAAAA/GREECE
G. C. Eleftheroudakis SA International Bookstore Panepistriniou 17
GR-10564 Athina
Tel. (30-1) 331 41 80/1/2/3
Fax (30-1) 323 98 21
E-mail: elebooks @ netor.gr

ESPAÑA
Mundi Prense Libros, SA
Castelló, 37
E-28001 Madrid
Tel (34-1) 431 33 99
Fax (34-1) 575 39 98
Fax (34-1) 575 39 98
URL: http://www.mundiprense.es

Boletin Official del Estavo-Tratalgar, 27 E-28010 Macrid Tel. (34-1) 539 21 11 (Librosy) 394 17 15 (Suscripciones) Estavología (1) (Librosy) Estavología (1) (Librosy) Experience (1) (Librosy) Exp

FRANCE
Journal official
Service des publications des CE
26, rue Desaix
F-75727 Paris Cedex 15
Tól. (33) 140 58 77 0131
Fax (33) 140 58 77 00

IRELAND
Government Supplies Agency
Publications Section
4-5 Harcourt Roed
Dublin 2
Tel. (353-1) 661 31 11
Fax (353-1) 475 27 60

Via Duca di Catabria, 1/1 Casella postale 552 1-501 25 Firenze Tel. (39-55) 64 54 15 Fax (39-55) 64 12 57 E-mail: ilicosa @ fiboc.it/ URL: http://www.fiboc.it/licosa

LUXEMBOURG

Messagaries du tivre SARL
5. rue Raiffeisen
L-2411 Luxembourg
Tél. (352) 49 10 20
Fax (332) 49 06 61
E-mai: mgl @pt.lu

Abonnements:
Messageries Paul Kraus
11, nue Christophe Plantin
L-2339 Luxembourg
764 (352) 49 98 88-8
Fax (352) 49 98 88-444
E-mail: mpk @pt.tu
URL: http://www.mpk.tu

NEDERLAND
SDU Servicecentrum Ultgevers
Extems Footbale
Positius 20014e
12500 EA Own Haing
Tel. (31-70) 378 98 80
Fax (31-70) 378 98 85
E-mail: indu di edu.uri
URL: http://www.arb.u.inl

OSTERREICH
Ment'sche Verlags- und
Universitätsbuchhandlung GmbH
Slebenbrunnengasse 21
Postfach in
Fostfach in
Tel. (46-1) 53 16 13 30-406
Fax (46-1) 53 16 13 30
E-mail: aussielerung @ manz, co.at
URL: http://www.austria.EU.netic.il/manz

PORTUGAL Imprensa Nacional-Casa da Moeda, EP Rus Marqués de Sá da Bandeira, 18 A P-1050 Lisboa Codex Tel. (351-1) 353 09 99 Fax (351-1) 353 02 94, 384 01 32

Distributions de Livroe Bertrand Ld.* Rue das Terras dos Vales, 4/A Apartado 60037 P-2701 Amadora Codex Tel. (351-1) 496 02 55 Fax (351-1) 496 02 55

SUOMI/FINLAND
Akateeminen Kirjaksupps/Akademiska

Pohjoleasplaned 39/ Norra esplanaden 39 PUPB 128 FIN-00101 Helsinki/Helsingfors PJ/m (358-6) 121 41 FJ/mx (358-6) 121 44 35 E-milk: akallisu @ stockmann.mailnet,fi UPL: http://booknet.custnet.fi/aka/index.htm

SVERIGE
BTJ AB
Traktorvlegen 11
S-221 82 Lund
Tra (46-46) 18 00 00
Fax (46-46) 30 79 47
Epost brilley-pul 6 big se
URL: http://www.big.se/meds/eu

The Stationery Office Ltd International Sales Apercy 51 Nine Elms Lane London SW6 50R Tel. (44-171) 873 99 99 Fax (44-171) 873 64 63 E-mail: jill speed @theso.co.uk

SLAND

Bokabud Larusar Blöndar
Skólavördustig, 2
IS-101 Reykjavik
Tel. (354) 551 59 50
Fax (354) 552 55 60

NORGE
NIC Info A/S
Ostenjoveien 18
Boks 6512 Ettenstad
N-0606 Oslo
Tel. (47-22) 97 45 00
Eax (47-22) 87 45 45

SCHWEIZ/SUISSE/SVIZZERA
OSEC
Stampferobachstraße 85
CH-8035 Zünch
Tel. (41-1) 365 53 15
Fax (41-1) 365 5417
E-mail: udeimbacher 9 csec.ch
URL: http://www.csec.ch

BÅLGARIJA Europress-Euromedia Ltd 59, Bid Vitosha BG-1000 Sofia Tel. (359-2) 980 37 66

ČESKÁ REPUBLIKA

NIS CR -- prodejna

Korviktská 5

CZ-11357 Praha 1

Tel. (420-2) 24 22 94 33, 24 23 09 07

Fax (420-2) 24 22 94 33

E-mali: náposp © doc. nís cz

UPIL: http://www.nis.cz

Cyprus Chember of Commerce & Inclustry
Grive-Olgent 38 & Designorg 3
Mail ordens:
PO Box 1455
CY-1509 Nicosis
Tel. (357-2) 44 95 00, 46 23 12
Fax (357-2) 30 10 44
E-mail: cy1691_elc_cyprus Ø vans.infonet.com

MAGYARORSZÁG
Euro trifo Service
Európa Ház
MargitazjógPÓ Boz 475Hő Boz 475

MALTA
Miller Distributors Ltd
Malta International Airport
PO Box 25
LQA 05 Malta
Tel. (356) 66 44 88
Fax (356) 87 67 99

POLSKA

Ars Polona Krakowskie Przedmiescie 7 Skr. pocztowa 1001 PL-00-950 Warszawa 154, (49-22) 826 12 01 Fax (49-22) 826 12 40, 826 53 34, 826 86 73 E-mail: an., pol ©bery, hen.com.pl

ROMÂNIA Euromedia Str. G-rai Berthelot Nr 41 RO-70749 Bucuresti Tél. (40-1) 210 44 01, 614 06 64 Fax (40-1) 210 44 01, 312 96 46

SLOVAKIA
Slovak Centre of Scientific and Technica
Information
Namestie alchody: 10
SK-81223 Bratislava 1
Tel. (421-7) 531 83 84
Fax (421-7) 531 83 64
Fax (421-7) 531 83 64
Fax (421-7) 531 83 64

SLOVENIA
Gospodarski Vestnik
Zalozniska skupina d.d.
Dunajska cesta 5
SLO-1000 Lubtipana
Tek (386) 611 33 03 54
Fax (388) 611 39 12 8
E-mall: belicd © gresstnik, si
URL: http://www.gresstnik.si

TÜRKIYE

DÜnya Infotel AS
Istikál Cad. No: 469
TR-80050 Tünel-Istanbul
Tel. (90-212) 251 91 96
Fax (90-212) 251 91 97

AUSTRALIA Hunter Publications PO Box 404 3167 Abbotsford, Victoria Tel. (61-3) 94 17 53 1 Fax (61-3) 94 19 51 54

CANADA
Subscription crely/Uniquement abonnement
Remout Publishing Co. Ltd
S980 Chemin Canotes, Road Unt 1
K11 33,0 Citava, Chemin
Tel. (1-613) 745 26 65
Fax: (1-613) 745 26 65
E-mail: remout 69 for. rash.c.
Tel. (1-10) 740 75 76 60
E-mail: remout 69 for. rash.c.

EGYPT
The Middle East Observer
41, Sherif Street
Cairo
Tel. (20-2) 393 97 32
Fax (20-2) 393 97 32

Mediatrade Ltd Pavta Hatza 1 HR-10000 Zagreb Tel. (385-1) 43 03 92 Fax (385-1) 43 03 92

HRVATSKA

INDIA
EBIC Indie
BAC Indie
Gen. J. Brossle Mary.
400 t21 Nambel
101 (91-22) 222 60 64
Fax (91-22) 235 45 64
Fax (91-22) 235 45 64
Fax (91-22) 255 45 64

ISRAÉL
ROY International
17, Shimon Haterssi Street
PO Box 13056
61130 Tel Alw
Tel, (972-3) 546 14 42
E-mail: royl@newission.net.8
Sub-agent for the Palestinian Author

Sub-agent for the Palestinian Authority: Index Information Services PO Box 19502 Jerussiem Tel. (972-2) 627 16 34 Fax (972-2) 627 12 19

JAPAN
PSI-Japan
Asahi Sanbancho Ptaza #206
7-1 Sanbancho, Chiyoda-ku
Tokyo 102
Tel. (61-3) 32 34 69 21
Fax (61-3) 32 34 69 15
E-mail: psispan @ gol com
URL: http://www.psi-japan.com

MALAYSIA

EBIC Malaysis
Level 7, Wisma Hong Leong
18 Jalan Perak
50450 Kasla Lumpur
Tel. (60-3) 262 62 98
Fax (60-3) 262 61 98
E-mail: ebio-id@mol.net.my

PHILIPPINES

EBIC Philippines

19th Floor, PS Bank Tower Sen.

Gill J. Puyal Ave. cor. Tindalo St.

Makati City

Metro Manita

Tel. (63-2) 759 68 80

Fax (63-2) 759 68 90

E-mail: ecopcom@globe.com.ph

RUSSIA CCEC 60-letlya Oktyabrya Av. 9 117312 Moscow Tel. (70-95) 135 52 27 Fax (70-95) 135 52 27

SOUTH AFRICA
Sefto
Sth Floor Export House,
CNR Maude & West Streets
PO Box 782 706
2148 Sandton
Tel. (27-11) 883 37 37
Fax (27-11) 833 65 69

SOUTH KOREA Kyowe Book Company 1 F1, Phyung Hwa Bidy 11-2 Hap Leong Dong, Mapo Ku 121-220 Seoul 781, (82-2) 325 87 80'1 Fax (82-2) 325 87 80'1 Fax (82-2) 325 87 80'1 Fax (82-2) 325 87 80'1

EBIC Thailland
Vanissa Bulding 8th Floor
29 Soi Children
Ploenchill
10330 Bangkok
Tel. (66-2) 655 06 27
Fax. (56-2) 655 06 28
E-mail: ebicblok@isc15.th.com

UNITED STATES OF AMÉRICA

THAILANDE

Bernan Associates
4611F Assembly Orive
MD20706 Lanham
Tot. (800) 274 44 47 (toli free letephone)
Fax (800) 855 34 50 (toli free fax)
E-mait, quary d'bernan.com
URL: http://www.bernan.com

ANDERE LÂNDER/OTHER COUNTRIES/ AUTRIES PAYS

Bittle wenders Sie sich an ein Büro Ihrer Wahl / Pfease contact the sales office of your choice / Veutilitz vous adresser au bursau die werts de votre choix

2

င္ပ

6

European Centre for the Development of Vocational Training Marinou Antipa 12, GR-57001 Thessaloniki

Tel. (30-31) 49 01 11 Fax (30-31) 49 01 02 E-mail: info@cedefop.gr

Internet: http://www.cedefop.gr

Price (excluding VAT) in Luxembourg: ECU 8.50

BEST COPY AVAILABLE

0-5585-858-5P

ISBN

OFFICE FOR OFFICIAL PUBLICATIONS OF THE EUROPEAN COMMUNITIES



U.S. DEPARTMENT OF EDUCATION

Office of Educational Research and Improvement (OERI) Educational Resources Information Center (ERIC)



NOTICE

REPRODUCTION BASIS

This document is covered by a signed "Reproduction Release (Blanket)" form (on file within the ERIC system), encompassing all or classes of documents from its source organization and, therefore, does not require a "Specific Document" Release form.
This document is Federally-funded, or carries its own permission to reproduce, or is otherwise in the public domain and, therefore, may be reproduced by ERIC without a signed Reproduction Release form (either "Specific Document" or "Blanket").

